Université Paris 13 Laboratoire d'Informatique de Paris Nord

Habilitation à Diriger des Recherches

Spécialité : Sciences

Calcul Formel pour la Combinatoire des Marches

Soutenue le 15 décembre 2017 par

Alin Bostan (Inria)

devant le jury composé de :

Mme.	Frédérique Bassino	Université Paris 13, Villetaneuse	
M.	Olivier Bodini	Université Paris 13, Villetaneuse	
Mme.	Mireille Bousquet-Mélou	CNRS, Université de Bordeaux	
Mme.	Lucia DI VIZIO	CNRS, Université de Versailles	
M.	Mark Giesbrecht	Université de Waterloo, Canada	(rapporteur)
M.	Florent Hivert	Université Paris 11, Orsay	
M.	Christian KRATTENTHALER	Université de Vienne, Autriche	(rapporteur)
M.	Gilles Villard	CNRS, ENS de Lyon	(rapporteur)

COMPUTER ALGEBRA FOR LATTICE PATH COMBINATORICS

ALIN BOSTAN*

Abstract. Classifying lattice walks in restricted lattices is an important problem in enumerative combinatorics. Recently, computer algebra has been used to explore and to solve a number of difficult questions related to lattice walks. We give an overview of recent results on structural properties and explicit formulas for generating functions of walks in the quarter plane, with an emphasis on the algorithmic methodology.

Key words. Enumerative combinatorics, random walks in cones, lattice paths in the quarter plane, Gessel walks, generating functions, computer algebra, automated guessing, creative telescoping, diagonals, binomial sums, algebraic functions, D-finite functions, hypergeometric functions, elliptic integrals.

AMS subject classifications. Primary 05A10, 05A15, 05A16, 97N70, 33F10, 68W30, 14Q20; Secondary 33C05, 97N80, 13P15, 33C75, 12Y05, 13P05, 14Q20.

This document is structured as follows. Section 1 gives an overview of recent results obtained in lattice path combinatorics with the help of computer algebra, with a focus on the exact enumeration of walks confined to the quarter plane. Sections 2 and 3 then go into more details of two classes of fruitful algorithmic approaches: *guess-and-prove* and *creative telescoping*.

1. General presentation.

1.1. Prelude. Consider the following innocent-looking problem.

A *tandem-walk* is a path in \mathbb{Z}^2 taking steps from $\{\uparrow, \leftarrow, \searrow\}$ only. Show that, for any integer $n \ge 0$, the following quantities are equal: (*i*) the number a_n of tandem-walks of length n (i.e., using n steps), confined to the upper half-plane $\mathbb{Z} \times \mathbb{N}$, that start and end at (0,0); (*ii*) the number b_n of tandem-walks of length n confined to the quarter plane \mathbb{N}^2 , that start at (0,0) and finish on the diagonal x = y.

For instance, for n = 3, this common value is $a_3 = b_3 = 3$, as shown below.



The problem establishes a rather surprising connection between tandem-walks in the lattice plane, submitted to two different kinds of constraints: the evolution domain of the walk, and its ending point. The domain constraint is weaker for the first family of walks, while the ending constraint is relaxed for the second family.

It appears that this problem is far from being trivial. Several solutions exist, but none of them is elementary. One of the main aims of the present text is to

^{*}Inria, Université Paris-Saclay, 91120 Palaiseau, France (alin.bostan@inria.fr).

convince the reader that this problem (and many others with a similar flavor) can be solved with the help of a computer. More precisely, Computer Algebra tools, extensively described in the following sections, can be used to discover and to prove the following equalities

(1)
$$a_{3n} = b_{3n} = \frac{(3n)!}{n!^2 \cdot (n+1)!}$$
, and $a_m = b_m = 0$ if 3 does not divide m .

It goes without saying that such a simple and beautiful expression cannot be an element of chance. As it will turn out, closed forms are quite rare for this kind of enumeration problems. Nevertheless, even in absence of nice formulas, the structural properties of the corresponding enumeration sequences reflect the symmetries of the step set and of the evolution domain. Equation (1) shows that the sequences (a_n) and (b_n) are P-recursive, that is, they satisfy a linear recurrence with polynomial coefficients (in the index *n*). One of the messages that will emerge from the text is that this important property of the enumeration sequences is intimately related to the finiteness of a certain group, naturally attached to the step set { $\uparrow, \leftarrow, \searrow$ }.

1.2. General context: lattice paths confined to cones. Let us put the previous problem into a more general framework. Let $d \ge 1$ be an integer (dimension), let \mathfrak{S} be a finite subset (called step set, or model) of vectors in \mathbb{Z}^d , and $p_0 \in \mathbb{Z}^d$ (starting point). A \mathfrak{S} -path (or \mathfrak{S} -walk) of length n starting at p_0 is a sequence (p_0, p_1, \ldots, p_n) of elements in the lattice \mathbb{Z}^d such that $p_{i+1} - p_i \in \mathfrak{S}$ for all $0 \le i < n$. Let \mathfrak{C} be a cone of \mathbb{R}^d , that is a subset of \mathbb{R}^d such that $r \cdot v \in \mathfrak{C}$ for any $v \in \mathfrak{C}$ and $r \ge 0$, assumed to contain p_0 . We will be interested in the (exact and asymptotic) enumeration of \mathfrak{S} -walks confined to the cone \mathfrak{C} , and potentially subject to additional constraints.

Example 1. Consider the model $\mathfrak{S} = \{(1,0), (-1,0), (1,-1), (-1,1)\}$ (called the *Gouyou-Beauchamps model*) in dimension d = 2, with starting point $p_0 = (0,0)$ and with cone $\mathfrak{C} = \mathbb{R}^2_+$ (the quarter plane). The picture below displays the step set of the model (on the left), and a \mathfrak{S} -walk of length n = 17 confined to \mathfrak{C} (on the right).



The main typical questions in this context are then the following:

- What is the number a_n of *n*-step \mathfrak{S} -walks contained in \mathfrak{C} and starting at p_0 ?
- For fixed $i \in \mathfrak{C}$, what is the number $a_{n;i}$ of such walks that end at *i*?
- What is the nature of their generating functions

$$A(t) = \sum_{n} a_n t^n \text{ and } A(t; \mathbf{x}) = \sum_{n,i} a_{n,i} t^n \mathbf{x}^i?$$

As expected from the introductory example of tandem-walks, the answers to these questions are not simple, and heavily depend on the various parameters. The aim of this text is to provide a survey of recent results —notably classification results and closed form expressions— obtained using Computer Algebra.

1.3. Why count walks in cones?. Lattice paths are fundamental objects in combinatorics. They have been studied at least since the second half of the 19th century, in connection with the *ballot problem* (see $\S1.4$). Even earlier, embryonic occurrences (around 1650) are in Pascal's and Huygens' solutions of the so-called problem of division of the stakes (or, problem of points), and of the gambler's ruin problem, which motivated the beginnings of modern probability theory [170, 226, 157]. Despite these historically important examples, the enumeration of lattice walks has long remained part of what may be called recreational mathematics. It is only in the late 1960s that their study really became an independent field of research, at the crossroads of pure and applied mathematics. Since then, various approaches have been progressively involved, separately or in interaction, in the study of lattice walks. These methods arise from various fields of classical mathematics (algebra, combinatorics, complex analysis, probability theory), and more recently from computer science. There are several reasons for the ubiquity of lattice walks, but the most solid one is that they encode several important classes of mathematical objects, in discrete mathematics (permutations, trees, words, urns, ...), in statistical physics (magnetism, polymers, ...), in probability theory (branching processes, games of chance, ...), in operations research (birth-death processes, queueing theory, ...). Therefore, many questions from all these various fields can be reduced to solving lattice path problems. For more motivations, the reader is referred to the introduction of [26]. Nowadays, several books are entirely devoted to lattice paths and their applications [355, 312, 315, 160, 146, 384, 180, 388, 47, 284, 44], and an international conference titled *Lattice path combinatorics and applications* is entirely devoted to this field. We recommend Humphreys' article [237] for a brief review of the history of lattice path enumeration and for a survey of the recent evolution of the field. Also, Krattenthaler's recent survey [269] is an excellent overview of various results and methods in lattice path enumeration.

1.4. The ballot problem and the reflection principle. As mentioned before, the enumeration of lattice walks is an old topic. We want to illustrate this using Bertrand's ballot problem [36, 10]. The aim is not only to provide the flavor of a nice piece of combinatorial reasoning, but especially to introduce the so-called *reflection principle*, seemingly invented by Aebly and Mirimanoff [5, 306], which contains the roots of a systematic method for lattice walks, to be presented later, and based on the notion of *group of a walk*, see §1.18. Bertrand's problem is the following:

Suppose that two candidates *A* and *B* are running in an election. If *a* votes are cast for *A* and *b* votes are cast for *B*, where a > b, then what is the probability that *A* stays (strictly) ahead of *B* throughout the counting of the ballots?

The problem admits an obvious lattice path reformulation. Let us call a *Dyck path* a walk in the lattice plane \mathbb{Z}^2 , with step set $\mathfrak{S} = \{(1,1), (1,-1)\} = \{\nearrow, \searrow\}$, that starts at the origin. Then, the problem asks for the number of Dyck paths consisting of *a* upsteps \nearrow and *b* downsteps \searrow such that no step ends on the *x*-axis. Let us call these *good paths*. Clearly, any such good path starts with a step from (0,0) to (1,1), and finishes at the point T(a + b, a - b). Instead of counting good paths, it is actually easier to count *bad paths*: these are Dyck paths consisting of *a* upsteps \nearrow and *b* downsteps \searrow that touch the *x*-axis at least once. Now enters the crucial observation, based on a reflection argument (see the picture).



To any bad path one may bijectively attach an unconstrained path in \mathbb{Z}^2 from (1, -1) to *T* by simply reflecting, with respect to the horizontal axis, the first portion of the walk, which lies strictly above the horizontal axis before touching it for the first time. Therefore, the number of good paths is exactly the difference between the unconstrained Dyck paths in \mathbb{Z}^2 from (1, 1) to T(a + b, a - b) and the unconstrained Dyck paths are simply counted by binomials, that number is:

$$\binom{a+b-1}{a-1} - \binom{a+b-1}{b-1} = \frac{a-b}{a+b}\binom{a+b}{a},$$

from which one directly deduces the answer (a - b)/(a + b) to Bertrand's problem. Observe that, when a = n + 1 and b = n, the number of good paths is the famous Catalan number

$$C_n = \frac{1}{2n+1} {2n+1 \choose n+1} = \frac{1}{n+1} {2n \choose n},$$

that counts a plethora of different combinatorial objects [115, 116, 358].

There exists a second (non-strict) version of the problem, in which *A* has at least as many votes as *B* all along the counting. The reflection principle still applies, and the answer is 1 - b/(a + 1). More information, and historical background, on the ballot problem is provided in the articles [28, 340].

Last, but not least, let us mention that a higher dimensional version of the reflection principle [218, 391] can be used to solve the following generalization of the ballot problem: Assume there are *d* candidates in an election, say A_1, \ldots, A_d , with each A_i receiving a_i votes. What is the probability that, throughout the counting of the ballots, A_i has at least as many votes as A_{i+1} for all $1 \le i \le d-1$? This amounts to counting paths in \mathbb{Z}^d from the origin to (a_1, \ldots, a_d) that use only unit positive steps (in the direction of some coordinate axis) and that are confined to the *edge cone* $\{x_1 \ge x_2 \ge \cdots \ge x_d \ge 0\}$. The natural setting for the most general version of the reflection principle is the one of reflection groups: it applies when the set of steps is left invariant by a Weyl group and the walks are confined to a corresponding Weyl chamber see [205, 211] and [269, §10.18].

1.5. Pólya's "promenade au hasard" / "Irrfahrt". Another old and famous result on lattice paths is Pólya's theorem [328, 329]* about the so-called *drunkard walk* in the *d*-dimensional integer lattice \mathbb{Z}^d . By definition, such a walk is a random path in \mathbb{Z}^d for the so-called *simple model*, or *Pólya's model*. After a busy night at the bar (some vertex of \mathbb{Z}^d), a drunkard wishes to get home (another vertex of \mathbb{Z}^d). Given his mental and physical state, he cannot do better than executing a random walk starting from the bar: at each tick of the clock he moves to one of the 2*d* neighbors of the current vertex, chosen uniformly at random. What is the probability that he

^{*}References to Pólya's work [8] will appear repeatedly and crucially in the three main parts of this text. It is thus not an exaggeration to pretend that Pólya's influence is our guiding thread.

ever reaches his destination? The interesting fact is that the long-term behavior of the drunkard's walk depends on the dimension *d*.

THEOREM 2 (Pólya, 1921). Consider the simple random walk on \mathbb{Z}^d . If $d \in \{1, 2\}$, then the walk returns to its starting position with probability 1 (the simple walk is recurrent). If $d \geq 3$, then with positive probability, the walk never returns to its starting position (the simple walk is transient).[†]

Several proofs exist for this classical result. Probably the most direct one [185, §XIV.7] is based on the observation that the probability for the *d*-dimensional drunkard to be back at the origin after 2n steps is equal to the (d - 1)-folded sum

$$u_{2n}^{(d)} = \sum_{i_1 + \dots + i_d = n} \frac{(2n)!}{(i_1! \cdots i_d!)^2} \left(\frac{1}{2d}\right)^{2n}.$$

Then some algebraic manipulations and Stirling's formula imply the asymptotic estimate $u_{2n}^{(d)} = \Theta(n^{-d/2})$. On the other hand, it is not hard to see that the walk is transient if and only if the series $\sum_{n\geq 0} u_{2n}^{(d)}$ converges, namely to a value m_d which is the expected number of returns at the origin.

As a consequence of Theorem 2, if the drunkard lives in a 2-dimensional city, then he will eventually get home, even though possibly after a very long amount of time. But if, by misfortune, he lives in a 3-dimensional city, then the probability p_3 of return home will be less than 1. Pólya did not find a value for p_3 ; this was done later by McCrea and Whipple [300] who showed that $p_3 \approx 0.34053$. A beautiful exact formula for p_3 was found by Glasser and Zucker [207], in terms of Euler's gamma function $\Gamma(x) = \int_0^\infty e^{-t}t^{x-1}dt$. It reads $p_3 = 1 - 1/m_3$, where $m_3 = \frac{\sqrt{6}}{32\pi^3} \Gamma\left(\frac{1}{24}\right) \Gamma\left(\frac{5}{24}\right) \Gamma\left(\frac{7}{24}\right) \Gamma\left(\frac{11}{24}\right) \approx 1.516386060$, see also [160, §2.3.5] and [54, 222, 397, 265]. No similar closed-form expression is known for $d \ge 4$, although it was proved [314] that the probability of return p_d equals $1 - 1/m_d$, with

$$m_d = \frac{d}{(2\pi)^d} \int_{-\pi}^{\pi} \cdots \int_{-\pi}^{\pi} \frac{dx_1 \cdots dx_d}{d - \cos x_1 - \dots - \cos x_d} = \int_0^{\infty} (I_0(t/d))^d e^{-t} dt,$$

where $I_0(t)$ is the modified Bessel function of the first kind $I_0(t) = \sum_{k\geq 0} \frac{(t^2/4)^k}{k!^2}$. A question closely related with Pólya's theorem will be discussed in §1.7.

1.6. Blending Experimental Mathematics and Computer Algebra in the service of lattice paths combinatorics. The examples in §1.4 and §1.5 show that the study of lattice walks is an old field of research. The following sections will demonstrate that their exact and asymptotic enumeration is still a topical issue, with a lot of recent activity, new and exciting results, and many open questions. For instance, even when only restricting to articles published since 2000, and when only focusing to the case of walks confined to the quarter plane, one realizes that this particular case has received special attention, and much progress has been done by many recent contributors [129, 366, 25, 26, 94, 95, 238, 103, 239, 96, 318, 100, 31, 307, 257, 19, 84, 254, 308, 310, 45, 85, 101, 181, 182, 220, 275, 276, 183, 277, 336, 367, 273, 301, 339, 338, 90, 164, 303, 302, 7, 89, 156, 184, 179, 256, 278, 20, 32, 60, 99, 98, 153, 196, 304, 305, 76, 86, 150, 162, 255, 309]. And this is certainly not an exhaustive list.

[†]As Feller says [185, p. 360], the statement "all roads lead to Rome" is justified in two dimensions.

The dominating point of view in these works is to develop uniform approaches, rather than ad-hoc solutions to a specific question. My personal bias is twofold: combine an *experimental mathematics approach*, as promoted in the beautiful and inspiring books by Borwein and collaborators [49, 22, 48, 51], with *modern tools from the Computer Algebra arsenal* as described in the recent reference textbooks [383, 70], in order to *conjecture and prove* enumerative and asymptotic results for lattice paths.

Over the last three decades a fundamental shift has been operated in the way mathematics is practiced. As a consequence of the continued advance of computing power and of the unceasing availability of modern computational software, one can nowadays really take advantage of computer-aided research in order to solve significant and difficult mathematical problems. Our goal in this memoir is to overview computational approaches to discovery of new results in lattice path combinatorics. We entirely share Borwein's viewpoint that mathematical discovery through experimentation and the use of increasingly intelligent software is going to play an essential role in other fields of mathematics.

1.7. Another example, from the SIAM 100-Digit Challenge [375, 46]. In a 2002 SIAM News article [375], L. N. Trefethen, head of the Numerical Analysis Group at Oxford University, proposed a contest which consisted of ten challenging problems in numerical computing. Each problem was stated in at most three simple sentences and had a single real number as a solution. The objective was to compute each number to as many digits of precision as possible. Scoring for the contest would be simple: each correct digit of the answer, up to ten per problem, would earn a single point. Trefethen warned that the problems were hard and indicated that he would be impressed if anyone managed to score even 50 points. Problem 6 in his list was about lattice walks in the plane, and appears to be related to Pólya's problem.

Problem 6 (Biasing for a Fair Return)

A flea starts at (0,0) on the infinite two-dimensional integer lattice and executes a biased random walk: At each step it hops north or south with probability 1/4, east with probability $1/4 + \varepsilon$, and west with probability $1/4 - \varepsilon$. The probability that the flea returns to (0,0) sometime during its wanderings is 1/2. What is ε ?

As demonstrated in the wonderful book [46, Chap. 6], and in §3.2.1, Computer Algebra is able to *conjecture* and to *prove* the following formula

$$p(\epsilon) = 1 - \sqrt{\frac{A}{2}} \cdot {}_{2}F_{1} \left(\begin{array}{c} \frac{1}{2}, \frac{1}{2} \\ 1 \end{array} \middle| \frac{2\sqrt{1 - 16\epsilon^{2}}}{A} \right)^{-1}, \text{ with } A = 1 + 8\epsilon^{2} + \sqrt{1 - 16\epsilon^{2}},$$

where ${}_{2}F_{1} \left(\begin{array}{c} \frac{1}{2}, \frac{1}{2} \\ 1 \end{array} \middle| t \right) = \sum_{n \ge 0} {\binom{2n}{n}}^{2} \left(\frac{t}{16} \right)^{n}.$

From this *exact* expression, it is easy to get the first 100 digits of the result

 $\epsilon \approx 0.0619139544739909428481752164732121769996387749983 \\ 6207606146725885993101029759615845907105645752087861 \ldots$

and actually millions of digits, if needed, in not more than a couple of seconds.

1.8. Two basic cones: the full space and a (rational) half-space. Let us now turn back to the general problem as stated in §1.2, using notion introduced in there. The simplest possible cone is the full space $\mathfrak{C} = \mathbb{R}^d$. In that case, the situation is very simple: the full generating function has the most basic structure, it is *rational*.

THEOREM 3. If $\mathfrak{S} \subset \mathbb{Z}^d$ and $\mathfrak{C} = \mathbb{R}^d$, then

$$a_n = |\mathfrak{S}|^n$$
, i.e. $A(t) = \sum_{n \ge 0} a_n t^n = \frac{1}{1 - |\mathfrak{S}| t}$.

More generally:

$$A(t;\mathbf{x}) = \sum_{n,i} a_{n,i} \mathbf{x}^i t^n = \frac{1}{1 - t \sum_{\mathbf{s} \in \mathfrak{S}} \mathbf{x}^{\mathbf{s}}}.$$

The next case by increasing order of difficulty is when the cone is a half-space. The full generating function is not rational anymore, but nevertheless it still has a very important property: it is *algebraic*.

THEOREM 4. If $\mathfrak{S} \subset \mathbb{Z}^d$ and if \mathfrak{C} is a rational half-space, then $A(t; \mathbf{x})$ is algebraic, given by an explicit system of polynomial equations.

This result is due to Bousquet-Mélou and Petkovšek, see [102, Theorem 13] and [103, Proposition 2]. Roots of it are in [323, 324]. The important particular case of 2D "generalized Dyck paths" had been treated before, see [203, 280, 279, 163]. The most basic illustration is provided by the ballot problem (§1.4), for which A(t; 1) = $\sum_{n>0} C_n t^n = (1 - \sqrt{1 - 4t})/(2t)$, see Example 5 below.

The main ingredient in the proof of [102] of Theorem 4, called the kernel method (terminology coined in [26]), seems to belong to the "mathematical folklore". One source of this method, identified by Banderier and Flajolet in [26, p. 55], is Knuth's book [261, §2.2.1], more precisely his solutions to Exercises 4 and 11, which use a "new method for solving the ballot problem". Knuth's trick may have been better known at that time in probability theory, as suggested by its use in a more involved context [293, 294, 191, 177, 178]. Various examples of its use in combinatorics are presented by Prodinger in [335]. More historical notes on the origins of the kernel method can be found in [26, §2.2] and in [27, §1]. It is my feeling that the origins of the method amount at least to Kingman's article [260] in queueing theory, a reference that seems to have been previously overlooked. A very nice and powerful generalization of the kernel method is presented in [100].

Example 5. Let us illustrate the kernel method on the simplest example, in relation with the ballot problem introduced in §1.4. Set $\mathfrak{S} = \{(1,1), (1,-1)\} = \{\nearrow, \searrow\}$ and denote by $M_{n,k}$ be the number of \mathfrak{S} -walks in \mathbb{N}^2 of length *n* that start at (0,0)and end at vertical altitude k. Let $M(x, y) = \sum_{n,k} M_{n,k} x^n y^k$. We will show that:

- (a) *M* obeys the functional equation $(y x(1 + y^2)) \cdot M(x, y) = y x \cdot M(x, 0)$. (b) *M* is algebraic, namely $M(x, y) = \frac{\sqrt{1 4x^2} + 2xy 1}{2x(y x(1 + y^2))}$.

The starting point is an obvious recurrence relation, together with initial conditions, that translate the enumerative problem.

(2)
$$M_{n+1,k} = M_{n,k-1} + M_{n,k+1}, \quad M_{0,0} = 1, \ M_{-1,k} = M_{n,-1} = 0 \text{ for } k, n \ge 0.$$

Multiplying the recurrence relation by $x^{n+1}y^{k+1}$, and summing over $n, k \in \mathbb{N}$ yields

$$y \cdot \left(M(x,y) - \underbrace{\sum_{k \ge 0} M_{0,k} y^k}_{M(0,y) = 1} \right) = y^2 x \cdot M(x,y) + x \cdot \left(M - \underbrace{\sum_{n \ge 0} M_{n,0} x^n}_{M(x,0)} \right),$$

which rewrites as the so-called *kernel equation*

(3)
$$(y - x(1 + y^2)) \cdot M(x, y) = y - x \cdot M(x, 0).$$

Observe that simple manipulations like setting y = 0 in (3) lead to tautologies. The *kernel method* consists in the following simple observation: let $y_0 \in \mathbb{Q}[[x]]$

be the power series root of $K = y - x(1 + y^2)$, the coefficient of M(x, y) in Eq. (3):

$$y_0 = \frac{1 - \sqrt{1 - 4x^2}}{2x} = x + x^3 + 2x^5 + 5x^7 + 14x^9 + \dots \in \mathbb{Q}[[x]].$$

(One recognizes the generating function of Catalan numbers $y_0 = \sum_{n>0} C_n x^{2n+1}$.)

Then, plugging $y = y_0$ into the kernel equation (3) delivers $M(x, \overline{0}) = y_0(x)/x$. This provides an alternative, algebraic, proof of the (non-strict version of the) ballot problem. Finally, plugging back this value into (3) proves (b):

$$M(x,y) = \frac{y - y_0}{K(x,y)} = \frac{\sqrt{1 - 4x^2 + 2xy - 1}}{2x(y - x(1 + y^2))}.$$

We will encounter more sophisticated uses of the kernel method in §2 and §3.

1.9. Lattice walks with small steps in the quarter plane. The next case by increasing level of complexity is the one of a cone obtained as the intersection of two half-spaces. Up to modifying the step set by a linear transformation, one may assume that the cone is the basic orthant $\mathfrak{C} = \mathbb{R}^d_+$. This reduction is illustrated in the picture below, where the simple (Pólya) walks in the 2-dimensional cone of opening $\pi/4$ are put in bijection with the Gouyou-Beauchamps walks in the quarter plane.



The power series expansions of many special functions in combinatorics and physics, including algebraic functions, are D-finite: they satisfy linear differential equations with polynomial coefficients, see §1.11 for definitions and main properties. For example, 60 % of the handbook [2] describe D-finite functions.

That generating functions for walks constrained to evolve in an orthant need not be algebraic, and not even D-finite, was first observed by Bousquet-Mélou and Petkovšek in [103]. Preliminary results in this direction had been obtained by the same authors in [323, 324, 102]. The first model of walks in the quarter plane for which the generating function was proved to be non-D-finite [103, §3] is the so-called *knight walks* model: these are walks confined to \mathbb{N}^2 that start from $p_0 = (1, 1)$ and take their steps in $\mathfrak{S} = \{(2, -1), (-1, 2)\}$. This surprising result was the starting point of a massive classification effort, initiated by Mishna [307, 308], intensified in a germinal work by Bousquet-Mélou and Mishna [101], and continued by many researchers [257, 19, 84, 254, 310, 85, 277, 90, 278]. The rest of this section is devoted to tell the story of this classification, with a viewpoint towards computerized proofs.

Before restricting our attention to the special but important case of walks with small steps in the quarter plane, let us mention two general criteria that contain sufficient conditions for D-finiteness of the full generating function A(t; x). One was obtained by Bousquet-Mélou in [94, §3]. (A combinatorial proof for the particular case of the length generating function A(t; 1) was given in [103, §2].)

THEOREM 6. Let $\mathfrak{C} = \mathbb{R}^2_+$ and let $\mathfrak{S} \subset \mathbb{Z} \times \{-1, 0, 1\}$ be symmetric with respect to the horizontal axis. Then $A(t; \mathbf{x})$ is D-finite, given by an explicit system of linear differential equations.

The other criterion, whose precise statement is too involved to be given here, was already mentioned in §1.4 in connection with the reflection principle. Its underlying idea (an algebraic version of the reflection principle) was discovered independently by Gessel and Zeilberger [205] and Biane [42]. Roughly, the result asserts the following: if the set of steps is left invariant by a finite Weyl group, if the cone where the walks are confined to is a corresponding Weyl chamber and if no allowed step can traverse the boundary of the cone, then the generating function A(t; x) is D-finite. The precise assumptions can be found in [205] and in [269, Th. 10.18.3]. The criterion then follows by combining [205, Th. 3] with results on D-finiteness of positive parts and constant terms such as [287] (see also §3 of this document).

From now on, we focus on small-step walks (or, nearest-neighbor walks) in the *quarter plane.* These are walks in the lattice \mathbb{Z}^2 , confined to the cone $\mathfrak{C} = \mathbb{R}^2_+$ (we will often say *confined to* \mathbb{N}^2), that start at $p_0 = (0,0)$ and use steps in a model \mathfrak{S} which is a fixed subset of $\{\swarrow, \leftarrow, \nwarrow, \uparrow, \nearrow, \rightarrow, \searrow, \downarrow\}$.

An example of a small-step walk for the model $\mathfrak{S} = \{ \swarrow, \leftarrow, \uparrow, \rightarrow, \searrow, \downarrow \}$, with length n = 45 and ending point (i, j) = (14, 2), is depicted below.



Let us denote by $f_{n;i,j}$ the number of walks of length *n* ending at (i, j). The full *counting sequence* $(f_{n;i,j})_{n,i,j}$ admits several interesting specializations:

• $f_{n;0,0}$, the number of walks of length *n* returning to origin ("excursions");

• $f_n = \sum_{i,j\geq 0} f_{n;i,j}$, the number of walks with prescribed length *n*.

As customary in combinatorics, to these enumeration sequences one attaches (univariate, or multivariate) power series, namely the complete generating function

$$F_{\mathfrak{S}}(t;x,y) = \sum_{n=0}^{\infty} \left(\sum_{i,j=0}^{\infty} f_{n;i,j} x^{i} y^{j} \right) t^{n} \in \mathbb{Q}[x,y][[t]],$$

and its corresponding univariate specializations:

- $F_{\mathfrak{S}}(t;0,0)$, the generating function of *excursions*; $F_{\mathfrak{S}}(t;1,1) = \sum_{n\geq 0} f_n t^n$, the *length generating function*;
- $F_{\mathfrak{S}}(t;1,0)$, resp. $F_{\mathfrak{S}}(t;0,1)$, the generating function of walks ending on the horizontal, resp. vertical, axis, also called boundary returns;
- " $F_{\mathfrak{S}}(t;0,\infty)$ " := $[x^0]$ $F_{\mathfrak{S}}(t;x,1/x)$, the generating function of walks ending on the diagonal x = y of \mathbb{N}^2 , also called *diagonal returns*.

The general questions addressed in §1.2 specialize to the quarter-plane setting as follows: Given the model S, what can be said about the generating function $F_{\mathfrak{S}}(t; x, y)$, resp. about the counting sequence $(f_{n;i,j})_{i,j,n}$, and their specializations? More precise sub-questions concern structures, explicit forms and asymptotics:

- Structures: is $F_{\mathfrak{S}}$ algebraic? Is it D-finite? None of them?
- Explicit forms: do $F_{\mathfrak{S}}(t; x, y)$ and $(f_{n;i,j})_{i,j,n}$ admit closed-form expressions?

• Asymptotics: what is the behavior of $(f_{n;0,0})_n$, and $(f_n)_n$ when $n \to \infty$? The emphasis will be put on how Computer Algebra can be used to give computational answers to these questions.

1.10. Small-step models of interest. Among the 2^8 models $\mathfrak{S} \subseteq \{-1, 0, 1\}^2 \setminus \{(0,0)\}$, some are trivial (e.g., if $\mathfrak{S} \subseteq \{\swarrow, \leftarrow, \nwarrow, \searrow, \downarrow\}$, then $F_{\mathfrak{S}}(t; x, y) \equiv 1$), others are intrinsic to the half-plane (therefore $F_{\mathfrak{S}}(t; x, y)$ is algebraic, cf. Theorem 4), others come in pairs by diagonal symmetry (if \mathfrak{S} and \mathfrak{S}' are symmetric with respect to the diagonal of \mathbb{N}^2 , then $F_{\mathfrak{S}}(t; x, y) \equiv F_{\mathfrak{S}'}(t; y, x)$), see Fig. 1.



FIGURE 1. Some discarded models: trivial; intrinsic to the half-plane; symmetric.

After discarding these cases, Bousquet-Mélou and Mishna [101] found that there are exactly 79 interesting distinct models of small-step walks in the quarter plane. They are represented in Fig. 2, and are grouped in two classes: 74 *non-singular models* (or *genus-1 models* in the terminology of [180]) and 5 *singular models* (or *genus-0 models*). Singular models are the ones for which walks never return to the origin, that is for which the excursions generating function is trivial $F(t;0,0) \equiv 1$.



FIGURE 2. The 79 models of small-step walks in the quarter plane: 74 non-sigular, 5 singular.

Among the 79 models, there are "special" ones, that are considered interesting enough and were enough studied to deserve names: Pólya: \Leftrightarrow ; Kreweras: \checkmark ; Gessel: \checkmark ; Gouyou-Beauchamps: \circlearrowright ; King: \bigotimes ; Tandem: \diamondsuit .



FIGURE 3. The most basic classes of power series, and their dependencies.

One objective is then to understand and classify all these 79 models according to the structural properties of their generating functions.

1.11. Classification of power series. Before stating the main results, we still need a few definitions on (univariate and multivariate) power series.

DEFINITION 7. Let $S(t) = \sum_{n=0}^{\infty} s_n t^n$ be a power series in $\mathbb{Q}[[t]]$. Then, S(t) is called

- algebraic if it is a root of a non-trivial polynomial $P \in \mathbb{Q}[t, T]$, i.e., P(t, S(t)) = 0;
- *transcendental if it is not algebraic;*
- *D*-finite (or holonomic) if it is satisfies a non-trivial linear differential equation $p_r(t)S^{(r)}(t) + \cdots + p_0(t)S(t) = 0$ with polynomial coefficients $p_i(t) \in \mathbb{Q}[t]$;
- hypergeometric if its coefficients sequence $(s_n)_n$ satisfies a non-trivial linear homogeneous recurrence of order 1 with polynomial coefficients in $\mathbb{Q}[n]$.

A very important class of hypergeometric series is that of *Gauss hypergeometric functions* $_2F_1$ with parameters $a, b, c \in \mathbb{Q}$, $c \notin -\mathbb{N}$, defined by

$${}_2F_1\begin{pmatrix}a & b \\ c & t\end{pmatrix} = \sum_{n=0}^{\infty} \frac{(a)_n(b)_n}{(c)_n} \frac{t^n}{n!},$$

where $(x)_n = x(x+1)\cdots(x+n-1)$ is the Pochhammer symbol.

This notion admits an obvious extension to the so-called *generalized hypergeo*metric function ${}_{p}F_{q}$ depending on p + 1 rational parameters appearing in the top Pochhammer symbols, and on q rational parameters on the bottom. For example, ${}_{3}F_{2}\begin{pmatrix}a & b & c \\ d & e \end{pmatrix} t = \sum_{n=0}^{\infty} \frac{(a)_{n}(b)_{n}(c)_{n}}{(d)_{n}(e)_{n}} \frac{t^{n}}{n!}$, where $a, b, c, d, e \in \mathbb{Q}$ and $d, e \notin -\mathbb{N}$.

The way these three important classes of power series (algebraic, D-finite, hypergeometric) are connected is illustrated in Fig. 3.

That hypergeometric series are D-finite is an immediate consequence of the simple fact that coefficient sequences of D-finite series are exactly *P-recursive sequences*, satisfying linear recurrences with polynomial coefficients [356].

That algebraic series are D-finite has been observed in 1827 by Abel [1, p. 287]. Cockle [145] gave an algorithm for the computation of such a differential equation of the minimal possible order, that Harley [227] called *differential resolvent*. The method was then rediscovered by Tannery [372, §17], see also [212, §2.4]. One of the applications of these differential equations is the efficient power series expansions of algebraic series: a linear differential equation translates into a linear recurrence, with the consequence that the number of operations required to compute the first N coefficients grows only linearly with N. This method has been popularized in the combinatorics community by Comtet [148] and studied from the complexity point

of view by Chudnovsky and Chudnovsky [136, 137], and more recently in [72].

Finally, understanding power series that are simultaneously algebraic and hypergeometric is an old and difficult question. Fuchs asked in 1866 [192] for a classi-question was solved in 1873 by Schwarz [349], who showed using geometric arguments (sphere tilings by spherical triangles) that, up to some normalization of the parameters, and apart from an explicitly given finite number of sporadic cases,

$${}_{2}F_{1}\begin{pmatrix} r & 1-r \\ \frac{1}{2} \end{pmatrix} t = \frac{\cos((1-2r) \cdot \arcsin(\sqrt{t}))}{\sqrt{1-t}}, r \in \mathbb{Q}$$

is the only family of algebraic $_2F_1$ functions. Building on work by Eisenstein [172, 231], Landau [282, 283] and Stridsberg [365], Errera [174] obtained an alternative arithmetic proof of Schwarz' result, which is more elementary and algorithmic. Assume w.l.o.g. that $a, b, c \in \mathbb{Q}$ such that $a, b, c - a, c - b \notin \mathbb{Z}$. Then Errera's criterion states that ${}_{2}F_{1}\begin{pmatrix}a&b\\c&\\ \end{pmatrix}t$ is algebraic if and only if for every *r* coprime with the denominators of a, b and c, either $\{ra\} \leq \{rc\} < \{rb\}$ or $\{rb\} \leq \{rc\} < \{ra\}$, where $\{x\}$ denotes the fractional part x - |x| of x. For instance, this allows to prove immediately that

- $\frac{{}_{2}F_{1}\left(-\frac{1}{2} \frac{1}{6} \mid 16t\right) 1}{{}_{2}t} = 1 + 2t + 11t^{2} + 85t^{3} + 782t^{4} + \cdots \text{ is algebraic,}$ ${}_{2}F_{1}\left(\frac{1}{12} \frac{5}{12} \mid 1728t\right) = 1 + 60t + 39780t^{2} + 38454000t^{3} + \cdots \text{ is transcen-}$

A generalization of this result, which completely solves Fuchs' question, was obtained by Beukers and Heckman in 1989 [40].

THEOREM 8. Let $\{a_1, \ldots, a_k\}$ and $\{b_1, \ldots, b_{k-1}, b_k = 1\}$ be two subsets of \mathbb{Q} , assumed disjoint modulo \mathbb{Z} . Let D be their common denominator. Then $_{k}F_{k-1}\begin{pmatrix}a_{1} & a_{2} & \cdots & a_{k} \\ b_{1} & \cdots & b_{k-1} \end{pmatrix}$ is algebraic if and only if $\{e^{2i\pi ra_j}, j \leq k\}$ and $\{e^{2i\pi rb_j}, j < k\}$ interlace on the unit circle for all $1 \leq r < D$ with gcd(r, D) = 1.

For instance, the following hypergeometric function [342], arising from Chebychev's work on the distribution of primes numbers [373]

$$\sum_{n} \frac{(30n)!n!}{(15n)!(10n)!(6n)!} t^{n} = {}_{8}F_{7} \left(\begin{array}{c} \frac{1}{30} & \frac{7}{30} & \frac{11}{30} & \frac{13}{30} & \frac{7}{30} & \frac{19}{30} & \frac{23}{30} & \frac{29}{30} \\ \frac{1}{5} & \frac{1}{3} & \frac{2}{5} & \frac{1}{2} & \frac{3}{5} & \frac{2}{5} \\ \frac{1}{5} & \frac{1}{3} & \frac{2}{5} & \frac{1}{3} & \frac{2}{5} & \frac{1}{3} \\ \end{array} \right) \left| 2^{14} & 3^{9} & 5^{5} & t \right)$$

is an algebraic power series. Indeed, for all $1 \le r < 30$ with gcd(r, 30) = 1, one obtains the picture in Fig. 4, where red circles that correspond to upper parameters of the $_{8}F_{7}$, are interlaced with blue circles that correspond to lower parameters.

Similar definitions for algebraicity and D-finiteness apply to multivariate power series. For instance, $S \in \mathbb{Q}[[x, y, t]]$ is *algebraic* if it is the root of a non-trivial polynomial $P \in \mathbb{Q}[x, y, t, T]$, and it is *D*-finite if the set of all partial derivatives of S spans a finite-dimensional vector space over Q(x, y, t), in other words if *S* satisfies a system



FIGURE 4. The Beukers-Heckman interlacing criterion [40] at work.

of linear partial differential equations with polynomial coefficients of the form

$$\sum_{i} a_i(t, x, y) \frac{\partial^i S}{\partial x^i} = 0, \quad \sum_{i} b_i(t, x, y) \frac{\partial^i S}{\partial y^i} = 0, \quad \sum_{i} c_i(t, x, y) \frac{\partial^i S}{\partial t^i} = 0.$$

As in the univariate case, multivariate algebraic series are D-finite [288].

The concept of hypergeometric series also admits extensions to several variables, but they are beyond the scope of the present text. One such generalization was introduced around 1988 by Gel'fand, Kapranov and Zelevinsky [198, 200, 201, 199, 169, 359] and is known as *GKZ-hypergeometric functions*, or *A-hypergeometric functions*. Let us just mention that Beukers [39] obtained a characterization of the class of algebraic GKZ-hypergeometric functions, that extends the interlacing criterion from [40].

1.12. Kreweras' walks. An interesting model in the world of quarter-plane walks is Kreweras' model $\mathfrak{S} = \{\downarrow, \leftarrow, \nearrow\}$. It is related to a version of the three-candidate ballot problem, more difficult than the one mentioned at the end of §1.4. Let *A*, *B*, *C* be candidates in an election, that receive *a*, *b*, *c* votes respectively. What is the probability p(a, b, c) that, throughout the counting of the ballots, *A* has at least as many votes as *B* and at least as many votes as *C*? This amounts to counting paths in \mathbb{Z}^3 from the origin to (a, b, c) that use only unit positive steps and that are confined to the cone $\{x_1 \ge \max(x_2, x_3) \ge 0\}$ of \mathbb{Z}^3 . It appears that the reflection principle does not apply here, contrary to the case of the *edge cone* $\{x_1 \ge x_2 \ge x_3 \ge 0\}$.

Equivalently, the question amounts to counting paths in the quarter plane for the model $\mathfrak{S} = \{\downarrow, \leftarrow, \nearrow\}$. In a long paper, Kreweras [270] obtained a closed-formula for p(a, b, c) as a binomial double-sum:

$$p(a,b,c) = 1 - \frac{b+c}{a+1} + \frac{1}{(a+1)(a+2)} \sum_{i=1}^{b} \sum_{j=1}^{c} {\binom{b}{i} \binom{c}{j} \binom{2i+2j-2}{2i-1}} / {\binom{i+j+a}{a+2}},$$

which simplifies to P(a, b, 0) = 1 - b/(a + 1) for the two-candidate ballot problem (cf. §1.4), and to a simple formula in the special case c = a:

(4)
$$p(a,b,a) = 2^{2b+1} \left(\frac{a!}{(a-b)!}\right)^2 \frac{(2a-2b+1)!}{(2a+2)!}.$$

The same problem was considered independently by Flatto and Hahn [190] in an applied probabilistic context (double queue that arises when arriving customers simultaneously place two demands handled independently by two servers).



FIGURE 5. The simple walk in the cones with angle 45° and 135°: Gouyou-Beauchamps and Gessel walks.

As a consequence of Eq. (4), Kreweras obtained the following result, which was reproved using various methods in [271, 317, 204, 94, 96, 31, 257, 85]. The last two references in this list provide two different computer-aided proofs. In what follows, we denote by $K(t; x, y) = F_{\mathfrak{S}}(t; x, y)$ the full generating function for Kreweras walks $\mathfrak{S} = \{\downarrow, \leftarrow, \nearrow\}$ in the quarter plane, and by K(t; 0, 0) the generating function for Kreweras excursions.

THEOREM 9 (Kreweras, [270]). The generating function K(t; 0, 0) is equal to (5)

$$_{3}F_{2}\begin{pmatrix} 1/3 & 2/3 & 1\\ 3/2 & 2 \end{pmatrix} = \sum_{n=0}^{\infty} \frac{4^{n} \binom{3^{n}}{n}}{(n+1)(2n+1)} t^{3^{n}} = 1 + 2t^{3} + 16t^{6} + 192t^{9} + \cdots$$

As a corollary of Theorem 9, the results in §1.11 (e.g., Theorem 8) imply that K(t; 0, 0) is an algebraic power series. In fact, much more is true:

THEOREM 10 ([190, 204, 96]). The full generating function K(t; x, y) for the Kreweras walks is algebraic.

In §2 we will sketch a computer-aided proof of this result [85] based on the guess-and-prove paradigm.

1.13. Gessel's walks. Probably the most difficult model of walks in the quarter plane is Gessel's model $\mathfrak{S} = \{ \nearrow, \swarrow, \leftarrow, \rightarrow \}$. In 2001, Ira Gessel formulated, in private conversations with colleagues (including Mireille Bousquet-Mélou, Doron Zeilberger and Guoce Xin), two conjectures equivalent to the following statements:

CONJECTURE 1. The generating function G(t; 0, 0) of Gessel excursions is equal to

$$_{3}F_{2}\begin{pmatrix} 5/6 & 1/2 & 1 \\ 5/3 & 2 & 1 \end{pmatrix} = \sum_{n=0}^{\infty} \frac{(5/6)_{n}(1/2)_{n}}{(5/3)_{n}(2)_{n}} (4t)^{2n} = 1 + 2t^{2} + 11t^{4} + 85t^{6} + \cdots$$

CONJECTURE 2. The full generating function G(t; x, y) is not D-finite.

Here, as for the Kreweras walks, we denoted by $G(t; x, y) = F_{\mathfrak{S}}(t; x, y)$ for $\mathfrak{S} = \{ \nearrow, \checkmark, \leftarrow, \rightarrow \}$ the full generating function for Gessel walks in the quarter plane, and by G(t; 0, 0) the generating function for Gessel excursions.

The genesis of Gessel's conjectures is related to his interest in finding examples of cones in \mathbb{Z}^2 for which the generating functions for the simple (Pólya's) walk would admit nice formulas. As discussed in §1.5, Pólya [329] first observed that there are exactly $\binom{2n}{n}^2$ simple excursions of length 2n in the plane \mathbb{Z}^2 , and that the full generating function is rational in that case. Still for the Pólya model, but now restricted to the half plane, resp. to the quarter plane, Arquès [17] proved that excursions of length 2n are counted by nice formulas: $\binom{2n+1}{n}C_n$ for $\mathbb{Z} \times \mathbb{N}$, and C_nC_{n+1} for \mathbb{N}^2 . Concerning the nature of the full generating function, it is

algebraic for the cone $\mathbb{Z} \times \mathbb{N}$ [102], and D-finite for the cone \mathbb{N}^2 [94]. Gouyou-Beauchamps [210] found a similar formula $C_n C_{n+2} - C_{n+1}^2$ for the number of simple excursions of length 2n in the cone with angle 45° (the first octant). The generating function for this cone is again D-finite [205]. It was thus natural to consider the cone with angle 135° , and this is what Gessel did. See [89] for more historical details.

1.14. Algebraic reformulation: solving a functional equation. Gessel's problem admits the following purely algebraic reformulation, which should be seen as a quarter-plane analogue of Equation (3) from Example 5. If $G(t; x, y) \in \mathbb{Q}[x, y][[t]]$ denotes the full generating function for Gessel walks in the quarter plane then a simple inclusion-exclusion reasoning represented pictorially in Fig. 6 implies that G(t; x, y) satisfies a functional equation called the *kernel equation*

(6)

$$G(t;x,y) = 1 + t\left(xy + x + \frac{1}{xy} + \frac{1}{x}\right)G(t;x,y) - t\left(\frac{1}{x} + \frac{1}{x}\frac{1}{y}\right)G(t;0,y) - t\frac{1}{xy}\left(G(t;x,0) - G(t;0,0)\right).$$



FIGURE 6. The functional equation for Gessel walks in the quarter plane, pictorially.

Moreover, G(t; x, y) is completely characterized by the functional equation (6): it is its unique solution in $\mathbb{Q}[x, y][[t]]$, and even in the ring $\mathbb{Q}[[x, y, t]]$. Therefore, the task is simply to solve equation (6).

Similarly, to any of the 79 models introduced in §1.10 is attached a very similar functional equation. Again, this equation merely reflects a step-by-step construction of quarter-plane walks, and is based on the most elementary decomposition: a walk is either the empty walk, or it is a shorter walk followed by a permissible step. This observation is naturally translated into a generating function equation using *the inventory* $\chi_{\mathfrak{S}}(x,y) := \sum_{(i,j) \in \mathfrak{S}} x^i y^j$, and *the kernel* $\mathfrak{K}_{\mathfrak{S}}(t;x,y) = xy(1 - t\chi_{\mathfrak{S}}(x,y))$. Note that for a non-trivial model with small steps the kernel is a polynomial. The decomposition is translated into the *kernel equation* (we omit the subscript \mathfrak{S}): (7)

 $\mathfrak{K}(t;x,y)F(t;x,y) = xy + \mathfrak{K}(t;x,0)F(t;x,0) + \mathfrak{K}(t;0,y)F(t;0,y) - \mathfrak{K}(t;0,0)F(t;0,0).$

Observe that the last term of the right-hand side occurs only if the step \swarrow belongs to the model \mathfrak{S} .

Following Zeilberger's terminology [395], the variables x and y are called *catalytic* for equation (7). (This means that one cannot simply set x = 0 or y = 0 in the equation to solve for F(t; x, 0) and F(t; 0, y) first.) The number of catalytic variables is related to the number of constraints imposed to the walk. The case of kernel equations with a single catalytic variable corresponds to uni-directional walks and it is well-understood, the solutions being always algebraic [102], see Theorem 4.

Classifying lattice walks in the quarter plane thus amounts to solving 79 such equations. In the remaining part of Section 1 we describe several classes of results in this direction that have been obtained using Computer Algebra tools.

1.15. Main results (I): algebraicity of Gessel walks. After an almost successful attempt in [257], Gessel's first conjecture was finally solved in 2009 by Kauers, Koutschan and Zeilberger in [254] using an extension of the guess-and-prove approach described in [257].

Theorem 11 ([254]). $G(t;0,0) = {}_{3}F_{2} \begin{pmatrix} 5/6 & 1/2 & 1 \\ 5/3 & 2 \end{pmatrix} | 16t^{2} \end{pmatrix}.$

This result implies in particular that G(t;0,0) is D-finite, but has no immediate implications concerning the D-finiteness of G(t; x, y). It came as a total surprise when Bostan and Kauers [85] proved that Gessel's second conjecture was false.

THEOREM 12 ([85]). The generating function G(t; x, y) for Gessel walks is algebraic.

Prior to this result, even the algebraicity of G(t; 0, 0) had been overlooked, even though the classical results recalled in §1.11 obviously apply. For instance, because of the alternative representation

(8)
$${}_{3}F_{2}\left(\begin{array}{cc|c} 5/6 & 1/2 & 1 \\ 5/3 & 2 \end{array} \middle| 16t^{2} \right) = \frac{1}{t^{2}} \left(\frac{1}{2} {}_{2}F_{1}\left(\begin{array}{cc|c} -1/6 & -1/2 \\ 2/3 \end{array} \middle| 16t^{2} \right) - \frac{1}{2} \right),$$

it is clear that algebraicity of G(t;0,0) could have been decided using Schwarz's classification, but it appears that, quite strangely, nobody recognized that the parameters (-1/6, -1/2; 2/3) actually fit to Case III of Schwarz's table [349].

The original discovery and proof of Theorem 12 was computer-driven, and used a guess-and-prove approach, based on *Hermite-Padé approximants*. This will be explained in more details in §2. Note that as a byproduct of this proof, an estimate on the size of the minimal polynomial of G(t; x, y) has been given: according to [85], that minimal polynomial has more than 10^{11} terms when written in dense (expanded) form, for a total size of $\approx 30 \text{ Gb}$ (!) Several *human proofs* of Theorem 12 have been discovered since the publication of [85]: the first one used complex analysis [86], the second one was purely algebraic [99], and the more recent one is probably the most elementary [32, 33]. These proofs also contain a proof of Theorem 11.

1.16. Main results (II): Explicit form for G(t; x, y). An interesting consequence of Theorem 12 is the following result, which contains a closed-formula for the full generating function G(t; x, y) of Gessel walks [85].

THEOREM 13 ([85]). Let $V = 1 + 4t^2 + 36t^4 + 396t^6 + \cdots$ be the unique root in $\mathbb{Q}[[t]]$ of

$$(V-1)(1+3/V)^3 = (16t)^2,$$

let $U = 1 + 2t^2 + 16t^4 + 2xt^5 + 2(x^2 + 83)t^6 + \cdots$ be the unique root in $\mathbb{Q}[x][[t]]$ of

$$x(V-1)(V+1)U^{3} - 2V(3x + 5xV - 8Vt)U^{2}$$
$$-xV(V^{2} - 24V - 9)U + 2V^{2}(xV - 9x - 8Vt) = 0,$$

and let $W = t^2 + (y+8)t^4 + 2(y^2 + 8y + 41)t^6 + \cdots$ be be the unique root in $\mathbb{Q}[y][[t]]$ of

$$y(1-V)W^3 + y(V+3)W^2 - (V+3)W + V - 1 = 0.$$

ALIN BOSTAN

	OEIS	\mathfrak{S}	Pol size	LDE siz	e Rec size		OEIS	\mathfrak{S}	Pol size	LDE size	Rec size
1	A005566	↔		(3, 4)	(2, 2)	13	A151275	\mathbb{X}	—	(5, 24)	(9, 18)
2	A018224	Х		(3, 5)	(2, 3)	14	A151314	\mathbb{X}	—	(5, 24)	(9, 18)
3	A151312	\mathbb{X}		(3, 8)	(4, 5)	15	A151255	ک		(4, 16)	(6, 8)
4	A151331	鋖		(3, 6)	(3, 4)	16	A151287	捡		(5, 19)	(7, 11)
5	A151266	Y		(5, 16)	(7, 10)	17	A001006	ί.	(2, 2)	(2, 3)	(2, 1)
6	A151307	₩		(5, 20)	(8, 15)	18	A129400	殺	(2, 2)	(2, 3)	(2, 1)
7	A151291	.₩7		(5, 15)	(6, 10)	19	A005558	\mathbf{X}	—	(3, 5)	(2, 3)
8	A151326	₩.		(5, 18)	(7, 14)						
9	A151302	\mathbb{X}		(5, 24)	(9, 18)	20	A151265	\leftarrow	(6, 8)	(4, 9)	(6, 4)
10	A151329	翜		(5, 24)	(9, 18)	21	A151278		(6, 8)	(4, 12)	(7, 4)
11	A151261	A		(4, 15)	(5, 8)	22	A151323	¥₹	(4, 4)	(2, 3)	(2, 1)
12	A151297	鏉		(5, 18)	(7, 11)	23	A060900	\mathbf{A}	(8, 9)	(3, 5)	(2, 3)

FIGURE 7. Models with D-Finite length generating function $F_{\mathfrak{S}}(t; 1, 1)$; sizes (order, degree) of the equations.

Then G(t; x, y) is equal to

$$\frac{\frac{64(U(V+1)-2V)V^{3/2}}{x(U^2-V(U^2-8U+9-V))^2} - \frac{y(W-1)^4(1-Wy)V^{-3/2}}{t(y+1)(1-W)(W^2y+1)^2}}{(1+y+x^2y+x^2y^2)t-xy} - \frac{1}{tx(y+1)}$$

Again, the original discovery and proof of this result was computer-driven. During the computerized proof, a few other remarkable facts have been noticed, namely that G(t; x, y) can be expressed using nested radicals; for instance the length generating function $G(t; 1, 1) = 1 + 2t + 7t^2 + 21t^3 + 78t^4 + \cdots$ reads

$$\begin{split} G(t;1,1) &= -\frac{1}{2t} + \frac{\sqrt{3}}{6t} \sqrt{H(t)} + \sqrt{\frac{16t(2t+3)+2}{(1-4t)^2H(t)}} - H(t)^2 + 3} \;, \end{split}$$
 where $H(t) &= \sqrt{1+4t^{1/3}(1+4t)^{2/3}/(1-4t)^{4/3}}. \end{split}$

Actually, the proof uses the minimal polynomials for G(t; x, 0) and G(t; 0, y) that were guessed and proved during the algebraicity proof. A striking feature of Theorem 13 is the relative simplicity of the closed-form expression, especially when compared to the size of the minimal polynomial of G(t; x, y). As in the case of Theorem 12, the result in Theorem 13 admits several recent *human proofs* [86, 99, 32, 33].

1.17. Main results (III): Models with D-Finite length generating function. The computer-driven approach that allowed Bostan and Kauers [84] to discover and prove the properties of the puzzling generating function for Gessel walks was used as soon as 2008 by the same authors to provide a (conjecturally) exhaustive list of models having (conjecturally) D-finite and algebraic generating functions. That resulted in an experimental classification synthesized in Fig. 7, which displays 23 models of walks in the quarter plane for which the length generating function F(t; 1, 1) was conjectured to be D-finite. The computerized discovery used again a guess-and-prove method, based on Hermite–Padé approximation. Details will be

	OEIS	S	algebraic?	asymptotics		OEIS	\mathfrak{S}	algebraic?	asymptotics
1	A005566	\Leftrightarrow	Ν	$\frac{4}{\pi} \frac{4^n}{n}$	13	A151275	\mathbf{X}	Ν	$\frac{12\sqrt{30}}{\pi} \frac{(2\sqrt{6})^n}{n^2}$
2	A018224	X	Ν	$\frac{2}{\pi} \frac{4^n}{n}$	14	A151314	₩	Ν	$\frac{\sqrt{6}\lambda\mu C^{5/2}}{5\pi}\frac{(2C)^n}{n^2}$
3	A151312	\mathbb{X}	Ν	$\frac{\sqrt{6}}{\pi} \frac{6^n}{n}$	15	A151255	ک	Ν	$\frac{24\sqrt{2}}{\pi} \frac{(2\sqrt{2})^n}{n^2}$
4	A151331	鋖	Ν	$\frac{8}{3\pi}\frac{8^n}{n}$	16	A151287	☆	Ν	$\frac{2\sqrt{2}A^{7/2}}{\pi}\frac{(2A)^n}{n^2}$
5	A151266	Y	Ν	$\frac{1}{2}\sqrt{\frac{3}{\pi}}\frac{3^n}{n^{1/2}}$	17	A001006	ξ¢,	Y	$\frac{3}{2}\sqrt{\frac{3}{\pi}}\frac{3^n}{n^{3/2}}$
6	A151307	₩	Ν	$\frac{1}{2}\sqrt{\frac{5}{2\pi}}\frac{5^n}{n^{1/2}}$	18	A129400	敎	Y	$\frac{3}{2}\sqrt{\frac{3}{\pi}}\frac{6^n}{n^{3/2}}$
7	A151291	.₩ ⁷ .	Ν	$\frac{4}{3\sqrt{\pi}}\frac{4^n}{n^{1/2}}$	19	A005558	X	Ν	$\frac{8}{\pi}\frac{4^n}{n^2}$
8	A151326	₩.	Ν	$\frac{2}{\sqrt{3\pi}} \frac{6^n}{n^{1/2}}$					
9	A151302	X	Ν	$\frac{1}{3}\sqrt{\frac{5}{2\pi}}\frac{5^n}{n^{1/2}}$	20	A151265	\checkmark	Y	$\frac{2\sqrt{2}}{\Gamma(1/4)}\frac{3^n}{n^{3/4}}$
10	A151329	翜	Ν	$\frac{1}{3}\sqrt{\frac{7}{3\pi}}\frac{7^n}{n^{1/2}}$	21	A151278		Y	$\frac{3\sqrt{3}}{\sqrt{2}\Gamma(1/4)}\frac{3^n}{n^{3/4}}$
11	A151261	ф.	Ν	$\frac{12\sqrt{3}}{\pi}\frac{(2\sqrt{3})^n}{n^2}$	22	A151323	×	Y	$\frac{\sqrt{2}3^{3/4}}{\Gamma(1/4)} \frac{6^n}{n^{3/4}}$
12	A151297	鏉	Ν	$\frac{\sqrt{3}B^{7/2}}{2\pi} \frac{(2B)^n}{n^2}$	23	A060900		Y	$\frac{4\sqrt{3}}{3\Gamma(1/3)}\frac{4^n}{n^{2/3}}$
	$A = 1 + \sqrt{2}, \ B = 1 + \sqrt{3}, \ C = 1 + \sqrt{6}, \ \lambda = 7 + 3\sqrt{6}, \ \mu = \sqrt{\frac{4\sqrt{6}-1}{19}}$								

FIGURE 8. Models with D-Finite length generating function $F_{\mathfrak{S}}(t;1,1)$: asymptotics of $f_n = [t^n]F(t;1,1)$. For models 11, 13 and 15, estimates only hold for even n; for odd n, the constants change into $\frac{18}{\pi}$, $\frac{144}{\sqrt{5}}$ and $\frac{32}{\pi}$ [305].

presented in Section 2. The labels used in column "OEIS" are taken from Sloane's On-Line Encyclopedia of Integer Sequences [354]. The columns "LDE size", resp. "Rec size", refer to the minimal-order homogeneous linear differential, resp. recurrence, equation satisfied by F(t;1,1); they contain the order of the equation, and the maximum degree of its polynomial coefficients. The "Pol size" column refers to the algebraicity or transcendence of F(t;1,1): cases marked "—" were conjectured transcendental, the other cases were conjectured algebraic and the bidegree of the minimal polynomial was displayed. For example, the generating function F(t;1,1) for Kreweras walks (A151265) satisfies a differential equation of order 4 with polynomial coefficients of degree 9 and an algebraic equation P(F(t;1,1),t) = 0 for a polynomial P(T,t) of degree 6 in T and 8 in t. The coefficient sequence of F(t;1,1) satisfies a recurrence equation of order 6 with polynomial coefficients of degree 4.

For cases 1–22, these conjectural results on D-finiteness, resp. algebraicity, were confirmed by human proofs[‡] obtained almost simultaneously with [84] by Bousquet-Mélou and Mishna [101], using an uniform approach that we will present in §3. We discussed the difficult case 23 (Gessel's model) in §1.15 and §1.16. Concerning the conjectural transcendence results, the first unified proof was given in [76] and it is computer-driven; this will be discussed in §1.20. The reference [76] also contains the first proof, again computer-driven, that the (differential / recurrence / algebraic) equations conjectured in [84] are indeed correct.

As a complement to the results contained in Fig. 7, Bostan and Kauers demon-

[‡]Apart from Kreweras' and Gessel's models 20 and 23, the D-finiteness of $F_{\mathfrak{S}}(t; x, y)$ also follows from: Theorem 6 for the symmetric models 1–16; the Gessel-Zeilberger formula [205] for the "Weyl chamber models" 17–19; [308, Th. 2.4] for the "reverse Kreweras model" 21. For the "doubly Kreweras model" 22, [101, Prop. 15] seems to contain the first proof of D-finiteness, and even of algebraicity.

strated that Computer Algebra tools are also able to produce conjectural expressions for the asymptotics of $f_n = [t^n]F(t;1,1)$. Their results are displayed in Fig. 8 and have been obtained using a combination of algorithmic tools, including Hermite– Padé approximation, constant recognition algorithms built on integer relation detection algorithms like LLL [285] and PSLQ [186], and convergence acceleration techniques [109, 110]. These results have been confirmed a few years later by human proofs by Melczer and Wilson [305], using the theory of analytic combinatorics in several variables [322]. (Partial results had been previously obtained by Fayolle and Raschel [183], Johnson, Mishna and Yeats [243], Duraj [164], Melczer and Mishna [303], Garbit and Raschel [196]).

1.18. The group of a model. In order to formulate more results on the classification of lattice walks in the quarter plane, we need to introduce an important concept, *the group of the walk*. To a small-step walk model \mathfrak{S} one attaches the *generating polynomial* (also called the *inventory*) $\chi_{\mathfrak{S}}(x, y) := \sum_{(i,j) \in \mathfrak{S}} x^i y^j$. This is a bivariate Laurent polynomial in $\mathbb{Q}[x, x^{-1}, y, y^{-1}]$, that can be decomposed along powers of *x*, resp. of *y*, as follows:

$$\chi_{\mathfrak{S}} = \sum_{(i,j)\in\mathfrak{S}} x^i y^j = \sum_{i=-1}^1 B_i(y) x^i = \sum_{j=-1}^1 A_j(x) y^j.$$

The basic, yet fundamental, observation is that $\chi_{\mathfrak{S}}(x, y)$ is left invariant under two rational transformations

$$\psi(x,y) = \left(x, \frac{A_{-1}(x)}{A_{+1}(x)}\frac{1}{y}\right), \quad \phi(x,y) = \left(\frac{B_{-1}(y)}{B_{+1}(y)}\frac{1}{x}, y\right),$$

and thus under any element of the group $\mathcal{G}_{\mathfrak{S}} := \langle \psi, \phi \rangle$ of birational transformations generated by ψ and ϕ . When it is finite, $\mathcal{G}_{\mathfrak{S}}$ is isomorphic to a dihedral group, since ψ and ϕ are involutions. This notion of group of a walk originates from a similar notion, introduced in a probabilistic context by Malyshev in the 1970s [293]. It was first formally imported in the combinatorial framework by Mishna [307, 308], who realized that the method used in one of Bousquet-Mélou's solutions of the Kreweras model [96, §2.3], the algebraic kernel method, can be used to solve all models with cardinality at most 3. This method is a variation of the classical kernel method: instead of canceling the kernel, it finds a group of actions which fixes the kernel, and which is then used to generate more functional equations that are finally combined together using an algebraic method similar to the reflection principle. Mishna [307, 308] showed that in the 23 models in Fig. 7, the group is finite, and she determined explicitly its cardinality, which appears to be either 4 (for models 1–16 with an axial symmetry), or 6 (for the models 17, 18, 20, 21, 22, with a diagonal or an anti-diagonal symmetry), or 8 (for the remaining models 19 and 23), see Fig. 9. In a subsequent joint paper, Bousquet-Mélou and Mishna [101] exploited this idea and managed to solve 22 out of the 23 models in Fig. 7. Their solution will be explained in §3.4.

Bousquet-Mélou and Mishna [101] proved in addition that for all the other 56 models, the group is infinite. Let us sketch their argument, since it is simple, beautiful and very similar to the one used in §1.21. It reduces the question of the infinitude of the group to a (non-)cyclotomy question. Similarly, the argument in §1.21 will reduce the question of non-D-finiteness to the same (non-)cyclotomy question for the *same* polynomials. (This coincidence, which apparently has not been noticed



FIGURE 9. *Examples of models with groups of orders 4, 6, 8 and* ∞ *, respectively.*

before, is not fortuitous, see §1.21.) The argument goes as follows. Assume that $\mathcal{G}_{\mathfrak{S}}$ is finite. Then, denoting by θ the composition $\psi \circ \phi$, the order of θ is finite. Using a Taylor expansion, it follows that for any point $(a, b) \in \mathbb{C}^2$ fixed by θ , the order of the Jacobian matrix Jac (θ) at (a, b) is finite, and in particular its two eigenvalues are roots of unity. Now, for all models with infinite group[§], there exists a fixed point of θ , and a multiple in $\mathbb{Q}[t]$ of the characteristic polynomial of Jac (θ) at that fixed point, that does not contain any cyclotomic factor. This proves that $\mathcal{G}_{\mathfrak{S}}$ is infinite.

At this point, we know that the finiteness of the group for some model implies the D-finiteness of the generating function for that model. One important remaining question is: is the converse true? Another important pending question is: in the Dfinite cases, are there any closed-form expressions for the generating functions? The next two subsections will bring answers and completely clarify the situation.

1.19. Main results (IV): explicit expressions for models 1–19. Models 20–23 in Fig. 7 admit full generating functions that are *algebraic*. Moreover, closed formulas exist for them. For the three models 20–22 related to the Kreweras model, such formulas are displayed in [101, §6]. The most difficult case among these four is model 23 (Gessel's), for which Theorem 13 provides a closed-form expression.

We now focus on models 1–19. The natural question is whether closed-form expressions also exist in these cases. This question has been recently answered in a positive way using Computer Algebra tools in [76]: $F_{\mathfrak{S}}$ is uniformly expressible using iterated integrals of hypergeometric $_2F_1$ expressions. More precisely, the following structure result, already conjectured in [84, §3.2], holds true. Note that a similar expression also appears in a related combinatorial context [77] for rook paths on a three-dimensional chessboard, see Theorem 35 in §3.1.2.

THEOREM 14 ([76]). Let \mathfrak{S} be one of the models 1–19 in Fig. 7. Then $F_{\mathfrak{S}}(t; x, y)$ is expressible as a finite sum of iterated integrals of products of algebraic functions in x, y, t and of expressions of the form ${}_2F_1\begin{pmatrix}a&b\\c&w(t)\end{pmatrix}$, where $c \in \mathbb{N}$ and $w(t) \in \mathbb{Q}(t)$.

Once again, the discovery and the proof of this result are computer-driven; no human proof is available yet. The proof is based, among other tools, on *creative telescoping*, an efficient algorithmic technique for the symbolic integration of multivariate functions. Details will be discussed in §3.

[§]Bousquet-Mélou and Mishna [101, §3] do so for the 51 non-singular models, but F. Chyzak [private communication] points out that the argument still works on some iterate of θ .

ALIN BOSTAN

	S	occurring $_2F_1$	w		S	occurring $_2F_1$	w
1	\Leftrightarrow	$_2F_1\left(\begin{array}{c c} \frac{1}{2}, \frac{1}{2} \\ 1 \end{array}\right)$	$16t^{2}$	11		$_2F_1\left(\begin{array}{cc} \frac{1}{2}, \frac{1}{2}\\ 1 \end{array}\right)$	$\frac{16t^2}{4t^2+1}$
2	Χ	$_{2}F_{1}\left(\begin{array}{c} \frac{1}{2}, \frac{1}{2} \\ 1 \end{array} \middle w \right)$	$16t^{2}$	12	₩	$_{2}F_{1}\left(\begin{array}{c} \frac{1}{4}, \frac{3}{4}\\ 1 \end{array}\right)$	$\tfrac{64t^3(2t+1)}{(8t^2-1)^2}$
3	\mathbf{X}	$_{2}F_{1}\left(\begin{array}{c}1\\4\\1\end{array}\right)$	$\frac{64t^2}{(12t^2+1)^2}$	13	\mathbf{X}	$_{2}F_{1}\left(\begin{array}{c} \frac{1}{4},\frac{3}{4}\\ 1 \end{array}\right)$	$\tfrac{64t^2(t^2+1)}{(16t^2+1)^2}$
4	鋖	$_{2}F_{1}\left(\begin{array}{c}1\\2\\1\end{array}\right)$	$\frac{16t(t+1)}{(4t+1)^2}$	14	×	$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\\1\end{array}\right)$	$\frac{64t^2(t^2+t+1)}{(12t^2+1)^2}$
5	Ŷ	$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\end{array}\right)$	$64t^{4}$	15		$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\\1\end{array}\right)$	$64t^{4}$
6	₩	$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\end{array}\right)$	$\tfrac{64t^3(t+1)}{(1-4t^2)^2}$	16	会	$_{2}F_{1}\left(\begin{array}{c}1\\4\\1\end{array}\right)$	$\tfrac{64t^3(t+1)}{(1-4t^2)^2}$
7		$_{2}F_{1}\left(\begin{array}{c} \frac{1}{2},\frac{1}{2}\\ 1 \end{array}\right)$	$\frac{16t^2}{4t^2+1}$	17	4	$_{2}F_{1}\left(\begin{array}{c}\frac{1}{3},\frac{2}{3}\\1\end{array}\right)$	$27t^{3}$
8	₩.	$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\end{array}\right)$	$\tfrac{64t^3(2t+1)}{(8t^2-1)^2}$	18	敎	$_{2}F_{1}\left(\begin{array}{c} \frac{1}{3}, \frac{2}{3}\\ 1 \end{array}\right)$	$27t^2(2t+1)$
9	X	$_{2}F_{1}\left(\begin{array}{c}1\\4\\4\end{array}\right)$	$\tfrac{64t^2(t^2+1)}{(16t^2+1)^2}$	19	₩.	$_{2}F_{1}\left(\begin{array}{c} \frac{1}{2},\frac{1}{2}\\ 1 \end{array}\right)$	$16t^2$
10	篾	$_{2}F_{1}\left(\begin{array}{c}1\\4\\1\end{array}\right w\right)$	$\tfrac{64t^2(t^2+t+1)}{(12t^2+1)^2}$				

FIGURE 10. Hypergeometric series occurring in explicit expressions for F(t; x, y). The $_2F_1$ are given up to contiguity and derivation, that is, up to integer shifts of the parameters.

The parameters *a*, *b*, *c* of the occurring $_2F_1$'s as well as the rational functions w(t) are explicitly given in Table 10. The full expressions of the generating functions F(t;0,0), F(t;0,1), F(t;1,0), F(t;1,1), F(t;x,0), F(t;0,y) and F(t;x,y) are too large to be displayed here, and are available on-line. It turns out by inspection that the involved hypergeometric functions have a very particular form: they are intimately related to elliptic integrals, namely to the complete elliptic integrals of first and second kinds,

$$K(k) = \int_0^{\pi/2} (1 - k^2 \sin^2 \theta)^{-1/2} d\theta = \frac{\pi}{2} {}_2F_1\left(\frac{1}{2}, \frac{1}{2} \middle| k^2\right),$$
$$E(k) = \int_0^{\pi/2} (1 - k^2 \sin^2 \theta)^{1/2} d\theta = \frac{\pi}{2} {}_2F_1\left(-\frac{1}{2}, \frac{1}{2} \middle| k^2\right).$$

For instance, for King walks 💥 (case 4), the length generating function is equal to

(9)
$$F(t;1,1) = \frac{1}{t} \int_0^t \frac{1}{(1+4x)^3} \cdot {}_2F_1\left(\frac{3}{2} \frac{3}{2} \left| \frac{16x(1+x)}{(1+4x)^2} \right| dx.$$

See §3.4 for a detailed presentation of this example. Alternatively, an expression of F(t; 1, 1) in terms of elliptic integrals is

$$F(t;1,1) = \frac{1}{t} \int_0^t \frac{1}{\pi(1+4x)^2 \sqrt{x(1+x)}} \cdot K'\left(\frac{4\sqrt{x(1+x)}}{1+4x}\right) dx.$$

22

The relationship to elliptic integrals appears to hold true in a far more general setting. Indeed, taking Theorem 14 as starting point, van Hoeij has checked that for *many* (more than 100) integer sequences $(a_n)_{n\geq 0}$ in the OEIS whose generating function $A(t) = \sum_{n>0} a_n t^n$ is both D-finite and convergent in a small neighborhood of t = 0, all second-order irreducible factors of the minimal-order linear differential operator annihilating A(t) are solvable either in terms of algebraic functions, or in terms of complete elliptic integrals. This surprisingly general feature, reminiscent of Dwork's conjecture mentioned in [84, §3.2], begs for a combinatorial explanation.

1.20. Main results (V): transcendence for models 1-19. As said before, models 20–23 in Fig. 7 admit full generating functions that are *algebraic*. What about the full generating function $F_{\mathfrak{S}}(t; x, y)$, and its combinatorially meaningful specializations $F_{\mathfrak{S}}(t;0,0), F_{\mathfrak{S}}(t;1,0), F_{\mathfrak{S}}(t;0,1), F_{\mathfrak{S}}(t;1,1)$ for the models 1–23? Computer algebra is able to answer this question.

THEOREM 15 ([76]). Let \mathfrak{S} be one of the models 1–19 in Fig. 7. Then for any $(\alpha, \beta) \in$ $\{(0,0),(1,0),(0,1),(1,1)\}$, the power series $F_{\mathfrak{S}}(t;\alpha,\beta)$ is transcendental, except in the following four cases:

- $\mathfrak{S} = 4$ (model 17) and $(\alpha, \beta) = (1, 1)$, $\mathfrak{S} = 4$ (model 18) and $(\alpha, \beta) \in \{(1, 0), (0, 1), (1, 1)\}$.

As a consequence, the power series $F_{\mathfrak{S}}(t;x,y)$, $F_{\mathfrak{S}}(t;x,0)$, and $F_{\mathfrak{S}}(t;0,y)$ are transcendental for all the 19 models. Additionally, the generating functions of the four algebraic cases are equal to:

- $F_{4}(t;1,1) = \frac{1}{2t^2} \left(1 t \sqrt{(1+t)(1-3t)} \right),$
- $F_{\text{XX}}(t;1,1) = \frac{1}{8t^2} \left(1 2t \sqrt{(1+2t)(1-6t)} \right),$ $F_{\text{XX}}(t;1,0) = F_{\text{XX}}(t;0,1) = \frac{1}{32t^3} \left((1-6t)^{3/2}(1+2t)^{1/2} 4t^2 + 8t 1 \right).$

Again, the proof of Theorem 15 is computer-driven and crucially relies on the use of several modern Computer Algebra algorithms. This will be discussed in §2.4.5.

Algebraicity/transcendence proofs were first considered in some isolated cases: for model 15, F(t; x, y) was proved transcendental by Mishna [308, Th. 2.5]; for model 17, Mishna [308, §2.3.3] and Bousquet-Mélou and Mishna [101, §5.2], showed that F(t; x, y) and F(t; 0, 0) are transcendental and that F(t; 1, 1) is algebraic; for model 18, F(t; 1, 1) was proved algebraic by Bousquet-Mélou and Mishna [101, $\{5.2\}$; for model 19, Bousquet-Mélou and Mishna [101, $\{5.3\}$] showed that F(t;0,0), F(t;0,1), F(t;1,0) and F(t;1,1) are transcendental. The first unified transcendence proof for F(t; x, y) applying to all 19 models is by Fayolle and Raschel [181, Theorem 1.1], although they attribute that result to Bousquet-Mélou and Mishna [101]. They actually proved more, namely that $F(t_0; x, y)$ is transcendental for each $t_0 \in$ $(0, \#\mathfrak{S}^{-1}]$, using the approach in [180, Chap. 4]. However, this result does not provide any transcendence information about specializations at $x, y \in \{0, 1\}$.

Note that, for all the 19 models, the excursions generating functions F(t;0,0)could alternatively be proved transcendental by an argument based on asymptotics, similar to the one in [90]: using results from [156], one can show that the coefficient of t^{12n} in F(t;0,0) grows like $\kappa \rho^n n^{\alpha}$ for $\alpha \in \{-3, -4, -5\}$, and this implies transcendence of F(t;0,0) by [188, Theorem D]. By contrast, note that this asymptotic argument is not sufficient to prove the transcendence of all the other transcendental specializations, as showed for instance by Fig. 8 in the case of F(t; 1, 1) for models



FIGURE 11. Rotations of a scarecrow: models with zero drift that have a non-D-finite generating function.

5–10, for which $\alpha = -1/2$ is not incompatible with algebraicity.

1.21. Main results (VI): non-D-finiteness for models with an infinite group. The last question in view of the complete classification of small step walks in the quarter plane concerns the 56 models with an infinite group. Among them, 5 models are singular; for them, a variant of the kernel method, called the *iterated kernel method* was used by Mishna and Rechnitzer [310] (for two models) and by Melczer and Mishna [301] (for all five models), who showed that the length generating function F(t; 1, 1), and thus also the full generating function F(t; x, y), are non-D-finite.

The remaining question concerns the 51 non-singular models with an infinite group: is the full generating function (and its specializations) still non-D-finite?

Computer Algebra is able to help proving the following result.

THEOREM 16 ([90]). Let $\mathfrak{S} \subseteq \{0, \pm 1\}^2$ be any of the 51 nonsingular step sets in \mathbb{N}^2 with infinite group $\mathcal{G}_{\mathfrak{S}}$. Then the generating function $F_{\mathfrak{S}}(t;0,0)$ of \mathfrak{S} -excursions is not *D*-finite. Equivalently, the excursion sequence $(f_{n;0,0})_{n\geq 0}$ does not satisfy any nontrivial linear recurrence with polynomial coefficients.

In particular, the full generating function $F_{\mathfrak{S}}(t; x, y)$ is not D-finite in the 51 cases, since D-finiteness is preserved by specialization [288]. This corollary had been already obtained by Kurkova and Raschel [277], but the approach in [90] is at the same time simpler, and delivers a more accurate information. This new proof only uses asymptotic information about the coefficients of $F_{\mathfrak{S}}(0, 0, t)$, and arithmetic information about the constrained behavior of the asymptotics of these coefficients when their generating function is D-finite. More precisely, [90] first makes explicit consequences of the general results by Denisov and Wachtel [156] in the case of walks in the quarter plane. This analysis implies that, when *n* tends to infinity, the excursion sequence $f_{n;0,0}$ behaves like $\kappa \cdot \rho^n \cdot n^{\alpha}$, where $\kappa = \kappa(\mathfrak{S}) > 0$ is a real number, $\rho = \rho(\mathfrak{S})$ is an algebraic number, and $\alpha = \alpha(\mathfrak{S})$ is a real number such that $c = -\cos(\frac{\pi}{1+\alpha})$ is an algebraic number. More precisely,

(10)
$$\rho := \chi(x_0, y_0), \qquad c := \frac{\frac{\partial^2 \chi}{\partial x \partial y}}{\sqrt{\frac{\partial^2 \chi}{\partial x^2} \cdot \frac{\partial^2 \chi}{\partial y^2}}}(x_0, y_0), \qquad \alpha := -1 - \pi / \arccos(-c),$$

where (x_0, y_0) is the unique solution in $\mathbb{R}^2_{>0}$ of the system $\frac{\partial \chi}{\partial x} = \frac{\partial \chi}{\partial y} = 0$.

Starting from the step set \mathfrak{S} , explicit real approximations for ρ , α and c can be determined to arbitrary precision. Moreover, exact minimal polynomials of ρ and c can be determined algorithmically, using tools from elimination theory, namely Gröbner bases [151]. A classical result in the arithmetic theory of linear differential equations [168, 12, 197] about the possible asymptotic behavior of an integer-valued, exponentially bounded D-finite sequence, states that if such a sequence grows like



FIGURE 12. The 9 models with a non-D-finite but D-algebraic generating function.

 $\kappa \cdot \rho^n \cdot n^{\alpha}$, then α is necessarily a *rational number*. For the 51 cases of nonsingular walks with infinite group, [90] proves that the constant $\alpha = \alpha(\mathfrak{S})$ is not a rational number. The proof amounts to checking that some explicit polynomials in Q[*t*] are not cyclotomic. This mirrors the proof of the infinitude of groups for the 51 models, sketched at the end of §1.18. The resemblance is not accidental: with the notations of §1.18, it is possible to prove that (x_0, y_0) is a fixed point for θ and that the characteristic polynomial of the Jacobian Jac(θ) at (x_0, y_0) is equal to $T^2 + (2 - 4c^2)T + 1$, which admits roots that are roots of unity if and only if $\alpha = -1 - \pi / \arccos(-c)$ is a rational number.

Example 17. Consider the three scarecrows models depicted in Fig. 11. For the first and the third, the approach sketched above shows that the excursions sequence $[t^n] F_{\mathfrak{S}}(t;0,0)$

1, 0, 0, 2, 4, 8, 28, 108, 372, ...

is asymptotically equivalent to $\kappa \cdot 5^n \cdot n^{\alpha}$, for $\alpha = -1 - \pi / \arccos(\frac{1}{4}) = -3.383396...$ The irrationality of α prevents $F_{\mathfrak{S}}(t;0,0)$ from being D-finite.

Let us note that a new line of research is currently under development: using a method based on Tutte invariants, Bernardi, Bousquet-Mélou and Raschel [32, 33] showed that for 9 of these 51 models, the generating function is nevertheless *D*-*algebraic*, i.e., it satisfies a system of polynomial (non-linear) differential equations. These models are represented in Fig. 12. In parallel, using differential Galois theory, Dreyfus, Hardouin, Roques and Singer [161] proved the hypertranscendence of the remaining 42 models.

1.22. Summary: Classification of 2D non-singular walks. By combining the previous results, we obtain the following classification theorem, which provides a complete characterization of the nonsingular small-step sets with D-finite generating function. Before stating the result, we introduce the notion of *orbit sum*, that will emerge in §3 in relation with the kernel method.

DEFINITION 18. The orbit sum of a quarter-plane model \mathfrak{S} with finite group $\mathcal{G}_{\mathfrak{S}}$ is the following polynomial in $\mathbb{Q}[x, x^{-1}, y, y^{-1}]$:

$$OS_{\mathfrak{S}} := \sum_{g \in \mathcal{G}_{\mathfrak{S}}} (-1)^g g(x) g(y),$$

where for $g \in \mathcal{G}_{\mathfrak{S}}$ we denote by $(-1)^g$ the sign of g, which is 1 if g is the product of an even number of generators ϕ and ψ , and -1 otherwise.

For example in the case of the simple walk OS $\Rightarrow x \cdot y - \frac{1}{x} \cdot y + \frac{1}{x} \cdot \frac{1}{y} - x \cdot \frac{1}{y}$.

A simple computation shows that for exactly the four models 20–23, the orbit sum is zero. E.g., for the Kreweras model:

$$OS_{\underbrace{y}} = x \cdot y - \frac{1}{xy} \cdot y + \frac{1}{xy} \cdot x - y \cdot x + y \cdot \frac{1}{xy} - x \cdot \frac{1}{xy} = 0.$$

We now state the main result of this memoir. Recall that the *drift of a model* \mathfrak{S} is defined as the sum of the vectors in \mathfrak{S} .

Тнеокем 19. Let $\mathfrak{S}\subseteq\{0,\pm1\}^2$ be any of the 74 nonsingular quarter-plane models in Fig. 2. The following assertions are equivalent:

(1) The full generating function $F_{\mathfrak{S}}(t; x, y)$ is D-finite;

(2) the excursions generating function $F_{\mathfrak{S}}(t;0,0)$ is D-finite;

(3) the excursions sequence $[t^{2n}] F_{\mathfrak{S}}(t;0,0)$ is $\sim K \cdot \rho^n \cdot n^{\alpha}$, with $\alpha \in \mathbb{Q}$;

(4) the group $\mathcal{G}_{\mathfrak{S}}$ is finite;

(5) \mathfrak{S} has either an axial symmetry, or zero drift and cardinality different from 5.

Moreover, under (1)–(5), the cardinality of $\mathcal{G}_{\mathfrak{S}}$ is equal to $2 \cdot \min\left\{\ell \in \mathbb{N}^* \mid \frac{\ell}{\alpha+1} \in \mathbb{Z}\right\}$. Still under (1)–(5), $F_{\mathfrak{S}}(t;x,y)$ is algebraic if and only if \mathfrak{S} has positive covariance $\sum_{(i,j)\in\mathfrak{S}} ij - \sum_{(i,j)\in\mathfrak{S}} i \cdot \sum_{(i,j)\in\mathfrak{S}} j > 0$ and if and only if $OS_{\mathfrak{S}} = 0$. In this case, $F_{\mathfrak{S}}(t;x,y)$ is

expressible using nested radicals.

Otherwise, $F_{\mathfrak{S}}(t; x, y)$ is expressible using iterated integrals of $_2F_1$ expressions.

Proof. Implication $(1) \Rightarrow (2)$ is easy; $(2) \Rightarrow (3)$ is highly non-trivial and follows the combination of a strong probabilistic result [156] and of a strong arithmetic result [168, 12, 197]; (3) \Rightarrow (4) is the core of the results in [90] discussed in §1.21; $(4) \Rightarrow (1)$ is a consequence of results in [101, 85]. The equivalence of (2) and (5) is read off the tables in Appendix A of [90]. Condition (5) might seem unnatural; its purpose is to eliminate the three rotations of the "scarecrow" model with step sets depicted in Fig. 11, which have zero drift and non-D-finite generating functions. Finally, the observation on the cardinality can be checked from the data [101, Tables 1–3].

The characterization of algebraicity in terms of covariance and drift follows by inspection using Theorem 15. The last assertion is Theorem 14. Π

The classification of walks with small steps in the quarter plane can then be summarized pictorially as follows:



1.23. Extensions and open questions. We conclude this first part of the document with some generalizations and some problems for future investigation.

Walks with unit steps in \mathbb{N}^2 . Although small step walks in the quarter plane are quite well understood by now, there remain some open problems. For example, it is still unknown whether the length generating function F(t; 1, 1) is non-D-finite for all 56 models with infinite group. On the other hand, a *unified proof* is still lacking for the correspondence finite group \leftrightarrow D-finite generating function.

Walks with unit steps in \mathbb{N}^3 . One direction of research concerns the classification

26

of lattice walks in higher dimension. For the moment, an extensive investigation of the case of small step walks in the octant \mathbb{N}^3 has been initiated in [60]. In this case, the notions of the group of a model and of the orbit sum can be mimicked on the 2D case. The first difficulty is the number of cases: there are $2^{3^3-1} \approx 67$ millions models, of which 11074225 models are inherently 3-dimensional (instead of 79 in dimension 2). The article [60] focuses on the 20 804 models that have at most six steps. Among them, 170 cases appear to have a finite group; in the remaining cases, experiments suggest that the group is infinite. Needless to add, Computer Algebra was of crucial help in this study. The full generating function has been proved D-finite in all the 170 cases, with the exception of 19 intriguing models for which the nature of the generating function still remains unclear. One of them (the fifth in the list below) is the 3D analogue of the Kreweras model.



This leaves an open question: are there 3D non-D-finite models with a finite group? If so, this would constitute a major difference with the 2D case. We have played with the 3D Kreweras model and we conjecture that its generating function is indeed non-D-finite. This is supported by the fact that two different computations suggest that the asymptotics of the sequence k_{4n} of 3D Kreweras excursions of length 4n (which starts 1, 6, 288, 24444, 2738592, 361998432, ...) grows like $k_{4n} \approx C \cdot 256^n / n^{3.3257570041744...}$, for some C > 0, and the exponent 3.3257570041744... does not appear to be a *simple* rational number.

Another difference with the case of quarter-plane walks is the disappearance of algebraic models. Certain models do admit algebraic specializations, but then the walks counted by these series do not use all steps of the model, and deleting the unused steps leaves a model of lower dimension. We conjecture that, apart from these degenerate cases, there is no algebraic series among the 3D octant models.

The study [60] can be summarized as follows.

 $\begin{aligned} 3\text{D octant models } \mathfrak{S} \text{ with } &\leq 6 \text{ steps: } 20804 \\ & |\mathcal{G}_{\mathfrak{S}}| < \infty: 170 \qquad |\mathcal{G}_{\mathfrak{S}}| = \infty: 20634 \text{ [162, Th. 1.3]} \\ & | \\ \text{orbit sum } \neq 0: 108 \qquad \text{orbit sum } = 0: 62 \quad \text{non-D-finite?} \\ & | \\ & \text{kernel method } 2\text{D-reducible: } 43 \quad \text{not } 2\text{D-reducible: } 19 \\ & | \\ & \text{D-finite } \quad \text{D-finite?} \end{aligned}$

These results have been recently extended in a computational *tour de force* by Bacher, Kauers and Yatchak [20] to *all 3D octant models*: they have found 170 models with $|\mathcal{G}_{\mathfrak{S}}| < \infty$ and orbit sum 0 (instead of 19 models found by [60]). Kauers and Wang [255] have determined the structure of the group of the models in all these cases, extending results previously obtained by Du, Hou and Wang [162].

Walks with weighted small steps in \mathbb{N}^2 . Another line of research concerns the classification of nearest neighbor walks in the quarter plane for models in which



FIGURE 13. Two interesting quadrant models with repeated steps. Both are D-finite, and model B is even algebraic. Note that with only one copy of the repeated step, none of these models would be D-finite (§1.21).

multiplicities are attached to each direction in the step set. The study has been initiated by Bostan, Bousquet-Mélou, Kauers and Melczer [60] during their classification of octant models, as it turns out that some 3D models can be reduced by projection to 2D models with multiplicities. Among the octant models, they have identified 14744 two-dimensional models with at most 6 steps, which yield by projection 527 distinct quadrant models with at most 6 (possibly repeated) steps. Among them, 118 models appeared to have a finite group, of which 95 have a non-zero orbit sum. For 94 of them, the kernel method establishes the D-finiteness of the full generating function, but for one of them (Model A in Fig. 13) Computer Algebra was needed. All the remaining 23 models with finite group and zero orbit sum have been proved algebraic. Among them, 22 can be reduced to a usual quarter-plane model with algebraic generating function, but for the last of them (Model B in Fig. 13) Computer Algebra was needed again. In some sense, models A and B in Fig. 13 are similar to the Gessel model, but much more difficult.

The study in [60] has been continued by Kauers and Yatchak [256], whose work also heavily relies on Computer Algebra. They carried out a systematic search over all the $4^8 = 65536$ models where each of the eight directions may have any of the four multiplicities 0, 1, 2, 3. Of these, 30 307 were found nontrivial and essentially different. Of these nontrivial models, 1457 turned out to be D-finite (of which 79 models are even algebraic). Of these, three models have a group of order 10, a cardinality that was not possible in the classical (unweighted) setting. Less surprisingly, the correspondence between *finite group* and *D-finite generating function* observed in [60] continues to hold in this weighted 2D context. One open question raised by this study is: does there exist for every $n \ge 2$ a quarter-plane model with multiplicities whose group has order 2n? This is true in a probabilistic context, but for a different notion of group [180]. In a very recent work, Courtiel, Melczer, Mishna and Raschel [150] push even further the investigation of weighted models.

Other extensions. There are many other questions on the combinatorics of lattice paths in the cones, and certainly Computer Algebra will have a word to say, at least for some of them. Counting walks in non-convex cones is currently under investigation: after the case of the *slit plane* [104, 93, 105, 389, 344], it is now the turn of the cone $\mathfrak{C} := \{(i, j) : i \ge 0 \text{ or } j \ge 0\}$ [99]. Also, walks with larger steps in the quadrant are currently under investigation [184, 61]. There are several challenges, among them to find (and to use!) a notion close to the group of a model, which was specific to small-step models. For instance, for $\mathfrak{S} = \{(0, 1), (1, -1), (-2, -1)\}$, Bostan, Bousquet-Mélou and Melczer show that such a notion exists, and allows to prove that $F_{\mathfrak{S}}(t; x, y)$ is D-finite, via the positive part representation:

$$xyF_{\mathfrak{S}}(t;x,y) = [x^{>0}y^{>0}]\frac{(x-2x^{-2})(y-(x-x^{-2})y^{-1})}{1-t(xy^{-1}+y+x^{-2}y^{-1})}.$$

2. Guess-and-Prove.

What is "scientific method"? Philosophers and non-philosophers have discussed this question and have not yet finished discussing it. Yet as a first introduction it can be described in three syllables: **Guess and test.**

Mathematicians too follow this advice in their research although they sometimes refuse to confess it. They have, however, something which the other scientists cannot really have. For mathematicians the advice is **First guess, then prove.**

G. Pólya [331].

In this second part of the document, we enter into more technical details related to the experimental mathematics methodology that was employed to discover and to prove an important part of the results presented in §1, notably related to the celebrated Gessel walks (§1.15, §1.16) and more generally to the classification of lattice path models with D-finite generating functions (§1.17, §1.23). The process of experimental mathematics is to discover mathematical phenomenona by observing them via computations, before formally proving them. The rigorous proving step may be human, in the spirit of classical mathematics, or itself computerized, in the spirit of the current memoir. One of the experimental mathematics paradigms that was intensively used in recent years in the lattice path combinatorics context is the so-called *guess-and-prove* approach. It was introduced in this combinatorial context in work by Bostan and Kauers [84, 85], but its roots can be found in Pólya's remarkable books [333, 332], who popularized it as a fruitful mathematical proof strategy. The power of the method is highly enhanced when used on a computer, in conjunction with fast Computer Algebra algorithms.

This enhancement could be called the *automated (or, algorithmic) guess-and-prove* approach, and it is the topic of the current section. The first half (the *guessing* part) of the approach is based on a "functional interpolation" phase, which consists in recovering equations starting from (truncations of) solutions. It is called *differential approximation* [224, 259], or *algebraic approximation* [106], depending on the type of equations to be reconstructed. For instance, differential approximation is an operation to get an ODE likely to be satisfied by a given approximate series expansion of an unknown function. This has been used at least since the 1970s by physicists [224, 221], and is a key stone in recent spectacular applications in experimental mathematics, such as [266]. Modern versions [345, 251, 230] are based on subtle algorithms for Hermite–Padé approximants [29]. The second half (the *proving* part) of the approach is based on fast manipulations (e.g., resultants and factorization) with exact algebraic objects (e.g., polynomials and differential operators).

2.1. Methodology for proving algebraicity and D-finiteness. We illustrate the general principles of the guess-and-prove method when applied to proving that, for some lattice path model \mathfrak{S} with small steps in the quarter plane, the full generating function $F_{\mathfrak{S}}(t; x, y)$ is D-finite or algebraic. Recall from §1.14 that the problem amounts to solving the kernel equation (7):

$$\Re(t; x, y)F(t; x, y) = xy + \Re(t; x, 0)F(t; x, 0) + \Re(t; 0, y)F(t; 0, y) - \Re(t; 0, 0)F(t; 0, 0),$$

where $\mathfrak{K}_{\mathfrak{S}}(t; x, y) = xy(1 - t\sum_{(i,j) \in \mathfrak{S}} x^i y^j)$ is the kernel polynomial.

The method can be decomposed into three main steps:

(S1) Data generation: one first computes a high order expansion of the power series $F_{\mathfrak{S}}(t; x, y)$;

- (S2) Conjecture: from the local information computed at Step (S1), one tries to guess a global information, namely a candidate for a polynomial, resp. for a system of linear differential equations, satisfied by $F_{\mathfrak{S}}(t; x, y)$; this is done by using algebraic, resp. differential, approximation;
- (S3) Proof: one rigorously certifies the output of Step (S2), by using (exact) computations on multivariate polynomials, and on linear differential equations with polynomial coefficients.

In practice, Steps (S1), (S2), (S3) are performed using efficient algorithms from Computer Algebra.

As it turns out, an important improvement from the complexity of computations viewpoint is to perform the guessing step (S2) on the *sections* $F_{\mathfrak{S}}(t;x,0)$ and $F_{\mathfrak{S}}(t;0,y)$ only. This is sufficient due to the kernel equation, since both algebraicity and D-finiteness are preserved by sums, products and specializations. This simplification is crucial, as equations for the sections are usually *much smaller* than equations for the full generating function.

In §2.2, §2.3 and §2.4 we take a closer look at Steps (S1), (S2) and (S3).

2.2. Step (S1): high order series expansions. The first step of the method is based on a very basic observation: the full counting sequence $(f_{n;i,j})$ satisfies a recurrence with constant coefficients

(11)
$$f_{n+1;i,j} = \sum_{(k,\ell)\in\mathfrak{S}} f_{n;i-k,j-\ell} \quad \text{for} \quad n,i,j \ge 0$$

with the initial conditions $f_{0;i,j} = \delta_{0,i,j}$ and $f_{n;-1,j} = f_{n;i,-1} = 0$. The recurrence simply translates the step-by-step construction of quarter plane walks with model \mathfrak{S} : a \mathfrak{S} -walk of length n finishing at (i, j) is obtained from a walk of length n - 1, followed by a step in \mathfrak{S} ; the initial conditions translate the quarter-plane constraint.

Notice that as in the case of the much simpler kernel equation (2), multiplying the recurrence (11) by $t^n x^i y^j$, summing over n, i, j, and using the initial conditions yields the kernel equation (7).

Example 20. For the Kreweras walks, where $k_{n;i,j}$ denotes $f_{n;i,j}$ for $\mathfrak{S} = \mathcal{G}$,



The recurrence (11) can be used to determine the value of $f_{n;i,j}$ for specific integers $n, i, j \in \mathbb{N}$. The inequality $f_{n;i,j} \leq \#\mathfrak{S}^n$ implies that $f_{n;i,j}$ is a non-negative integer whose bit size is at most O(n). Therefore, if $N \in \mathbb{N}$, the truncated power series $F_{\mathfrak{S}}(t; x, y) \mod t^N$ can be computed by a straightforward algorithm that uses $O(N^3)$ arithmetic operations and $\tilde{O}(N^4)$ bit operations. (We assume that two integers of bit-size N are multiplied in $\tilde{O}(N)$ bit operations using FFT [348]; here, the soft-O notation $\tilde{O}()$ hides logarithmic factors.) The memory storage requirement is proportional to N^3 . For the experiments made in [84], N = 1000 was chosen. With this choice, the computation of the $f_{n;i,j}$ becomes very time and memory consuming.

30

Example 21. For the Kreweras model, one obtains

$$\begin{split} K(t;x,y) &= 1 + xyt + (x^2y^2 + y + x)t^2 + (x^3y^3 + 2xy^2 + 2x^2y + 2)t^3 \\ &+ (x^4y^4 + 3x^2y^3 + 3x^3y^2 + 2y^2 + 6xy + 2x^2)t^4 \\ &+ (x^5y^5 + 4x^3y^4 + 4x^4y^3 + 5xy^3 + 12x^2y^2 + 5x^3y + 8y + 8x)t^5 + \cdots, \end{split}$$

from which the first terms of the length generating function K(t; 1, 1) are computed

$$K(t;1,1) = 1 + t + 3t^{2} + 7t^{3} + 17t^{4} + 47t^{5} + 125t^{6} + 333t^{7} + 939t^{8} + 2597t^{9} + 7183t^{10} + 20505t^{11} + 57859t^{12} + 163201t^{13} + 469795t^{14} + \cdots$$

To summarize, step (S1) is very simple mathematically, but the naive algorithm used for it is not satisfactory. Its weakness is that in order to compute an univariate series such as $F_{\mathfrak{S}}(t;1,1)$, or a bivariate series like $F_{\mathfrak{S}}(t;x,0)$, it needs to compute the trivariate series $F_{\mathfrak{S}}(t;x,y)$. An important problem is to accelerate this algorithm. Our suggestion is to devise a divide-and-conquer method based on eq. (18) on p. 38, in the spirit of the algorithms in [107, 108, 74, 67]. This would allow to compute the sections $F_{\mathfrak{S}}(t;x,0) \mod t^N$ and $F_{\mathfrak{S}}(t;0,y) \mod t^N$ in quasi-optimal time (i.e., almost linear in their size, up to logarithmic factors), from which $F_{\mathfrak{S}}(t;1,1) \mod t^N$ could be easily reconstructed using the kernel equation (7) evaluated at x = y = 1.

2.3. Step (S2): guessing equations. The purpose of the second step of the method is to *guess* (differential, or algebraic) equations for $F_{\mathfrak{S}}(t; x, y)$.

2.3.1. A first idea. A first, but crucial, simplification comes from the simple remark that the kernel equation (7) expresses the full generating function $F_{\mathfrak{S}}(t;x,y)$ as a linear combination with rational function coefficients in $\mathbb{Q}(x, y, t)$ of its sections $F_{\mathfrak{S}}(t;x,0)$, $F_{\mathfrak{S}}(t;0,y)$ and $F_{\mathfrak{S}}(t;0,0)$. Therefore, by closure properties of algebraic and D-finite functions [288], $F_{\mathfrak{S}}(t;x,y)$ is D-finite (resp., algebraic) if and only if its sections $F_{\mathfrak{S}}(t;0,y)$ and $F_{\mathfrak{S}}(t;0,0)$ are both D-finite (resp., algebraic).

Example 22. In terms of generating functions, the recurrence in Ex. 20 reads

(12)
$$(xy - (x + y + x^2y^2)t)K(t; x, y) = xy - xt K(t; x, 0) - yt K(t; 0, y).$$

In order to prove the D-finiteness, resp. the algebraicity, of K(t; x, y), it is enough to prove the D-finiteness, resp. the algebraicity, of its sections K(t; x, 0) and K(t; 0, y).

In some cases, this simplification is crucial; for instance, in the case of the Gessel model, the minimal polynomial of F(t; x, y) has a size of \approx 30Gb, while sizes of the minimal polynomials of the sections F(t; x, 0) and F(t; 0, y) are merely \approx 1Mb.

2.3.2. Guessing equations for the sections $F_{\mathfrak{S}}(t; x, 0)$ and $F_{\mathfrak{S}}(t; 0, y)$. At the end of Step (S1), we are reduced to performing the following *guessing* tasks.

Task 1 (*differential guessing*): Given the first N terms of $S = F_{\mathfrak{S}}(t; x, 0) \in \mathbb{Q}[x][[t]]$, search for a linear differential equation satisfied by S at precision N:

(13)
$$\mathcal{L}_{x,0}(S) = c_r(x,t) \cdot \frac{\partial^r S}{\partial t^r} + \dots + c_1(x,t) \cdot \frac{\partial S}{\partial t} + c_0(x,t) \cdot S = 0 \mod t^N.$$

Task 2 (*algebraic guessing*): Given the first *N* terms of $S = F_{\mathfrak{S}}(t;x,0) \in \mathbb{Q}[x][[t]]$, search for a polynomial equation satisfied by *S* at precision *N*:

(14)
$$\mathcal{P}_{x,0}(S) = c_r(x,t) \cdot S^r + \dots + c_1(x,t) \cdot S + c_0(x,t) \cdot 1 = 0 \mod t^N.$$

Here and below, we use the compact notation $\mathcal{P}_{x,0}$ for a trivariate polynomial in $\mathbb{Q}[T, t, x]$, and $\mathcal{L}_{x,0}$ for an operator in the Weyl algebra $\mathbb{Q}(x, t)\langle \partial_t \rangle$ of linear differential operators in $\partial_t = \frac{d}{dt}$ with rational function coefficients in $\mathbb{Q}(t, x)$.

tial operators in $\partial_t = \frac{d}{dt}$ with rational function coefficients in $\mathbb{Q}(t, x)$. We use the similar notation $\mathcal{L}_{0,y}(S')$ and $\mathcal{P}_{0,y}(S')$ for equations potentially satisfied by the other section $S' = F_{\mathfrak{S}}(t; 0, y) \in \mathbb{Q}[y][[t]]$.

The idea behind differential guessing is that if the given power series *S* happens to be D-finite, then for a sufficiently large *N*, a differential equation of type (13) (thus satisfied a priori only at precision *N*) will provide a differential equation which is really satisfied by *S* in $\mathbb{Q}[x][[t]]$ (i.e., at precision infinity). In other words, the (conjectural) D-finiteness of a power series can be eventually recognized using a finite amount of information. The same holds for the algebraic guessing.

Example 23 (continued). Using N = 80 terms of $K(t; x, 0) = F_{4}(t; x, 0)$, one can guess a linear differential operator of order 4, and degrees (14, 11) in (t, x):

$$\mathcal{L}_{x,0} = t^3 \cdot (3t-1) \cdot (9t^2 + 3t+1) \cdot (3t^2 + 24t^2x^3 - 3xt - 2x^2) \cdot (16t^2x^5 + 4x^4 - 72t^4x^3 - 18x^3t + 5t^2x^2 + 18xt^3 - 9t^4) \cdot (4t^2x^3 - t^2 + 2xt - x^2) \cdot \partial_t^4 + \cdots$$

such that $\mathcal{L}_{x,0}(K(t; x, 0)) = 0 \mod t^{80}$.

Similarly, one can guess a polynomial of degree (6, 10, 6) in (T, t, x)

$$\mathcal{P}_{x,0} = x^{6}t^{10}T^{6} - 3x^{4}t^{8}(x-2t)T^{5} + x^{2}t^{6}\left(12t^{2} + 3t^{2}x^{3} - 12xt + \frac{7}{2}x^{2}\right)T^{4} + \cdots$$

such that $\mathcal{P}_{x,0}(K(t; x, 0)) = 0 \mod t^{80}$.

Therefore, it is very likely that K(t; x, 0) verifies the linear differential equation $\mathcal{L}_{x,0}(K(t; x, 0)) = 0$ and the algebraic equation $\mathcal{P}_{x,0}(K(t; x, 0)) = 0$, but at this stage we only have experimental evidence, which is by no means a rigorous proof.

In Tasks 1 and 2, the unknowns c_i are (not simultaneously zero) polynomials in $\mathbb{Q}[x, t]$. If their degrees in t are bounded by some prescribed integer $d \ge 0$ such that (d + 1)(r + 1) > N, then a simple linear algebra argument shows that a differential equation of type (13), resp. an algebraic equation of type (14), should exist. On the other hand, if d, r and N are such that $(d + 1)(r + 1) \ll N$, then equation (13) and (14) translate into highly over-determined linear systems, which have no reason to possess non-trivial solutions, unless S really is D-finite, resp. algebraic.

All previous remarks also apply to any specialization of *S* to same value $x \in \mathbb{Q}$.

The pending question is: how to solve *efficiently* Tasks 1 and 2, given d, r, N? Obviously, both amount to solving linear algebra problems in size N over Q(x). More precisely, a candidate differential, resp. polynomial, equation of type (13), resp. (14), for S can be computed by Gaussian elimination. But the corresponding systems are not randomly dense linear systems. They possess a very special structure, that can be exploited algorithmically in several ways. First, instead of solving linear systems of size N over Q(x), it is better to use an evaluation-interpolation scheme: evaluate the system at several points x, solve the corresponding systems over Q, and recombine the results by interpolation. The evaluation and interpolation steps can be performed very efficiently [383, Chap. 10], especially at points in geometric progression [91, §5]. Second, instead of solving linear systems over Q, it is better to solve

several systems over finite fields \mathbb{F}_p using a modular approach: the linear algebra step is performed modulo several primes p, and the results are recombined over \mathbb{Q} via rational reconstruction based on an effective version of the Chinese remainder theorem. Again, this can be performed very efficiently [383, Chap. 10]. Third, instead of using Gaussian elimination for solving the linear systems over \mathbb{F}_p that arise from (13) and (14) by specialization and reduction, it is better to exploit their Toeplitz*like structure*: their matrices are obtained by concatenation of Sylvester-like blocks, that possess the Toeplitz property of diagonal invariance, see §2.5 for details. Said differently, equations (13) and (14) are particular instances of *Hermite-Padé approxi*mation problems, and can be solved very efficiently. More precisely, while Gaussian elimination in size N over \mathbb{F}_{p} has cubic arithmetic complexity in N, fast algorithms for Hermite-Padé approximation have quasi-linear complexity in N, see §2.5.3. Such sophisticated algorithms rely on fast (FFT-like) arithmetic for the polynomial ring $\mathbb{F}_{p}[t]$ [383, 43, 113] and for the Weyl algebra $\mathbb{F}_{p}[t]\langle\partial_{t}\rangle$ [217, 376, 55, 71, 30, 75, 377]. They are not needed for simple examples, but they become of crucial help in the treatment of examples of critical sizes, such as for the computations involved in Gessel's model, see Example 24.

In practical implementations, for a given precision N, one searches for equations of increasing order r = 1, 2, ..., and a corresponding degree $d \approx N/r$. If no differential equation like (13) is found, this definitely rules out the possibility that a differential equation of order r and degree d exists. However, this does not imply that the series at hand is not D-finite. It may still be that S satisfies a differential equation of order higher than r, or an equation with polynomial coefficients of degree exceeding d. In that case, one doubles the series precision N, and starts over.

Sometimes (see §2.3.3 and §2.4.5) one needs to obtain the minimal-order differential equation $\mathcal{L}_{min}(S) = 0$ satisfied by the given generating power series S. The choice (d, r) of the target degree and order does not necessarily lead to the minimal operator \mathcal{L}_{\min} . Worse, it may even happen that the number of initial terms N is not large enough to allow the recovery of \mathcal{L}_{\min} , while these N terms suffice to guess non-minimal order operators. The explanation of why such a situation occurs systematically was first given in [72] for the case of differential equations satisfied by algebraic functions: minimal-order differential equations are often cluttered with apparent singularities, which considerably increase the degree of their coefficients. Therefore, they require too many terms N of the series S, and this prevents, or slows down, the reconstruction of equations. Differential guessing can benefit from the calculation of non-minimal equations, by minimizing not the order but the total size of the output. These considerations are intimately related to the operation of desingularization [120, 121, 119, 126]. All in all, a good heuristic to get \mathcal{L}_{min} is to compute several non-minimal operators and to take their greatest common right divisor (gcrd); generically, the result is exactly \mathcal{L}_{min} .

Example 24. For Gessel walks, N = 1000 terms of $G(t; x, y) = F_{X}(t; x, y)$ are sufficient to guess

- a differential operator $\mathcal{L}_{x,0} \in \mathbb{Q}(x,t) \langle \partial_t \rangle$, of order 11 in ∂_t , bidegree (96,78) in (t, x), and integer coefficients of at most 61 digits
- a differential operator $\mathcal{L}_{0,y} \in \mathbb{Q}(y,t)\langle \partial_t \rangle$, of order 11 in ∂_t , bidegree (68,28) in (t, y), and integer coefficients of at most 51 digits

such that $\mathcal{L}_{x,0}(G(t;x,0)) = \mathcal{L}_{0,y}(G(t;0,y)) = 0 \mod t^{1000}$.

Here is the way this was done. For a *fixed value a*, and *modulo a fixed prime p*, several (non-minimal order) operators in $\mathbb{F}_p[t]\langle \partial_t \rangle$ for G(t;a,0) can be guessed by



FIGURE 14. Guessing differential operators for G(t; a, 0), for prime p and $a \in \mathbb{F}_p$: minimal-order operator (blue point above the hyperbola) obtained as gcrd of several non-minimal operators (blue points below the hyperbola). Points below the hyperbola correspond to operators obtainable by Hermite-Padé approximation with 1000 terms.

Hermite-Padé approximation using 1000 terms of G(t; a, 0). Some of them are represented by the blue points below the hyperbola in Figure 14, e.g., one of them has order 14 and degree 43 in t. However, interpolating from one of those an operator in $\mathbb{Q}[t, x] \langle \partial_t \rangle$ for G(t; x, 0) appears to be *computationally extremely expensive*. The reconstruction (w.r.t. x) becomes feasible (in reasonable degree 78) when applied to the *minimal-order* operators (represented by the blue point above the hyperbola), themselves obtained as gcrds in $\mathbb{F}_p(t) \langle \partial_t \rangle$ of several non-minimal operators. Note that without gcrds, the minimal-order operator could not have been found by Hermite-Padé approximation with only 1000 terms. Also note that guessing $\mathcal{L}_{x,0}$ naively by *undetermined coefficients* would have required solving a dense linear system of size 91956 with ≈ 1000 digits entries! As a historical note, the discovery in 2008 of $\mathcal{L}_{x,0}$ and $\mathcal{L}_{0,y}$ first led Bostan and Kauers [85] suspect that G(t; x, y) is D-finite.

Efficient implementation of differential and algebraic guessing procedures have been implemented in most computer algebra systems, see e.g., the Maple package gfun written by Salvy and Zimmermann [345], the Mathematica package Guess.m by Kauers [251], or the FriCAS package Guess written by Hebisch and Rubey [230].

2.3.3. Empirical certification of guesses. Confidence in guessed equations can be complemented by using various filters. Once discovered a differential equation (13) or an algebraic equation (14) that the power series *S* seems to satisfy, it is useful to inspect several properties of these equations, in order to provide more convincing evidence that they are correct. These properties have various flavors: algebraic, analytic and even arithmetic. If the candidate guessed equations pass these filters, this offers striking experimental evidence that they are not artefacts.

Algebraic sieve: High order series matching. The equations (13) and (14) are obtained starting from *N* coefficients of the power series *S*. They are therefore satisfied a priori only modulo t^N . One can compute more terms of *S*, say 2*N*, and check whether the same equations still hold modulo t^{2N} . If this is the case, chances increase that the guessed equations also hold at infinite precision.

Analytic sieve: Singularity analysis. For any $a \in \mathbb{N}$, the univariate power series $F_{\mathfrak{S}}(t;a,0)$ has integer coefficients and positive radius of convergence. Thus, if in addition it is D-finite, then it is a *G*-function [168]. General results by Katz and Honda [246, 235], and Chudnovsky [135] then imply that the minimal order differential operator for $F_{\mathfrak{S}}(t;a,0)$ needs to be Fuchsian (it admits only regular singularities, including at infinity) and its exponents at each singularity must be rational numbers. See [11, 117, 168] for more details on this topic.

Arithmetic sieve: *G*-functions and global nilpotence. Last, but not least, one may check an arithmetic property of the guessed differential equations by exploiting the fact that those expected to arise in our combinatorial context are very special. Indeed, by a theorem due to the Chudnovsky brothers [135], the minimal order differential operator $\mathcal{L} = \mathcal{L}_{\min}^{S} \in \mathbb{Q}[t] \langle \partial_t \rangle$ killing a *G*-function *S* enjoys a remarkable arithmetic property: \mathcal{L} is globally nilpotent. By definition, this means that for almost every prime number *p* (i.e., for all with finitely many exceptions), there exists an integer $\mu \ge 1$ such that the remainder of the Euclidean (right) division of $\partial_t^{p\mu}$ by \mathcal{L} is congruent to zero modulo *p* [235, 167]. From a computational view-point, a fine feature is that the nilpotence modulo *p* is checkable. If *r* denotes the order of \mathcal{L} , let $A_p(\mathcal{L})$ be the *p*-curvature matrix of \mathcal{L} , defined as the $r \times r$ matrix with entries in $\mathbb{Q}(t)$ whose (i, j) entry is the coefficient of ∂_t^{j-1} in the remainder of the Euclidean (right) division of ∂_t^{p+i-1} by \mathcal{L} . Then, \mathcal{L} is nilpotent modulo *p* if and only if the matrix $A_p(\mathcal{L})$ is nilpotent modulo *p* [167, 347]. Faster tests exist [92, 62, 63, 64].

This yields a fast algorithmic filter: as soon as we guess a candidate differential equation satisfied by a generating function which is suspected to be a *G*-function (e.g., by F(t;1,1)), we check whether its *p*-curvature matrix $A_p(\mathcal{L})$ is nilpotent, say modulo the first 50 primes for which the reduced operator $\mathcal{L} \mod p$ is well-defined. If $A_p(\mathcal{L})$ is indeed nilpotent modulo *p* for all those primes *p*, then the guessed equation is, with very high probability, the correct one. This arithmetic sieving can be pushed even further. A famous conjecture, attributed to Grothendieck [248, 249, 13], asserts that the differential equation $\mathcal{L}(S) = 0$ possesses a basis of *algebraic solutions* (over $\mathbb{Q}(t)$) if and only if $A_p(\mathcal{L})$ is zero modulo *p* for almost all primes *p*. Even if the conjecture is, for the moment, fully proved only in special cases [117] (notably for Picard-Fuchs equations [248]) one can use it as an oracle to detect whether a guessed differential equation has a basis of algebraic solutions.

Example 25 (continued). For Gessel walks, the guessed (order-11) operators $\mathcal{L}_{x,0}$ and $\mathcal{L}_{0,y}$ for $G(t; x, y) = F_{\mathcal{L}, \mathcal{L}}(t; x, y)$ pass all the preceding filters, including the one based on *p*-curvatures. More precisely, for randomly chosen prime number *p*, and $a, b \in \mathbb{F}_p$, both $\mathcal{L}_{a,0}$ and $\mathcal{L}_{0,b}$ right-divide the pure power $\partial_t^{11\cdot p}$ in $\mathbb{F}_p(x)\langle \partial_t \rangle$. These operators actually have a stronger property: they even right-divide ∂_t^p ; in other terms, they have zero *p*-curvature for all the tested primes *p*. This was the key observation in the discovery [85] that the trivariate generating function for Gessel walks is algebraic.

The reader may wonder why the authors of [85] did not try algebraic guessing first. The first reason is that they had no reason to suspect that G(t; x, y) is algebraic, since even the specialization G(t; 0, 0) was generally thought to be transcendental. The second reason is that more terms of G(t; x, y) are needed to recognize alebraicity (1200, instead of 1000, see below), and the power series expansion to such high orders is computationally very expensive both in time and memory.

Example 26 (continued). Still for Gessel walks, now using N = 1200 terms of $G(t; x, y) = F_{i, \forall}(t; x, y)$ is sufficient to guess annihilating polynomials for sections:

- $\mathcal{P}_{x,0} \in \mathbb{Z}[T, t, x]$ of degree (24, 43, 32), integer coefficients of at most 21 digits,
- $\mathcal{P}_{0,y} \in \mathbb{Z}[T, t, y]$ of degree (24, 44, 40), integer coefficients of at most 23 digits,

such that $\mathcal{P}_{x,0}(G(t;x,0)) = \mathcal{P}_{0,y}(G(t;0,y)) = 0 \mod t^{1200}$.

2.4. Step (S3): rigorous proof.

Guessing is good, proving is better. G. Pólya [333].

2.4.1. Basic idea. The third and last step of the guess-and-prove method (for a quarter-plane model \mathfrak{S} for which the first two steps are assumed to have succeeded) consists in rigorously proving that the candidate (guessed) equations are indeed correct. Roughly, the basic idea is the following. Assume that one guessed equations for $F_{\mathfrak{S}}(t; x, y)$ which admit a solution S(t; x, y) in some power series ring \mathfrak{R} , typically $\mathbb{Q}[[x, y, t]]$ or $\mathbb{Q}[x, x^{-1}, y, y^{-1}][[t]]$, in which the kernel equation (7) has a *unique solution*, namely $F_{\mathfrak{S}}(t; x, y)$. Then, using effective closure properties for algebraic and D-finite functions [288] enables to compute the same (algebraic, or differential) equations for both sides of the kernel equation (7) with $F_{\mathfrak{S}}(t; x, y)$ replaced by S(t; x, y), and to eventually prove that the identity

(15)

$$\Re(t; x, y)S(t; x, y) = xy + \Re(t; x, 0)S(t; x, 0) + \Re(t; 0, y)S(t; 0, y) - \Re(t; 0, 0)S(t; 0, 0)$$

holds in \Re , where $\Re_{\mathfrak{S}}(t; x, y) = xy(1 - t\sum_{(i,j) \in \mathfrak{S}} x^i y^j)$. By uniqueness, it follows that $F_{\mathfrak{S}}(t; x, y)$ and S(t; x, y) coincide, and thus $F_{\mathfrak{S}}(t; x, y)$ is indeed algebraic (or D-finite), since S(t; x, y) is so, by design.

In practice, contrary to this ideal scenario, equations for the full generating function are too big to be computed, at least in many interesting cases. As explained in §2.3, one only has access to guessed equations for the sections $F_{\mathfrak{S}}(t;x,0)$ and $F_{\mathfrak{S}}(t;0,y)$. In this case, a variant of the method is used, and it is based on the *reduced kernel equation*, see §2.4.3 below. But before going into this, let us illustrate the guess-and-prove philosophy on a simpler example.

2.4.2. Warm-up: algebraicity of Gessel excursions. Let us prove that the generating function G(t; 0, 0) of Gessel excursions is algebraic, by taking Theorem 11 as the starting point. In other words, let us prove the algebraicity of the power series

$$g(t) := G(\sqrt{t}; 0, 0) = \sum_{n=0}^{\infty} \frac{(5/6)_n (1/2)_n}{(5/3)_n (2)_n} (16t)^n.$$

Of course, one could appeal to a proof that relies on equation (8) and on Schwarz's classification [349] of algebraic $_2F_1$ s, or other methods discussed in §1.11, like the Beukers-Heckman criterion (Theorem 8). Compared to these proofs, the constructive proof given below has the advantage that it can be applied similarly in situations where no classification results are available.

The guess-and-proof method works as follows: first guess a polynomial P(t, T) in $\mathbb{Q}[t, T]$, then prove that *P* admits the power series $g(t) = \sum_{n=0}^{\infty} g_n t^n$ as a root,

36
where $g_n = \frac{(5/6)_n(1/2)_n}{(5/3)_n(2)_n} 16^n$. In more details, the proof decomposes into three main steps:

- 1. (Guessing) A suitable polynomial P (see below) can be *guessed* automatically from the first 100 terms of g(t) using the approach explained in §2.
- 2. (Uniqueness) By the implicit function theorem, this polynomial *P* admits a root $r(t) \in \mathbb{Q}[[t]]$ with r(0) = 1. Since P(T, 0) = T 1 has a single root in \mathbb{C} , the series r(t) is the unique root of *P* in $\mathbb{C}[[t]]$.
- 3. (Proof) $r(t) = \sum_{n=0}^{\infty} r_n t^n$ being algebraic, it is D-finite (§1.11), and thus its coefficients satisfy a recurrence with polynomial coefficients, which is

(16)
$$(n+2)(3n+5)r_{n+1}-4(6n+5)(2n+1)r_n=0, r_0=1.$$

Thus $r_n = \frac{(5/6)_n(1/2)_n}{(5/3)_n(2)_n} 16^n = g_n$, and g(t) = r(t) is algebraic.

The concrete computations can be performed for instance in Maple using the package gfun, which provides the commands algeqtodiffeq for the algebraic guessing task in Step 1, algeqtodiffeq for Abel's theorem in Step 3 and diffeqtorec for the conversion differential equation \rightarrow recurrence in Step 3. The result of the two lines[¶]

is the recurrence (16).

2.4.3. Algebraicity proofs for Kreweras and Gessel walks. We now sketch the last part of the guess-and-prove method for proving the algebraicity of the generating functions for Kreweras and for Gessel walks. We focus on the Kreweras case, since the computations are easier and most of the ideas are already present.

The proof follows the same principles as the one just explained in §2.4.2. The idea is to guess, then to certify annihilating polynomials. The main difference with the situation in §2.4.2 is that an explicit closed form expression is no longer available beforehand for the power series whose algebraicity needs to be proved. Instead, we only have implicit equations that define that series. The method has three steps, and consists in applying the basic idea explained in §2.4.1, with the major difference that we cannot afford guessing of equations for the full generating function. The first step produces a so-called *reduced kernel equation* for the sections F(t; x, 0) and F(t; 0, y). In the Kreweras case, the step set being symmetric with respect to the main diagonal, the generating function K(t; x, y) enjoys the property K(t; x, y) = K(t; y, x), which simplifies the kernel equation (7) to

(17)
$$(xy - (x + y + x^2y^2)t)K(t;x,y) = xy - xtK(t;x,0) - ytK(t;y,0).$$

The proof goes as follows (the corresponding computations, performed in Maple, are displayed in Fig. 15):

 $[\]P In all that follows, we have used the version 3.76 (July 2015) of the package gfun, available at http://perso.ens-lyon.fr/bruno.salvy/software/the-gfun-package/.$

```
# HIGH ORDER EXPANSION (S1)
> st,bu:=time(),kernelopts(bytesused):
> f:=proc(n,i,j)
 option remember;
    if i<0 or j<0 or n<0 then 0
    elif n=0 then if i=0 and j=0 then 1 else 0 fi
    else f(n-1,i-1,j-1)+f(n-1,i,j+1)+f(n-1,i+1,j) fi
  end:
> S:=series(add(add(f(k,i,0)*x^i,i=0..k)*t^k,k=0..80),t,80):
# GUESSING (S2)
> P:=subs(Fx0(t)=T,gfun:-seriestoalgeq(S,Fx0(t))[1]):
# RIGOROUS PROOF (S3)
> ker := (T,t,x) -> (x+T+x^2*T^2)*t-x*T:
> pol := unapply(P,T,t,x):
> p1 := resultant(pol(z-T,t,x),ker(t*z,t,x),z):
> p2 := subs(T=x*T,resultant(numer(pol(T/z,t,z)),ker(z,t,x),z)):
> normal(primpart(p1,T)/primpart(p2,T));
# time (in sec) and memory consumption (in Mb)
> trunc(time()-st),trunc((kernelopts(bytesused)-bu)/1000^2);
                                     7, 617
```

FIGURE 15. Algebraicity of Kreweras walks: a computerized proof in a nutshell.

1. (Reduced kernel equation) Plugging

$$y_0 = \frac{x - t - \sqrt{-4t^2x^3 + x^2 - 2tx + t^2}}{2tx^2}$$

= $t + \frac{1}{x}t^2 + \frac{x^3 + 1}{x^2}t^3 + \frac{3x^3 + 1}{x^3}t^4 + \frac{2x^6 + 6x^3 + 1}{x^4}t^5 + \dots \in \mathbb{Q}[x, x^{-1}][[t]],$

in (17) shows that U = K(t; x, 0) satisfies the *reduced kernel equation*

(18)
$$0 = x \cdot y_0 - x \cdot t \cdot U(t, x) - y_0 \cdot t \cdot U(t, y_0).$$

- 2. (Uniqueness) Eq. (18) has a *unique solution* in $\mathbb{Q}[[x, t]]$, namely U = K(t; x, 0).
- 3. (Proof) Candidate $\mathcal{P}_{x,0}(T,t,x)$ guessed in (23) admits a root H in $\mathbb{Q}[[t,x]]$. Resultant computations prove that U = H(t,x) also satisfies (18).
- By uniqueness, K(t; x, 0) coincides with H(t, x), which is algebraic.

In the case of Gessel walks the proof follows the same strategy, but several complications occur:

- the diagonal symmetry of the step set is lost, so $G(t; x, y) \neq G(t; y, x)$;
- G(t;0,0) occurs in (7) (because of the step \swarrow);
- guessed equations are $\approx 5\,000$ times bigger.

To bypass these difficulties, one ingredient of the solution proposed in [85] is to replace equation (18) by an equivalent system of two reduced kernel equations, and to make use of fast algorithms for manipulating algebraic series, inspired by the algorithms for sums and products of algebraic numbers, designed in [81]. For more details, we refer the reader to the original article [85].

38

2.4.4. D-finiteness proofs for models A and B in Fig. 13. Here we simply state two recent results that have been discovered and proved using the guess-and-prove strategy explained before. The first one, Theorem 27, is remarkable in that it is a (more difficult) analogue of Theorem 11 (former Gessel Conjecture 1). The simple formulas beg for a combinatorial proof, but for the moment no human proof at all is known for it.

THEOREM 27 ([60]). The generating function $E(t) = F_A(t;0,0) = F_B(t;0,0)$ of excursions for the quadrant models A and B in Fig. 13 is

$${}_{4}F_{3}\left(\frac{5}{6}\frac{1}{5}\frac{1}{2}\frac{1}{2}\frac{7}{4}\frac{7}{4}\right|27t^{2}\right) = \sum_{n\geq 0}\frac{6(6n+1)!(2n+1)!}{(3n)!(4n+3)!(n+1)!}t^{2n} = 1 + 3t^{2} + 26t^{4} + 323t^{6} + \cdots$$

It is algebraic of degree 6, root of the polynomial

$$\begin{aligned} &16t^{10}T^6 + 48t^8T^5 + 8(6t^2 + 7)t^6T^4 + 32(3t^2 + 1)t^4T^3 \\ &+ (48t^4 - 8t^2 + 9)t^2T^2 + (48t^4 - 56t^2 + 1)T + (16t^4 + 44t^2 - 1). \end{aligned}$$

A parametric expression of E(t) is $t^2E(t) = Z(1-6Z+4Z^2)$, where Z is the unique series in t with constant term 0 satisfying

$$Z(1-Z)(1-2Z)^4 = t^2.$$

The second result, Theorem 28, has two parts. The first part is remarkable in that it provides the first example of D-finiteness result of a (non-algebraic) quadrant model that is currently proved uniquely via computer algebra. The second part is remarkable in that it is a (more difficult) analogue of Theorem 12 (former Gessel Conjecture 2). Again, no human proof is known for these results.

THEOREM 28 ([60]). (a) The full generating function $F_A(t; x, y)$ is D-finite. (b) The full generating function $F_B(t; x, y)$ is algebraic, of degree 12. It satisfies

$$F_B(t;x,y) = \frac{xy - t(1+x^2)F_B(t;x,0) - t(1+y)F_B(t;0,y) + tF_B(t;0,0)}{(y - t(1+y)(1+x^2(1+y)))}$$

The sections $F_B(t; x, 0)$ and $F_B(t; 0, y)$ can be written in parametric form as follows. Let $T(t) = t + 4t^3 + 48t^5 + \cdots$ be the unique series in t with constant term 0 such that

$$T(1-4T^2) = t.$$

Let $S(t) = t + 5t^3 + 62t^5 + \cdots$ be the unique series in t with constant term 0 such that

$$S(1-S^2)^2 = t(1+S^2)^3.$$

Then $F_B(t; x, 0)$ has degree 12 and is quadratic over $\mathbb{Q}(x, S)$:

$$F_B(t;x,0) = \left(\frac{1+S^2}{1-S^2}\right)^3 \times \frac{x(1+6S^2+S^4)-2S(1-S^2)(1+x^2)-(x-2S+xS^2)\sqrt{(1-S^2)^2-4xS(1+S^2)}}{2x(1+x^2)S^2}$$

Let finally W(t, y) be the unique power series in t with constant term 0 such that

$$W(1 - (1 + y)W) = T^2.$$

Then $F_B(t; 0, y)$ *has degree* 6 *and is rational in* T *and* W:

$$F_B(t;0,y) = t^{-2}W(1 - 4T^2 - 2W).$$

Moreover, its coefficients are doubly hypergeometric:

$$F_B(t;0,y) = \sum_{n \ge i \ge 0} \frac{6(2j+1)!(6n+1)!(2n+j+1)!}{j!^2(3n)!(4n+2j+3)!(n-j)!(n+1)} y^j t^{2n}.$$

2.4.5. Transcendence proofs for D-finite models. We have seen that the guessand-prove paradigm can be successfully used to prove D-finiteness and algebraicity. The proofs are constructive by design: they internally construct (differential, or algebraic) equations. It might thus look surprising that guess-and-prove can also be used to prove transcendence, that is, lack of algebraic equations. The framework is as follows. Assume that $f \in \mathbb{Q}[[t]]$ is a D-finite power series for which some linear differential equation L(f) = 0 (not necessarily of minimal order) is known. For instance, this differential equation could have been produced itself by a guessand-prove process. The question is how to prove that f is transcendental? This is interesting especially in cases where all known transcendence criteria (such as those in [188]) fail to apply. Such cases do occur, as seen in §1.20 for the length generating function F(t; 1, 1) for models 5–10 in Fig. 8, for which the asymptotic behavior is not incompatible with algebraicity. For these models, one possible workaround uses the factorization patterns of the differential operators for F(t; 1, 1): the operators systematically factor as a product of an order-2 operator on the left, and several order-1 operators on the right, so that Kovacic's algorithm [268] can be used to prove transcendence in an uniform way [76].

But factorization of linear differential operators, although quite well studied in theory [217, 353, 112, 380] is computationally very expensive, or even infeasible in practice, when applied to operators of high orders. Such an example is provided by Model B in Fig. 13. By Theorem 28, its full generating function $F_A(t; x, y)$ is D-finite, and by Theorem 27 its excursions generating function $F_A(t; 0, 0)$ is even algebraic. A natural question is: is $F_A(t; x, y)$ algebraic, or transcendental? The answer is contained in the Theorem 29 below.

THEOREM 29 ([60]). $F_A(t; 1, 0) = 1 + t + 4t^2 + 8t^3 + 39t^4 + 98t^5 + 520t^6 + \cdots$ is transcendental. In particular, the full generating function $F_A(t; x, y)$ is transcendental.

The only available proof [60] uses the guess-and-prove method. It consists in computing the minimal-order operator \mathcal{L}_{\min}^{f} for $f = F_A(t; 1, 0)$ and checking that \mathcal{L}_{\min}^{f} admits logarithms in some local expansions, which in particular prevents algebraicity of f. The computation of \mathcal{L}_{\min}^{f} is inspired by [380, §9]. The main idea can be traced back at least to [319]; similar arguments are used in [139, 35] and [149, §2].

All in all, the argument may be viewed as a general technique for proving transcendence of D-finite power series^{\parallel}; it reduces the transcendence question to dif-

^{||}There exists an alternative algorithmic procedure based on [352], that allows in principle to answer this question [351]. It involves, among other things, factoring linear differential operators, and deciding whether a linear differential operator admits a basis of algebraic solutions. However, this procedure would have a very high computational cost when applied to our situation.

ferential guessing. In the case of $f = F_A(t; 1, 0)$, the proof consists in the following steps:

- 1. (D-finiteness) Discover and certify a differential equation \mathcal{L} for f(t) of order 11 and degree 73
- 2. (Local analysis) \mathcal{L} is Fuchsian and has a logarithmic singularity at t = 0
- 3. (Bounds) If $\operatorname{ord}(\mathcal{L}_{\min}^f) \leq 10$, then \mathcal{L}_{\min}^f has coefficients of degrees ≤ 580 4. (Guessing) Differential Hermite-Padé approximants rule out this possibility
- 5. (Conclusion) Thus, $\mathcal{L}_{\min}^{f} = \mathcal{L}$, and so *f* is transcendental.

The bounds in Step 3 are the mathematical heart of the proof: they are obtained by using the Fuchsianity of L, and by bounding the apparent singularities of factors of *L* via Fuchs' equality, cf. [379, §4.4.1] and [334, §20].

2.5. Inside the toolbox: Hermite-Padé approximants. We now have a quick closer look at what is hidden behind guessing: Hermite-Padé approximants.

2.5.1. Definition. Let \mathbb{K} be a field, typically \mathbb{Q} or a finite field \mathbb{F}_p for a prime p. Given a column vector of power series $\mathbf{F} = (f_1, \dots, f_n)^T \in \mathbb{K}[[t]]^n$ and an *n*-tuple of integers $\mathbf{d} = (d_1, \ldots, d_n) \in \mathbb{N}^n$, a Hermite-Padé approximant of type \mathbf{d} for \mathbf{F} is a row vector of polynomials $\mathbf{P} = (P_1, \dots, P_n) \in \mathbb{K}[t]^n \setminus \{0\}$ such that:

(1) $\mathbf{P} \cdot \mathbf{F} = P_1 f_1 + \dots + P_n f_n = O(t^{\sigma})$ with $\sigma = \sum_i (d_i + 1) - 1$,

(2) $\deg(P_i) \leq d_i$ for all *i*.

The integer σ is called the *order* of the approximant **P**, and **d** is called its *type*.

When n = 2, Hermite-Padé approximants are called *Padé approximants*, a notion intimately related to rational approximations and continued fractions. When $f_{\ell} = A^{(\ell-1)}$, resp. $f_{\ell} = A^{\ell-1}$, for some $A \in \mathbb{K}[[t]]$, we talk about *differential approxi*mation, resp. of algebraic approximation, which form the basis of the differential, resp. algebraic, guessing described in §2.3.2.

These concepts were initially studied by Hermite [234] and by Padé [320], and turned out to be very useful in irrationality and transcendence proofs. For instance they (or, variants of them) served to prove the transcendence of e [233] and of π [286], see also [290, 291, 37]. The Chudnovsky brothers [139, 138, 135] used Hermite-Padé approximants for irrationality and transcendence proofs for values of quite general D-finite functions. A spectacular recent success using such approximants is the proof [24] of the irrationality of infinitely many values of the zeta function at odd integers. In most of these works, arithmetic results are obtained using explicit closed-form expressions for approximants, highly based on the structure of the functions to be approximated.

Our need is different, of algorithmic nature: we need fast algorithms that compute Hermite-Padé approximants on generic inputs. Before showing how to do that, we start with a very basic example.

2.5.2. Worked example. Let us compute a Hermite-Padé approximant of type (1,1,1) for $(1,C,C^2)$, where $C(t) = 1 + t + 2t^2 + 5t^3 + 14t^4 + 42t^5 + O(t^6)$.**

This boils down to finding $\alpha_0, \alpha_1, \beta_0, \beta_1, \gamma_0, \gamma_1 \in \mathbb{Q}$ (not all zero) such that

 $\alpha_0 + \alpha_1 t + (\beta_0 + \beta_1 t)(1 + t + 2t^2 + 5t^3 + 14t^4) + (\gamma_0 + \gamma_1 t)(1 + 2t + 5t^2 + 14t^3 + 42t^4) = O(t^5).$

^{**}The perceptive reader recognized the first terms of the generating function for Dyck walks (§1.4, §1.8).

FIGURE 16. A toy guessing example: the generating function of the Catalan sequence is recognized to be algebraic by gfun using its first 5 terms. It is also recognized to be D-finite using a few more terms.

Identifying coefficients, this is equivalent to a homogeneous linear system:

1 0 0 0 0	0 1 0 0 0	1 1 2 5 14	0 1 1 2 5	1 2 5 14 42	$\begin{bmatrix} 0\\1\\2\\5\\14 \end{bmatrix} >$	$\times \begin{array}{ c c } & \alpha_0 \\ & \alpha_1 \\ & \beta_0 \\ & \beta_1 \\ & \gamma_0 \\ & & \ddots \end{array}$	$= 0 \iff$	$\begin{bmatrix} 1\\0\\0\\0\\0\\0 \end{bmatrix}$	0 1 0 0 0	1 1 2 5 14	0 1 1 2 5	1 2 5 14 42	$\times \begin{bmatrix} \alpha_0 \\ \alpha_1 \\ \beta_0 \\ \beta_1 \\ \gamma_0 \end{bmatrix}$	$= -\gamma_1$	$\begin{bmatrix} 0\\1\\2\\5\\14\end{bmatrix}$	
Lo	0	14	5	44	14]	γ_1		Lo	0	14	9	44	[10]			

By homogeneity, one can choose $\gamma_1 = 1$. The bottom-right 3×3 minor shows that one can take $(\beta_0, \beta_1, \gamma_0) = (-1, 0, 0)$. Finally, the other values are $\alpha_0 = 1$, $\alpha_1 = 0$.

Thus the searched approximant is (1, -1, t): this means that we have guessed the candidate $P = 1 - y + ty^2$ such that $P(t, C(t)) = 0 \mod t^5$. This kind of functionality is implemented in most Computer Algebra systems. For instance, Maple's package gfun [345] implements the commands seriestoalgeq and listtoalgeq for algebraic approximants, resp. seriestodiffeq and listtodiffeq for differential approximants.

2.5.3. Existence and quasi-optimal computation. The existence of Hermite-Padé approximants is guaranteed by a simple linear algebra argument: the undetermined coefficients of a potential approximant $\mathbf{P} = (\sum_{j=0}^{d_i} p_{i,j}t^j)_i \in \mathbb{K}[t]^n$ satisfy a linear homogeneous system with $\sigma = \sum_i (d_i + 1) - 1$ equations and $\sigma + 1$ unknowns. This proof is constructive and gives a first, naive, algorithm for the effective computation of Hermite-Padé approximants, of complexity $O(\sigma^{\omega})$, where $2 \leq \omega \leq 3$ denotes a feasible linear algebra exponent, that is a constant that governs the complexity of most operations on dense matrices with coefficients in \mathbb{K} [383, Ch. 12].

However, as can be seen on the example in §2.5.2, the linear system has a *Toeplitz-like structure*: its matrix is obtained by concatenation of Sylvester-like blocks, that possess the Toeplitz property of invariance along diagonals. There are better algorithms that are able to exploit this structure. For instance, a generalization of the Euclidean algorithm yields a fast algorithm, of quadratic complexity $O(\sigma^2)$ [350] with respect to the order of approximation σ , see also [29] and the references therein. There are even faster algorithms that achieve a complexity softly-linear in σ , namely $O(\sigma \log^2 \sigma)$. They are based on fast (FFT-based) polynomial multiplication [383,

Chap. 8], and they rely on a divide-and-conquer scheme. Some are direct [29], other use the artillery of the theory of matrices with small displacement rank [321, 83, 82].

Here we give a rough sketch of the structure of the superfast Beckermann-Labahn algorithm [29, §4], when applied to compute a Hermite-Padé approximant of type (d, ..., d) for $\mathbf{F} = (f_1, ..., f_n) \in \mathbb{K}[[t]]^n$. The two main ideas are: to compute *a whole matrix of approximants* instead of just one approximants; to use a strategy of divide-and-conquer with respect to the order of the approximant $\sigma = n(d + 1) - 1$. The algorithm proceeds as follows:

1. If σ is below some chosen threshold, then use the naive algorithm

- 2. Else:
 - (a) recursively compute $\mathbf{P}_1 \in \mathbb{K}[t]^{n \times n}$ s.t. $\mathbf{P}_1 \cdot \mathbf{F} = O(t^{\sigma/2})$, $\deg(\mathbf{P}_1) \approx \frac{d}{2}$
 - (b) compute the residue $\mathbf{R} \in \mathbb{K}[[t]]^{n \times n}$ s.t. $\mathbf{P}_1 \cdot \mathbf{F} = t^{\sigma/2} \cdot (\mathbf{R} + O(t^{\sigma/2}))^2$
 - (c) recursively compute $\mathbf{P}_2 \in \mathbb{K}[t]^{n \times n}$ s.t. $\mathbf{P}_2 \cdot \mathbf{R} = O(t^{\sigma/2})$, $\deg(\mathbf{P}_2) \approx \frac{d}{2}$
 - (d) return $\mathbf{P} := \mathbf{P}_2 \cdot \mathbf{P}_1$

By construction, $\mathbf{P} \cdot \mathbf{F} = O(t^{\sigma})$. The precise choices of degrees is a delicate issue, and is one of the most difficult technical parts in the correctness proof. From the complexity point of view, up to logarithmic factors, the total cost of the whole algorithm is concentrated into the one of a product of $n \times n$ polynomial matrices of degree $\approx \frac{d}{2}$, that is $\tilde{O}(n^{\omega}d)$ operations in K. For more details, the reader is referred to the original article [29].

2.6. Back to the exercise in §1.1. To finish this section, we come back to the problem stated at the very beginning of the memoir, and show how to apply the Hermite-Padé approximation in order to guess the answer. A rigorous proof will be given in §3.5. In what follows, \mathfrak{S} denotes the step set { $\uparrow, \leftarrow, \searrow$ }.

2.6.1. A recurrence relation for \mathfrak{S} -walks in $\mathbb{Z} \times \mathbb{N}$. Let us denote by $h_{n;i,j}$ the number of \mathfrak{S} -walks in $\mathbb{Z} \times \mathbb{N}$ of length *n* from (0,0) to (i,j).

The numbers $h_{n;i,j}$ satisfy the following recurrence:

 $h_{n;i,j} = \begin{cases} 0 & \text{if } j < 0 \text{ or } n < 0, \\ \mathbb{1}_{i=j=0} & \text{if } n = 0, \\ \sum_{(k,\ell)\in\mathfrak{S}} h_{n-1;i-k,j-\ell} & \text{otherwise.} \end{cases}$

The following Maple lines compute the first terms of the generating function *A* of the sequence $(a_n)_n = (h_{n;0,0})_n$ counting \mathfrak{S} -walks in $\mathbb{Z} \times \mathbb{N}$ that end at the origin:

```
> h:=proc(n,i,j)
option remember;
    if j<0 or n<0 then 0
    elif n=0 then if i=0 and j=0 then 1 else 0 fi
    else h(n-1,i,j-1) + h(n-1,i+1,j) + h(n-1,i-1,j+1) fi
end:
> A:=series(add(h(n,0,0)*t^n, n=0..30), t,30);
```

They produce the output

(19) $A = 1 + 3t^{3} + 30t^{6} + 420t^{9} + 6930t^{12} + 126126t^{15} + 2450448t^{18} + 49884120t^{21} + 1051723530t^{24} + 22787343150t^{27} + O(t^{30}).$

2.6.2. A recurrence relation for \mathfrak{S} -walks in \mathbb{N}^2 . Let us denote by $q_{n;i,j}$ the number of \mathfrak{S} -walks in \mathbb{N}^2 of length *n* from (0,0) to (i, j).

The numbers $q_{n;i,j}$ satisfy the same recurrence as $h_{n;i,j}$, but with different boundary conditions:

 $q_{n;i,j} = \begin{cases} 0 & \text{if } i < 0 \text{ or } j < 0 \text{ or } n < 0, \\ \mathbbm{1}_{i=j=0} & \text{if } n = 0, \\ \sum_{(k,\ell) \in \mathfrak{S}} q_{n-1;i-k,j-\ell} & \text{otherwise.} \end{cases}$

The following Maple lines compute the first terms of the generating function *B* of the sequence $(b_n)_n = (\sum_k q_{n;k,k})_n$, counting \mathfrak{S} -walks in \mathbb{N}^2 that end on the diagonal:

```
> q:=proc(n,i,j)
option remember;
    if i<0 or j<0 or n<0 then 0
    elif n=0 then if i=0 and j=0 then 1 else 0 fi
    else q(n-1,i,j-1) + q(n-1,i+1,j) + q(n-1,i-1,j+1) fi
end:
> B:=series(add(add(q(n,k,k), k=0..n)*t^n, n=0..30), t,30);
```

The produced output is

(20) $B = 1 + 3t^{3} + 30t^{6} + 420t^{9} + 6930t^{12} + 126126t^{15} + 2450448t^{18} + 49884120t^{21} + 1051723530t^{24} + 22787343150t^{27} + O(t^{30}).$

We observe that $A = B \mod t^{30}$, but of course this is not yet a proof that A = B.

2.6.3. Guessing a closed form for the answer. From the first 30 terms of *A* and *B*, one can guess a nice formula for them. The following Maple lines show a way to do that. One could first guess a differential equation (seriestodiffeq), then convert it to a recurrence (diffeqtorec); here we appeal to a shortcut (seriestorec) which guesses directly a first-order recurrence for the coefficients of *A*. The series *A* is a hypergeometric function, that can be computed explicitly.

> gfun:-seriestorec(A, u(n))[1]; 2 2 {(-27 n - 81 n - 54) u(n) + (n + 9 n + 18) u(n + 3), u(0) = 1, u(1) = 0, u(2) = 0} > rsolve(%, u(n)): > A:=sum(subs(n=3*n, op(2,%))*t^(3*n), n=0..infinity); A := hypergeom([1/3, 2/3], [2], 27 t)

In other words, guessing predicts the following equality, equivalent to (1):

 $A(t) = B(t) = {}_{2}F_{1}\left(\frac{\frac{1}{3}}{2}\frac{\frac{2}{3}}{2}\right|27 t^{3}\right) = \sum_{n=0}^{\infty} \frac{(3n)!}{n!^{3}} \frac{t^{3n}}{n+1}.$

3. Creative telescoping.

Then we wish to show that $(n + 1)^2 b_{n+1} - n^2 b_{n-1} = (11n^2 + 11n + 3)b_n$, where $b_n = \sum_{k=0}^n F_{n,k}$ with $F_{n,k} = {n \choose k}^2 {n+k \choose k}$. Neither Cohen nor I had been able to prove this in the intervening 2 months. After a few days of fruitless effort the specific problem was mentioned to Don Zagier (Bonn), and with irritating speed he showed that indeed the sequence (b_n) satisfies this recurrence. We cleverly construct $B_{n,k} = (k^2 + 3(2n + 1)k - 11n^2 - 9n - 2)F_{n,k}$, with the motive that $B_{n,k} - B_{n,k-1} = (n + 1)^2 F_{n+1,k} - (11n^2 + 11n + 3)F_{n,k} - n^2 F_{n-1,k}$, and, O mirabile dictu, the sequence (b_n) does indeed satisfy the recurrence by virtue of the method of creative telescoping. A. van der Poorten [378].

3.1. Diagonals. Algebraic power series are D-finite (§1.11). An intermediate important class of power series is formed by *diagonals* of rational functions. All the examples of D-finite generating functions occurring in our combinatorial context of enumeration of walks appear to be diagonals, either directly (by their combinatorial definition), or indirectly (by the resolution method). The differential equations that they satisfy are special cases of Picard-Fuchs equations for periods of rational functions, and can be constructed algorithmically. A conjecture of Christol's [133] predicts even more: any D-finite power series $S \in \mathbb{Z}[[t]]$ with finite non-zero radius of convergence is the diagonal of a rational function.

In combinatorics, the importance of diagonals stems from the fact that numerous combinatorial constructions on generating functions (Hadamard products, constant terms or positive parts of Laurent series, ...) can be encoded as diagonals [357]. A classical result [287, 132] asserts that the diagonal of a rational function is D-finite (Theorem 33). A natural question is then: how to obtain algorithmically the linear differential equation satisfied by a diagonal? The problem can be reformulated in terms of the computation of a multiple integral with parameters taken on a cycle (§3), and can thus be attacked from a geometrical viewpoint [38, 154].



FIGURE 17. The diagonal of a bivariate power series (on the left) viewed as a residue (on the right).

DEFINITION 30. The diagonal of a multivariate power series $F \in \mathbb{Q}[[x_1, \ldots, x_n]]$

$$F = \sum_{i_1,\dots,i_n \ge 0} a_{i_1,\dots,i_n} x_1^{i_1} \cdots x_n^{i_n}$$

is the univariate power series

$$\operatorname{Diag}(F) = \sum_{i} a_{i,\dots,i} t^{i}.$$

3.1.1. Pólya's theorem. Almost a century ago, Pólya proved that diagonals of bivariate rational functions are algebraic [330]. Later, Furstenberg showed that the converse also holds [194]. Interestingly, Pólya's result becomes false for more than two variables. A simple example is provided by

Diag
$$\left(\frac{1}{1-x-y-z}\right) = \sum_{n\geq 0} {3n \choose n, n, n} t^n = {}_2F_1\left(\frac{1}{3} \frac{2}{3} \middle| 27t\right).$$

The transcendence of this series can be proved in various ways, for instance by using the asymptotics $\binom{3n}{n,n,n} = \frac{(3n)!}{n!^3} \sim 3^{3n} \frac{\sqrt{3}}{2\pi n}$ and [188, Theorem D], or by using the interlacing criterion from Theorem 8.

Pólya's result can be proved as follows. First, using the simple observation $\text{Diag}(F)(t) = [x^0] F(x, t/x)$, the diagonal of the rational function $F(x, y) \in \mathbb{Q}(x, y)$ is encoded as a complex integral using Cauchy's integral theorem (for some $\epsilon > 0$)

Diag (F) (t) =
$$[x^{-1}]\frac{1}{x}F\left(x,\frac{t}{x}\right) = \frac{1}{2\pi i}\oint_{|x|=\epsilon}F\left(x,\frac{t}{x}\right)\frac{dx}{x},$$

which in a second step can be evaluated using the residues theorem as a sum of residues (precisely: the residues of F(x, t/x)/x at its "small poles", having limit 0 at t = 0). Each of these residues are algebraic functions, and so is their sum Diag(F).

Example 31 (Dyck bridges). Let $\mathfrak{S} = \{\nearrow, \searrow\}$, and let B_n be the number of Dyck bridges (i.e. \mathfrak{S} -walks in \mathbb{Z}^2 starting at (0,0) and ending on the horizontal axis), of length *n*. Using a rotation counterclockwise by $\pi/4$, the integer B_n is seen to be the number of $\{\uparrow, \rightarrow\}$ -walks in \mathbb{Z}^2 from (0,0) to (n, n). This implies

$$B(t) = \sum_{n \ge 0} B_n t^n = \operatorname{Diag}\left(\frac{1}{1 - x - y}\right),$$

and the proof sketched above concludes:

$$B(t) = \frac{1}{2\pi i} \oint_{|x|=\epsilon} \frac{dx}{x-x^2-t} = \frac{1}{1-2x} \bigg|_{x=\frac{1-\sqrt{1-4t}}{2}} = \frac{1}{\sqrt{1-4t}} = \sum_{n\geq 0} \binom{2n}{n} t^n.$$

Rothstein-Trager resultant. It is not always possible to compute explicitly a closed-form expression for the poles and the residues, as we did in Example 31, for instance when the denominator of F(x, t/x)/x has degree more than 4. However, using resultants one can compute annihilating polynomials for them, and thus also for the diagonal. We show how this is done if F(x, t/x)/x has simple poles only.

Assume that \mathbb{K} is a field (in our case, \mathbb{K} is a placeholder for $\mathbb{Q}(t)$), and let $A, B \in \mathbb{K}[x]$ be such that deg $(A) < \deg(B)$, with B squarefree. Then the rational function F = A/B has simple poles only, and if F admits the partial fraction decomposition $F = \sum_{i} \frac{\gamma_i}{x - \beta_i}$, then the residue γ_i of F at the pole β_i equals $\gamma_i = \frac{A(\beta_i)}{B'(\beta_i)}$. Therefore, the residues γ_i of F are roots of the so-called *Rothstein-Trager resultant* [343, 374]:

$$R(t) = \operatorname{Res}_{x}(B(x), A(x) - t \cdot B'(x)),$$

which was originally introduced in computer algebra for the symbolic (indefinite) integration of rational functions.

A generalization of the Rothstein-Trager resultant to the case of multiple poles was given by Bronstein [111].



1, 2, 14, 106, 838, 6802, 56190, 470010, ...

FIGURE 18. Number d_N of diagonal 2D Rook paths from lower-left corner to $N \times N$ upper-right corner.

Example 32 (Diagonal Rook paths). Consider the following question [173, 147]: A Rook can move any number of squares horizontally or vertically. Assuming that the Rook moves right or up at each step, how many paths can the Rook take from the lower-left corner square to the upper-right corner square of an $N \times N$ chessboard?

Denote this number by d_N , see Fig. 18. The generating function of $(d_n)_n$ is

Diag(F) =
$$[x^0] F(x, t/x) = \frac{1}{2\pi i} \oint F(x, t/x) \frac{dx}{x}$$
, where $F = \frac{1}{1 - \frac{x}{1-x} - \frac{y}{1-y}}$

Then, Diag(F) is a sum of roots y(t) of the Rothstein-Trager resultant

- > F:=1/(1-x/(1-x)-y/(1-y)):
- > G:=normal(1/x*subs(y=t/x,F)):
- > factor(resultant(denom(G),numer(G)-y*diff(denom(G),x),x));

which is $t^2(1-t)(2y-1)(36ty^2-4y^2+1-t)$. By identifying which residues correspond to small poles, one concludes that the generating function of diagonal 2D

Rook paths is equal to the algebraic function $\frac{1}{2}\left(1+\sqrt{\frac{1-t}{1-9t}}\right)$.

The same method can be used for other walks of the same type [273].

Algorithmic questions related to the computation of algebraic equations for diagonals Diag(F) of bivariate rational functions F have been considered in connection with the enumeration of 1D lattice walks (bridges, excursions and meanders) by Banderier and Flajolet [26]. A general and efficient algorithm that computes an annihilating polynomial for Diag(F) was later proposed by Bostan, Dumont and Salvy [78, 80]; that solves positively the question of an effective version of Pólya's theorem. On the negative side, they showed that the minimal polynomial of Diag(F)has generically an exponential size with respect to the degree of the input rational function F. By contrast, linear differential equations satisfied by Diag(F) had been proved to have polynomial size [65]. This implies that for bivariate diagonals, the differential equations are the right data-structure, and not algebraic equations. It was shown in [78, 80] that the first N terms of generating functions for various 1D walks can be computed in quasi-linear complexity in N using this data-structure.

3.1.2. Lipshitz's theorem. Although diagonals of multivariate rational functions are not necessarily algebraic, they are still D-finite. In fact, much more holds.

THEOREM 33 (Lipshitz, [287]). *Diagonals of D-finite power series are D-finite*.

For *rational*^{††‡‡} power series, this result was previously obtained by Christol in an "elementary" way under a regularity assumption [130], and in the general case using quite involved geometric arguments [154, 131, 132], see also [289, 133, 134]. Very briefly, the argument is the following. First, as in the bivariate case, if $f \in$ $\mathbb{Q}(x_1, \ldots, x_n) \cap \mathbb{Q}[[x_1, \ldots, x_n]]$, the residue theorem allows to write (for some $\epsilon > 0$) (21)

$$\operatorname{Diag}(f)(t) = \frac{1}{(2\pi i)^{n-1}} \oint_{|x_1|=\dots=|x_{n-1}|=\epsilon} f\left(x_1,\dots,x_{n-1},\frac{t}{x_1\cdots x_{n-1}}\right) \frac{dx_1\cdots dx_{n-1}}{x_1\cdots x_{n-1}},$$

so that Diag(f)(t) is the (relative) period of a (family of) rational function(s) [245]. Its D-finiteness is then a consequence of the (highly non-trivial) finite-dimension property over $\mathbb{C}(t)$ of the de Rham cohomology for the complement of the variety in $\mathbb{A}_{\mathbb{C}(t)}^n$ defined by the equations denom $(f)(x_1, \ldots, x_n) = 0$ and $x_1 \cdots x_n = t$.^{§§} In more down-to-earth terms this proof guarantees, in a non-effective way, that repeated differentiation under the integral sign eventually produces a finite sequence of rational integrands that admit a linear combination with coefficients in $\mathbb{Q}(t)$ that becomes an exact differential. On the one hand, this geometric method allows access to more information about the minimal-order differential equation: it is Fuchsian and it has only rational exponents at each singularity (see [214, 215, 216] for an analytic proof and [246, 247] for an arithmetic proof^{¶¶}). On the other hand, it is non constructive. (See §3.2.3 for a way to make it constructive, using the so-called Griffiths-Dwork reductions.)

By contrast, Lipshitz' proof [287] is elementary and constructive. However, the algorithm behind its proof in highly inefficient. We demonstrate this using the example provided by the following combinatorial problem.

Example 34 (Diagonal 3D Rook paths). Consider the following question [173]: How many ways can a Rook move on a 3D chessboard from (0,0,0) to (N,N,N), where each step is a positive integer multiple of (1,0,0), (0,1,0), or (0,0,1)?

This is a 3D extension of the question in Example 32. Denote by D_N the number of diagonal 3D Rook paths of length *N*. The first terms of the sequence (D_N) are:

1, 6, 222, 9918, 486924, 25267236, 1359631776, 75059524392,...

The combinatorial problem readily translates into an algebraic problem. The generating function $\Delta(t) = \sum_{n\geq 0} D_n t^n$ of diagonal 3D Rook paths is the diagonal of

^{††}For *rational* series, Th. [287] had been conjectured by Stanley [356, §4(b)] and incompletely proved in [390, 202].

^{‡‡}For *algebraic* series, Th. [287] can be proved by reduction to the rational case [155, 3], for the price of doubling the number of variables.

^{§§}The finiteness proof needs Hironaka's resolution of singularities, among other things [219, 313, 228]. ^{¶¶}Katz first shows in [246, §5] that the minimal-order equation for a period is *globally nilpotent*; this result has been generalized by the Chudnovskys to any G-function [135], see also [168, Chap. VIII]. Then, Katz shows in [246, §13] that globally nilpotent operators are Fuchsian, with rational exponents; see also [235, 167] for a more elementary proof.

the rational function F(x, y, z) given by

$$\left(1 - \sum_{n \ge 1} x^n - \sum_{n \ge 1} y^n - \sum_{n \ge 1} z^n\right)^{-1} = \frac{(1 - x)(1 - y)(1 - z)}{1 - 2(x + y + z) + 3(xy + yz + zx) - 4xyz}.$$

A closed form for $\Delta(t)$ has been obtained by Bostan, Chyzak, van Hoeij and Pech [77], as an integral of a hypergeometric $_2F_1$. Its form is very similar to the ones in §1.19 for quarter-plane walks with small steps (models 1–19 in Fig. 7).

Тнеокем 35 ([77]).

$$\sum_{n} D_{n} t^{n} = 1 + 6 \cdot \int_{0}^{t} \frac{{}_{2}F_{1}\left(\frac{1/3}{2}\frac{2/3}{(1-4x)^{3}}\right)}{(1-4x)(1-64x)} dx.$$

The proof of Theorem 35 consists in first computing a differential equation for Diag(F), then in solving it in closed form, using algorithms in [77, 274, 240, 241]. In what follows, we focus on the first part, and describe the main steps on Lipshitz' proof when applied to prove the D-finiteness of Diag(F). The starting point is the following: If one is able to find a nonzero differential operator of the form

$$L(t, \partial_t, \partial_x, \partial_y) = P(t, \partial_t) + ($$
 higher-order terms in ∂_x and ∂_y $)$

that annihilates $G = \frac{1}{xy} \cdot F\left(x, \frac{y}{x}, \frac{t}{y}\right)$, then $P(t, \partial_t)$ annihilates Diag(F). This is explained by the sequence of equalities:

$$0 = [x^{-1}y^{-1}]L(G) = [x^{-1}y^{-1}]P(G) = P([x^{-1}y^{-1}]G) = P(\text{Diag}(F)).$$

The first equality comes from 0 = L(G), the second one is a consequence of $L(G) = P(G) + \partial_x(\cdot) + \partial_y(\cdot)$ and of the fact that derivatives w.r.t. *x* (resp. *y*) do not contain terms in x^{-1} (resp. in y^{-1}); the third equality is explained by the fact that *P* does only depend on *t*; the last one comes from $\text{Diag}(F) = [x^0y^0]F(x, y/x, t/y)$.

The remaining task is to show that such an *L* does indeed exist. To do this, a combinatorial argument is applied: By Leibniz's rule, the $\binom{N+4}{4}$ rational functions

$$t^i \partial_t^j \partial_x^k \partial_y^\ell(G), \quad 0 \leq i+j+k+\ell \leq N$$

are contained in the \mathbb{Q} -vector space of dimension $\leq 18(N+1)^3$ spanned by

$$\frac{t^i x^j y^k}{\operatorname{denom}(G)^{N+1}}, \quad 0 \leqslant i \leqslant 2N+1, \ 0 \leqslant j \leqslant 3N+2, \ 0 \leqslant k \leqslant 3N+2.$$

Thus, if $\binom{N+4}{4} > 18(N+1)^3$, then there exists $L(t, \partial_t, \partial_x, \partial_y)$ (resp. $P(t, \partial_t)$) of total degree at most N, such that LG = 0 (resp. P(Diag(F)) = 0).

At this point, note that N = 425 is the smallest integer satisfying $\binom{N+4}{4} > 18(N+1)^3$. Therefore, finding the operator *P* by Lipshitz' argument would require solving a linear system with 1,391,641,251 unknowns and 1,391,557,968 equations!***

^{***}By highly optimizing this argument [77] reduces the problem to a kernel computation of a polynomial matrix of size 8917×9139 , with entries in $\mathbb{Q}[x]$ of degree at most 37: these sizes are still too high to be dealt with in practice.

The conclusion is that Lipshitz's approach is not sufficient to obtain effectively differential equations for diagonals. This lack of efficiency motivates the creative telescoping algorithms described in the next section §3.

3.2. Creative telescoping for sums and integrals.

Toutes les relations mentionnées ci-dessus, y compris l'extraordinaire récurrence d'Apéry, sont retrouvées de manière systématique et automatique, et l'on dispose d'un outil qui permet de découvrir et de démontrer des identités d'un certain type. Le jour est sans doute proche où les formulaires classiques sur les fonctions spéciales seront remplacés par un logiciel d'interrogation performant.

P. Cartier [114].

Creative telescoping is an algorithmic paradigm for proving identities on multiple definite integrals and sums with parameters. This powerful computer algebra tool was introduced in the early 1990s by Zeilberger in the hypergeometric/hyperexponential setting [392, 393, 9, 394, 385], vastly generalized by Chyzak in the 2000s to the framework of holonomic functions [140, 141, 144, 142], and greatly enhanced and used in computerized proofs of difficult combinatorial applications by Koutschan in the 2010 [263, 262, 266, 264, 265, 229]. Since its birth, almost 30 years ago, the methodology of creative telescoping gained more and more popularity. As of 2017, it is one of the main topics in influential conferences like $ISSAC^{\dagger\dagger\dagger}$, where it has yearly its own dedicated special session.

We will give a brief account on creative telescoping, since several excellent texts already exist on this topic. We refer the reader to Chyzak's habilitation thesis [143], and to the surveys [264, 122].

Example 36. (Hypergeometric summation) Creative telescoping can automatically prove the following identities:

- $\sum_{k \in \mathbb{Z}} (-1)^k \binom{a+b}{a+k} \binom{a+c}{c+k} \binom{b+c}{b+k} = \frac{(a+b+c)!}{a!b!c!}$ (Dixon 1891 [158, 18]) $a_n = \sum_{k=0}^n \binom{n}{k}^2 \binom{n+k}{k}^2$ satisfies the recurrence (Apéry [16, 378])

$$(n+1)^{3}a_{n+1} = (2n+1)(17n^{2}+17n+5)a_{n} - n^{3}a_{n-1}$$

•
$$\sum_{k=0}^{n} {\binom{n}{k}}^2 {\binom{n+k}{k}}^2 = \sum_{k=0}^{n} {\binom{n}{k}} {\binom{n+k}{k}} \sum_{j=0}^{k} {\binom{k}{j}}^3$$
 (Strehl [363, 362, 364, 346])

Example 37. (Diagonals and integrals) Creative telescoping can automatically prove the following integral and diagonal evaluations:

• Diag $\frac{1}{(1-x-y)(1-z-u)-xyzu} = \sum_{n \ge 0} a_n t^n$ (Straub [15, 361])

$$\frac{1}{2\pi i} \oint \frac{(1+2xy+4y^2)\exp\left(\frac{4x^2y^2}{1+4y^2}\right)}{y^{n+1}(1+4y^2)^{\frac{3}{2}}} \, dy = \frac{H_n(x)}{\lfloor n/2 \rfloor!} \tag{Doetsch [159]}$$

•
$$\int_{0}^{+\infty} x J_1(ax) I_1(ax) Y_0(x) K_0(x) dx = -\frac{\ln(1-a^4)}{2\pi a^2}$$
 (Glasser-Montaldi [206])

^{†††}ISSAC, the International Symposium on Symbolic and Algebraic Computation, is the premier conference for research in symbolic computation and computer algebra http://www.issac-symposium.org.

where J_1 , Y_0 are Bessel functions, I_1 , K_0 are modified Bessel functions [2, Chap. 9], and H_n are Hermite polynomials [2, Chap. 22].

We briefly discuss the main principles of the Creative Telescoping paradigm for sums and integrals.

3.2.1. Creative Telescoping for sums. Let us explain the basic principle of the method on the simplest example possible. Denote by I_n the definite sum

$$I_n := \sum_{k=0}^n \binom{n}{k}.$$

We want to prove that $I_n = 2^n$. The idea is that **if** one writes Pascal's triangle identity under the "telescopic form":

$$\binom{n+1}{k} = \binom{n}{k} + \binom{n}{k-1} = 2\binom{n}{k} + \binom{n}{k-1} - \binom{n}{k},$$

then summation over k yields the recurrence

$$I_{n+1}=2I_n.$$

Taking into account the initial condition $I_0 = 1$ concludes the proof that $I_n = 2^n$.

More generally, assume that $(u_{n,k})$ is a bivariate sequence, and that one wants to "compute" its definite sum

$$F_n=\sum_k u_{n,k},$$

where "computing F_n " means, as in the example, finding a recurrence relation on it. The principle is the same as in the example. Let us denote by S_n and S_k the forward shift operators with respect to n and k, which act on bivariate sequences by the simple rules $S_n \cdot v_{n,k} = v_{n+1,k}, S_k \cdot v_{n,k} = v_{n,k+1}$. Then, **if** one knows recurrence operator $P(n, S_n)$ free of k and another recurrence operator $Q(n, k, S_n, S_k)$ such that

(22)
$$(P(n, S_n) - (S_k - 1)Q(n, k, S_n, S_k)) \cdot u_{n,k} = 0,$$

then the sum "telescopes", leading (under "nice" boundary assumptions) to the recurrence $P(n, S_n) \cdot F_n = 0$.

Observe that essentially the same idea was used in Lipshitz' proof of Theorem 33. The operator *P* is called a *telescoper* for $u_{n,k}$, while the operator *Q* is called a *certificate*.

The whole game is then to be able to produce an equality like (22). This is the objective of *creative telescoping*, a name seemingly coined by A. van der Poorten in his account of Apery's proof of the irrationality of $\zeta(3)$ [378], where Zagier is credited for having solved (22) for the sequence $u_{n,k} = {n \choose k}^2 {n+k \choose k}^2$. A decade later, it was Zeilberger who systematized, generalized, and quantified "Zagier's trick" in a series of articles [390, 393, 392, 394, 385]. The article [316] and the entire book [325] are devoted to popularize this summation framework.

Building on previous work by Fasenmyer [176] and Verbaeten [382], Wilf and Zeilberger [385] proved that the existence of a non-trivial solution (P, Q) of (22) is guaranteed if the summand sequence ($u_{n,k}$) is *proper hypergeometric*, i.e., of the form:

$$u_{n,k} = p(n,k)\alpha^n \beta^k \prod_{\ell=1}^L (a_\ell n + b_\ell k + c_\ell)!^{\pm 1},$$

where $p(n,k) \in \mathbb{Q}[n,k]$, where $a_{\ell}, b_{\ell}, c_{\ell} \in \mathbb{Z}$, and $\alpha, \beta \in \mathbb{Q}$. Moreover, they described an algorithm to compute such a pair (P, Q), similar in spirit to that of Theorem 33, and they extended these results to multiple sums and integrals. Although based on linear algebra only, the resulting algorithm suffered from too high a complexity and from long running times in implementations, just as in the case of Lipshitz's approach for diagonals (§3.1.2).

In parallel, Zeilberger came up with a fast algorithm for *definite hypergeometric* summation [392, 394], which is based on Gosper's decision algorithm for the *indefinite* summation of hypergeometric sequences [208]. Zeilberger actually realized that if the telescoper P in (22) were known beforehand, then the sequence $v_{n,k} = Q(n,k,S_n,S_k) \cdot u_{n,k}$, which satisfies $P(n,S_n) \cdot u_{n,k} = v_{n,k+1} - v_{n,k}$, could be obtained by simply calling Gosper's algorithm. To turn this remark into an algorithm, he explained that the simultaneous search for the coefficients of the telescoper $P(n,S_n)$ and for the rational function $v_{n,k}/u_{n,k}$ amounts to using a *parametrized variant* of Gosper's algorithm. Zeilberger named his fast algorithm *the method of creative telescoping*. It is implemented in many Computer Algebra systems. In Maple, a summation package SumTools[Hypergeometric] contains a command called Zeilberger.

Example 38 (Back to the SIAM flea). Keeping notation from §1.7, the probability $p_n(\varepsilon)$ of occupying the origin at step 2n is equal to $p_n(\varepsilon) = \sum_{k=0}^{n} U_{n,k}(\varepsilon)$, where

$$U_{n,k}(\varepsilon) := \binom{2n}{2k} \binom{2k}{k} \binom{2n-2k}{n-k} \left(\frac{1}{4}+\varepsilon\right)^k \left(\frac{1}{4}-\varepsilon\right)^k \frac{1}{4^{2n-2k}}$$

A linear recurrence for $(p_n(\varepsilon))_n$ can then be computed using Zeilberger's algorithm

whose output is

(23)
$$4(n+2)^2 S_n^2 + (2n+3)^2 (8\varepsilon^2 - 1) S_n + 16\varepsilon^4 (2n+3)(2n+1).$$

The probability $p(\varepsilon)$ is equal to $1 - \frac{1}{R_{\varepsilon}(1)}$, where $R_{\varepsilon}(t) = \sum_{n} p_{n}(\varepsilon)t^{n}$. Now (23) converts into a second-order differential equation satisfied by $R_{\varepsilon}(t)$, which is solved in terms of ${}_{2}F_{1}$'s, giving the answer announced in §1.7.

3.2.2. Creative Telescoping for integrals. A similar discussion applies to the case of parametrized integrals. Assume that H(t, x) is a bivariate function, and that one wants to "compute" its definite integral

$$I(t) = \oint_{\gamma} H(t, x) \, dx,$$

where "computing I(t)" means finding a linear differential equation satisfied by I(t). The principle from the discrete case applies to the continuous analogue. Let us denote by ∂_t and ∂_x the operators of partial derivation with respect to t and x, which act on bivariate functions by the simple rules $\partial_t \cdot f(x, t) = \frac{\partial f}{\partial t}$ and $\partial_x \cdot f(x, t) = \frac{\partial f}{\partial x}$.

Then, if one knows a differential operator $P(t, \partial_t)$ free of *x* and another differential operator $Q(t, x, \partial_t, \partial_x)$ such that

(24)
$$(P(t,\partial_t) - \partial_x Q(t,x,\partial_t,\partial_x)) \cdot H(t,x) = 0,$$

then the integral with respect to x "telescopes", leading (under "nice" assumptions on the integration domain) to the differential equation

$$P(t, \partial_t) \cdot I(t) = 0.$$

Again, the differential operator P is called a *telescoper* for the integrand H(t, x), while the operator Q is called a *certificate*.

Again, the whole game is then to be able to produce an equality like (22). First, the existence of a pair (P, Q) like in (24) is guaranteed if the integrand H(t, x) is *hyperexponential*, that is such that both $\frac{\partial H}{\partial t}$ and $\frac{\partial H}{\partial x}$ are rational functions in *t* and *x* [385]. Moreover, the computation of such a pair (P, Q) can be done in a slow fashion, *à la Lipshitz*, but also by an analogue of Zeilberger's fast creative telescoping, due to Alkmkvist and Zeilberger [9]. The algorithm from [9] is implemented for instance in Maple in the DEtools package, under the name Zeilberger.

Example 39 (Diagonal Rook paths, cont.). Using notation from Example 32, one needs to compute

Diag
$$(F) = [x^0] F(x, t/x) = \frac{1}{2\pi i} \oint F(x, t/x) \frac{dx}{x}$$
, where $F = \frac{1}{1 - \frac{x}{1-x} - \frac{y}{1-y}}$

A linear differential equation for Diag(F) can be computed using creative telescoping

> F:=1/(1-x/(1-x)-y/(1-y)): > G:=normal(1/x*subs(y=t/x,F)): > DEtools[Zeilberger](G, t, x, Dt)[1];

whose output is

$$(9t^2 - 10t + 1)\partial_t^2 + (18t - 14)\partial_t$$

and which can be solved explicitly, giving the answer $\text{Diag}(F) = \frac{1}{2} \left(1 + \sqrt{\frac{1-t}{1-9t}} \right)$.

3.2.3. Principle of Creative Telescoping for multiple integrals.. In the multivariate case, we restrict our attention to the integration of rational functions. This will be sufficient for our purposes in the combinatorial applications to lattice path enumeration. Let $H(t, \mathbf{x})$ be a rational function, where $\mathbf{x} = x_1, \ldots, x_n$ denote the integration variables, and t is the parameter left after integration. Let γ be an integration domain in \mathbb{C}^n , without boundary (more precisely, an *n*-cycle), on which H is assumed to take finite values only. The aim is to "compute" the parametrized integral, called *period*,

$$I(t) = \oint_{\gamma} H(t, \mathbf{x}) \mathrm{d}\mathbf{x}.$$

Example 40. The generating function for the Apéry numbers (sequence $(a_n)_n$ in Example 36) is the period of the rational integral [38, 41]

$$\frac{1}{(2\pi i)^3} \oint_{\gamma} \frac{dx \, dy \, dz}{1 - (1 - xy)z - txyz(1 - x)(1 - y)(1 - z)},$$

where γ is a suitable 3-cycle in \mathbb{C}^3 .

More generally, diagonals of rational functions are periods, due to equation (21).

It is a classical theorem that periods of rational integrals are D-finite; this generalizes Theorem 33. The corresponding linear differential equations are known under the name of *Picard-Fuchs* equations. They describe the variation of the family of periods with respect to the parameter of the family.

Particular cases of this theorem have been proved by Legendre (1825) and Kummer (1836) [272, §29], see also [213], for the periods of the complete elliptic integrals, and by Fuchs [193] and Picard [327] for the periods of hyperelliptic integrals and other abelian integrals on curves of arbitrary genus. The more general statements are due to Manin [296, 297], who coined the term *Picard-Fuchs equations* (see also [245, 250, 246]). and to Griffiths [214, 215, 216]. Modern proofs of this D-finiteness result are based, as in the case of diagonals (§3.1) on the finiteness of the relative de Rham cohomology of the complementary of the hypersurface defined by the singular locus of the rational integrand [219, 313, 228].

The principle of creative telescoping for the computation of Picard-Fuchs equations, already used by Manin in [297], is the following: **if** one knows a differential operator $P(t, \partial_t)$ free of **x** and rational functions $(A_1, ..., A_n)$ such that

(25)
$$P(t,\partial_t) \cdot H(t,\mathbf{x}) = \sum_{i=1}^n \frac{\partial A_i}{\partial x_i},$$

then the integral with respect to x "telescopes", leading to the differential equation

$$P(t,\partial_t)\cdot I(t)=0.$$

(The reason is simply that integrals over cycles of pure derivatives are equal to zero.)

The differential operator *P* is called a *telescoper* for the integrand H(t, x), and (A_1, \ldots, A_n) is called a *certificate*. The question is then how to produce effectively an equality like (25). Ideally, one would like to compute the telescoper without computing the certificate, for reasons that will become apparent in the next example.

Example 41 (Perimeter of an ellipse). Computations of differential equations for periods can be traced back to Euler [175, §7], in his study of the perimeter p(e) of an ellipse with semi-major axis 1, as a function of its eccentricity e:

$$p(e) = 4 \int_0^1 \sqrt{\frac{1 - e^2 x^2}{1 - x^2}} \, \mathrm{d}x = 2\pi - \frac{\pi}{2}e^2 - \frac{3\pi}{32}e^4 - \frac{5}{128}e^6 - \frac{175}{8192}e^8 + \cdots$$

The question can be casted into the framework of periods of rational integrals:

$$p(e) = \oint \frac{\mathrm{d}x\mathrm{d}y}{1 - \frac{1 - e^2 x^2}{(1 - x^2)y^2}},$$

and a telescopic relation of type (25) reads:

$$\left((e - e^3)\partial_e^2 + (1 - e^2)\partial_e + e \right) \cdot \left(\frac{1}{1 - \frac{1 - e^2 x^2}{(1 - x^2)y^2}} \right) = \\ \partial_x \left(-\frac{e(-1 - x + x^2 + x^3)y^2(-3 + 2x + y^2 + x^2(-2 + 3e^2 - y^2))}{(-1 + y^2 + x^2(e^2 - y^2))^2} \right) \\ + \partial_y \left(\frac{2e(-1 + e^2)x(1 + x^3)y^3}{(-1 + y^2 + x^2(e^2 - y^2))^2} \right).$$

From there, Euler's equation $(e - e^3)p''(e) + (1 - e^2)p'(e) + ep(e) = 0$ follows directly. The size of the certificate is much bigger than the size of the telescoper.

Several generations of Creative Telescoping algorithms. Algorithms for creative telescoping for periods can be divided into four generations. Algorithms from the first generation (1G) —*à la Lipshitz*— use holonomy theory and elimination for operator ideals [393, 369, 385, 370, 144]; they are not very efficient in practice. Algorithms from the second generation (2G), due to Chyzak [142] and to Koutschan [263], are generalizations of Zeilberger's fast algorithms for hypergeometric summation and hyperexponential integration [392, 394, 9]; they reduce the resolution of the telescopic equation (25) to the computation of the rational solutions of a system of linear differential equations. The roots of this method can be traced back to Picard [326] for n = 2. Algorithms from the third generation (3G) only use linear algebra, and are based on an idea that was first formulated by Apagodu and Zeilberger in [311, 14], and has later been refined and generalized [263, 121, 120, 123]. This approach is interesting not only because it is easier to implement and tends to run faster than earlier algorithms, but also because it is easy to analyze.

A common drawback of these three generations of algorithms is that they all compute certificates, whose size is much bigger than that of telescopers. Moreover, 1G algorithms are slow, 2G algorithms have a bad or unknown complexity, and 3G algorithms do not necessarily output telescopers of minimal orders. However, already algorithms from the second generation are able to solve non-trivial problems.

Example 42 (Diagonal 3D Rook paths, cont.). Using notation from Example 34 and from the proof of Theorem 35, the aim is to construct a linear differential operator $P(t, \partial_t)$, and two rational functions R and S in $\mathbb{Q}(t, x, y)$ such that

$$P(G) = \frac{\partial R}{\partial x} + \frac{\partial S}{\partial y}.$$

Maple's implementation of Chyzak's algorithm is able to do this in a few seconds:

It outputs the differential equation $P(\Delta) = 0$ satisfied by $\Delta = \sum_{n>0} D_n t^n$, where

$$P = t(t-1)(64t-1)(3t-2)(6t+1)\partial_t^3 + (4608t^4 - 6372t^3 + 813t^2 + 514t - 4)\partial_t^2 + 4(576t^3 - 801t^2 - 108t + 74)\partial_t,$$

which helps proving a recurrence conjectured in [173].

4G Creative Telescoping. Algorithms from the fourth and most recent generation of creative telescoping algorithms are called *reduction-based algorithms*. Its roots are in works by Hermite [232] and Picard [326, 327]. This approach was first applied to the integration of bivariate rational functions by Bostan, Chen, Chyzak and Li [65]. This first article generated a very active area of research [125, 66, 87, 118, 236, 281, 79, 124, 127, 88].

Let us explain the principle of the method in the univariate case, that is when n = 1 in the telescopic Equation (25).

The problem at hand is: given $H = P/Q \in \mathbb{K}(t, x)$, compute $\oint_{\gamma} H(t, x) dx$. The principle of the method originates from the Hermite reduction [232], a procedure for computing a normal form of a univariate function modulo derivatives. Hermite introduced his method as a way to compute the algebraic part of the primitive of a univariate rational function without computing the roots of its denominator, as opposed to the classical partial fraction decomposition method.

By Hermite reduction, the integrand *H* can be written in *reduced form*

$$H = \partial_x(g) + \frac{a}{Q^\star},$$

where Q^* is the squarefree part of Q and $\deg_r(a) < d^* := \deg_r(Q^*)$.

The principle of the algorithm in [65] is then the following:

1. For $i = 0, 1, ..., d^*$ compute the Hermite reduction of $\partial_t^i(H)$:

$$\partial_t^i(H) = \partial_x(g_i) + \frac{a_i}{Q^\star}, \quad \deg_x(a_i) < d^\star.$$

2. Find the first linear relation over $\mathbb{Q}(t)$ of the form $\sum_{k=0}^{r} c_k a_k = 0$.

Then $L = \sum_{k=0}^{r} c_k \partial_t^k$ is a telescoper, and $\sum_{k=0}^{r} c_k g_k$ the corresponding certificate. The method has been extended to the multivariate case of periods of rationals integrals by Bostan, Lairez and Salvy [87]. They have obtained the following result.

THEOREM 43 ([87]). Let $H = \frac{P}{Q}$ be a rational function in t and $\mathbf{x} = x_1, \ldots, x_n$ and denote by $d_{\mathbf{x}}$ the degree of Q w.r.t. \mathbf{x} , and $d_t = \max(\deg_t P, \deg_t Q)$. Assume $\deg_{\mathbf{x}} P + n + 1 \leq d_{\mathbf{x}}$. Then a telescoper for H can be computed using $\tilde{O}(e^{3n}d_{\mathbf{x}}^{8n}d_t)$ operations in Q, uniformly in all the parameters. The minimal telescoper has order $\leq d_{\mathbf{x}}^n$ and degree $O(e^n d_{\mathbf{x}}^{3n} d_t)$. These size bounds are generically reached.

There are three main ideas behind the proof of Theorem 43:

- in the *generic case*, a multivariate generalization of Hermite's reduction is used; it called the *Griffiths–Dwork method* [165, §3], [166, §8], [215];
- in the *general case*, a deformation technique is used to reduce to the generic case, by an input perturbation using a new free variable;
- fast linear algebra algorithms for polynomial matrices [360, 396] is used to deal with Macaulay matrices that encode Gröbner bases computations.

The algorithm behind Theorem 43 is the first algorithm for creative telescoping with *polynomial complexity* in the generic size of the output Picard-Fuchs equation. It avoids the costly computation of certificates. This is crucial since, generically, certificates have size $\Omega(d_x^{n^2/2})$. Previous algorithms have (at least) doubly-exponential complexity, inherited from the fact that they need to compute certificates. A recent, and highly non-trivial, extension of the results in [87] was given by Lairez [281]. It tremendously improves the practical efficiency of the algorithm in [87].

3.3. Binomial sums. As explained in §3.2.1, creative telescoping allows to prove identities like Dixon's (first item in Example 36), and to deal with definite sums like

(26)
$$\sum_{k=0}^{n} \frac{4^{k}}{\binom{2k}{k}}, \quad \sum_{k=0}^{n} \left(\sum_{j=0}^{k} \binom{n}{j}\right)^{3} \text{ or } \sum_{i=0}^{n} \sum_{j=0}^{n} \binom{i+j}{j}^{2} \binom{4n-2i-2j}{2n-2i}.$$

Many multiple sums can be cast into problems of rational integration by passing to generating functions. This observation was intensively used by Egorychev in his book [171], but its algorithmic consequences were studied only quite recently by Bostan, Lairez and Salvy [88]. They defined a class of multi-indexed sequences called (multiple) binomial sums, which is closed under partial summation, and contains most of the sequences obtained by multiple summation of products of binomial coefficients and also all the sequences with algebraic generating function. Not every sum that creative telescoping can handle is a binomial sum: for example, among the three sums in Eq. (26), the second one and the third one are binomial sums but the first one is not, since it contains the inverse of a binomial coefficient. Yet many sums coming from combinatorics and number theory are binomial sums. The starting point is that *integral representations* of the generating function of a binomial sum can be computed in an automated way. The outcome is twofold. Firstly, the generating functions of *univariate* binomial sums are exactly the diagonals of rational power series; this equivalence characterizes binomial sums in an intrinsic way. All the theory of diagonals transfers to univariate binomial sums and gives many interesting arithmetic properties. Secondly, integral representations can be used to actually compute with binomial sums (e.g. find recurrence relations or prove identities automatically) via the computation of Picard-Fuchs equations.

Example 44. (A particular instance of Dixon's identity) We will simply illustrate the main points of the method in [88] on the identity

(27)
$$\sum_{k=0}^{2n} (-1)^k {\binom{2n}{k}}^3 = (-1)^n \frac{(3n)!}{n!^3}.$$

The strategy is as follows: find an integral representation of the generating function of the left-hand side; simplify this integral representation using partial integration; use the simplified integral representation to compute a differential equation of which the generating function is solution; transform this equation into a recurrence relation; solve this recurrence relation.

First of all, the binomial coefficient $\binom{n}{k}$ is the coefficient of x^k in $(1 + x)^n$. Cauchy's integral formula ensures that

$$\binom{n}{k} = \frac{1}{2\pi i} \oint_{\gamma} \frac{(1+x)^n}{x^k} \frac{\mathrm{d}x}{x},$$

where γ is the circle $\left\{x \in \mathbb{C} \mid |x| = \frac{1}{2}\right\}$. Therefore, the cube of a binomial coefficient can be represented as a triple integral

$$\binom{2n}{k}^3 = \frac{1}{(2\pi i)^3} \oint_{\gamma \times \gamma \times \gamma} \frac{(1+x_1)^{2n}}{x_1^k} \frac{(1+x_2)^{2n}}{x_2^k} \frac{(1+x_3)^{2n}}{x_3^k} \frac{dx_1}{x_1} \frac{dx_2}{x_2} \frac{dx_3}{x_3}.$$

As a result, the generating function y(t) of the left-hand side of Equation (27) equals

$$= \frac{1}{(2i\pi)^3} \oint_{\gamma^3} \sum_{n=0}^{\infty} \sum_{k=0}^{2n} \left(t \prod_{i=1}^3 (1+x_i)^2 \right)^n \left(\frac{-1}{x_1 x_2 x_3} \right)^k \frac{dx_1}{x_1} \frac{dx_2}{x_2} \frac{dx_3}{x_3}$$

$$= \frac{1}{(2i\pi)^3} \oint_{\gamma^3} \sum_{n=0}^{\infty} \left(t \prod_{i=1}^3 (1+x_i)^2 \right)^n \frac{1 - \left(\frac{-1}{x_1 x_2 x_3} \right)^{2n+1}}{1 + \frac{1}{x_1 x_2 x_3}} \frac{dx_1}{x_1} \frac{dx_2}{x_2} \frac{dx_3}{x_3}$$

$$= \frac{1}{(2i\pi)^3} \oint_{\gamma^3} \frac{\left(x_1 x_2 x_3 - t \prod_{i=1}^3 (1+x_i)^2 \right) dx_1 dx_2 dx_3}{\left(x_1^2 x_2^2 x_3^2 - t \prod_{i=1}^3 (1+x_i)^2 \right) \left(1 - t \prod_{i=1}^3 (1+x_i)^2 \right)}.$$

The partial integral with respect to x_3 along the circle $|x_3| = \frac{1}{2}$ is the sum of the residues of the rational function being integrated at the poles whose modulus is less than $\frac{1}{2}$. When |t| is small and $|x_1| = |x_2| = \frac{1}{2}$, the poles coming from the factor $x_1^2 x_2^2 x_3^2 - t \prod_{i=1}^3 (1+x_i)^2$ all have a modulus that is smaller than $\frac{1}{2}$: they are asymptotically proportional to $|t|^{1/2}$. In contrast, the poles coming from the factor $1 - t \prod_{i=1}^3 (1+x_i)^2$ behave like $|t|^{-1/2}$ and have all a modulus that is bigger than $\frac{1}{2}$. In particular, any two poles that come from the same factor are either both asymptotically small or both asymptotically large. This implies that the partial integral is a rational function of t, x_1 and x_2 ; and we compute that

$$y(t) = \frac{1}{(2i\pi)^2} \oint_{\gamma \times \gamma} \frac{x_1 x_2 dx_1 dx_2}{x_1^2 x_2^2 - t(1+x_1)^2 (1+x_2)^2 (1-x_1 x_2)^2}$$

This formula echoes the original proof of [158] in which the left-hand side of (27) is expressed as the coefficient of $(xy)^{4n}$ in $((1-y^2)(1-z^2)(1-y^2z^2))^{2n}$. Using any algorithm described in §3.2.3 that performs definite integration of rational functions reveals a differential equation satisfied by y(t):

$$t(27t+1)y'' + (54t+1)y' + 6y = 0.$$

Looking at the coefficient of t^n in this equality leads to the recurrence relation

$$3(3n+2)(3n+1)u_n + (n+1)^2u_{n+1} = 0,$$

where $u_n = \sum_{k=0}^{2n} (-1)^k {\binom{2n}{k}}^3$. Since $u_0 = 1$, it proofs Dixon's identity (27).

Note that the method avoids the computation of certificates; this nice feature is inherited from the computation of Picard-Fuchs equations for periods of rational integrals, which can be achieved efficiently without computing the corresponding certificate and without introducing spurious singularities (§3.2.3). This should be contrasted with the usual creative telescoping methods for sums (§3.2.1).

3.4. Creative Telescoping for quarter plane walks. Let us now turn back to quarter plane walks with small steps. We focus on models 1–19 in Fig. 7, and to Theorem 14. We write F(t; x, y) for the full generating function $F_{\mathfrak{S}}(t; x, y)$, where \mathfrak{S} is one of the 19 models.

Using the kernel method, Bousquet-Mélou and Mishna showed in [101, Prop. 8] that the generating function F(t; x, y) can be written in the form

(28)
$$F(t;x,y) = \frac{1}{xy} [x^{>}][y^{>}] \frac{N(x,y)}{1 - tS(x,y)}$$

where N(x, y) and S(x, y) are certain Laurent polynomials in y with coefficients that are rational functions in x. The intended reading of (28) is: first interpret N(x,y)/(1-tS(x,y)) as an element of Q(x)[y,1/y][[t]]; let $[y^>]$ act term by term, obtaining a series in Q(x)[y][[t]] that actually belongs to Q[x,1/x][y][[t]] for all cases in Figure 7; then let $[x^>]$ act term by term, finally obtaining an element of Q[x][y][[t]]. In this reading, the composition $[x^>][y^>]$ of positive-part operators is only applied to Laurent polynomials, for which it is well-defined, in a unique way.

As pointed out by Bousquet-Mélou and Mishna, Equation (28) already implies the D-finiteness of F(t; x, y), by Theorem 33 and since positive parts can be encoded as diagonals. To be more specific, the positive part $[x^>][y^>]R(t; x, y)$ of a formal power series $R \in \mathbb{Q}[[x, y, t]]$ can be encoded as

(29)
$$\frac{x}{1-x}\frac{y}{1-y}\odot_{x,y}R(t;x,y) = \text{Diag}_{x,x'}\text{Diag}_{y,y'}\frac{x}{1-x}\frac{y}{1-y}R(t;x',y'),$$

where the Hadamard product denoted $\odot_{x,y}$ is the term-wise product of two series, while the diagonal operator $\text{Diag}_{x,x'}$ selects those terms with equal exponents of xand x'. This argument also implies an algorithm for computing linear differential equations satisfied by F(t;x,y), since diagonals can be computed using creative telescoping. Therefore, from (28) one could, in principle, determine differential equations for F(t;x,y). However, the direct use of (29) in our context leads to infeasible computations; worse, the intermediate algebraic objects involved in the calculations would probably have too large sizes to be merely written and stored. This is really unfortunate, since our need is mere evaluations of the diagonals in (29) at specific values for x and y.

Example 45. (King Walks in the Quarter Plane) We illustrate the approach on the *king walks* (model 4 with $\mathfrak{S} = \bigotimes$ in Fig. 7). The first terms of the length generating function F(t; 1, 1) read (see http://oeis.org/A151331)

$$F(t;1,1) = 1 + 3t + 18t^{2} + 105t^{3} + 684t^{4} + 4550t^{5} + 31340t^{6} + 219555t^{7} + \cdots,$$

and we describe the method used in [76] to obtain the closed formula (9) for it. First, the kernel equation (7) reads

(30)
$$xy\mathfrak{J}(x,y)F(x,y) = xy - tx(x+1+\bar{x})F(x,0) - ty(y+1+\bar{y})F(0,y) + tF(0,0),$$

where $F(x, y) \equiv F(t; x, y)$, $\bar{x} := 1/x$, $\bar{y} := 1/y$ and $\mathfrak{J}(x, y)$ is the Laurent polynomial

$$\mathfrak{J}(x,y) = 1 - t \sum_{(i,j) \in \mathfrak{S}} x^i y^j = 1 - t(xy + y + \bar{x}y + x + \bar{x} + x\bar{y} + \bar{y} + \bar{x}\bar{y}).$$

The group of \mathfrak{S} has order 4: it contains the elements $(x, y), (\bar{x}, y), (\bar{x}, \bar{y}), (x, \bar{y})$, which leave invariant $\mathfrak{J}(t; x, y)$. Applying these rational transformations to the kernel equation (30) yields the four relations:

$$\begin{aligned} xy\mathfrak{J}(x,y)F(x,y) &= xy - tx(x+1+\bar{x})F(x,0) - ty(y+1+\bar{y})F(0,y) + tF(0,0), \\ -\bar{x}y\mathfrak{J}(x,y)F(\bar{x},y) &= -\bar{x}y + t\bar{x}(x+1+\bar{x})F(\bar{x},0) + ty(y+1+\bar{y})F(0,y) - tF(0,0), \\ \bar{x}\bar{y}\mathfrak{J}(x,y)F(\bar{x},\bar{y}) &= \bar{x}\bar{y} - t\bar{x}(x+1+\bar{x})F(\bar{x},0) - t\bar{y}(y+1+\bar{y})F(0,\bar{y}) + tF(0,0), \\ -\bar{x}y\mathfrak{J}(x,y)F(\bar{x},y) &= -x\bar{y} + tx(x+1+\bar{x})F(x,0) + t\bar{y}(y+1+\bar{y})F(0,\bar{y}) - tF(0,0). \end{aligned}$$

Upon adding up these equations, all terms in the right-hand side involving F disappear, resulting in

$$xyF(x,y) - \bar{x}yF(\bar{x},y) + \bar{x}\bar{y}F(\bar{x},\bar{y}) - x\bar{y}F(x,\bar{y}) = \mathfrak{J}(x,y)^{-1}\left(xy - \bar{x}y + \bar{x}\bar{y} - x\bar{y}\right).$$

Now, the main observation is that on the left-hand side, all terms except the first one involve negative powers either of x or of y. Therefore, extracting positive parts expresses the generating series xyF(x, y) as the positive part (w.r.t. x and y) of a trivariate rational function:

(31)
$$xyF(x,y) = [x^{>}][y^{>}] \left(\frac{xy - \bar{x}y + \bar{x}\bar{y} - x\bar{y}}{1 - t(xy + y + y\bar{x} + \bar{x} + \bar{x}\bar{y} + \bar{y} + x\bar{y} + x)} \right)$$

Up to this point, the reasoning is borrowed from Bousquet-Mélou's and Mishna's article [101]. Combined with Theorem 33, it already implies that F(x, y) is D-finite; in particular, F(1, 1) is also D-finite. Our aim is to refine this qualitative result, and explicitly obtain a linear differential equation satisfied by F(1, 1).

Starting from (31) and following more closely Lipshitz' encoding [287], a first observation is that F(x, y) is equal to the iterated diagonal $\text{Diag}_{x_1, x_2} \text{Diag}_{y_1, y_2}$ of the rational function

(32)
$$\frac{x_2y_2(x_1y_1 - \bar{x}_1y_1 + \bar{x}_1\bar{y}_1 - x_1\bar{y}_1)}{(1 - x_2)(1 - y_2)(1 - t(x_1y_1 + y_1 + y_1\bar{x}_1 + \bar{x}_1 + \bar{x}_1\bar{y}_1 + \bar{y}_1 + x_1\bar{y}_1 + x_1))}.$$

However, this computation is too difficult, and exceeds by far the limits of the best existing algorithms for diagonals. The reason is that differential equations w.r.t. t and with polynomial coefficients in x, y, t are really huge, so the main limitation of algorithms computing (32) already comes from the size of the output. Another weakness of the diagonal encoding (32) is that it does not provide direct access to the univariate series F(1,1), since taking diagonals and specializing variables are operations that do not commute.

To circumvent these difficulties and to make the computation feasible, the key idea in [76] is to encode the positive part in (31) as a formal residue: (33)

$$F(\alpha,\beta) = [x^{-1}y^{-1}] \left(\frac{xy - \bar{x}y + \bar{x}\bar{y} - x\bar{y}}{(1 - \alpha x)(1 - \beta y)(1 - t(xy + y + y\bar{x} + \bar{x} + \bar{x}\bar{y} + \bar{y} + x\bar{y} + x))} \right).$$

The formal proof of this encoding is delicate. The advantage of (33) over (32) is twofold. On the one hand, the residue computation can be carried out by using a single call to the creative-telescoping algorithm for rational functions, while the diagonal computation (32) has two steps, the first for a rational function in five variables, the second for an algebraic function in four variables. On the other hand, and more importantly, taking residues commutes with specialization, contrarily to positive parts and diagonals. Therefore, the generating series for walks F(1,1) is

$$F(1,1) = [x^{-1}y^{-1}] \left(\frac{xy - \bar{x}y + \bar{x}\bar{y} - x\bar{y}}{(1-x)(1-y)(1-t(xy+y+y\bar{x}+\bar{x}+\bar{x}\bar{y}+\bar{y}+x\bar{y}+x))} \right),$$

and a differential equation L(F(1,1)) = 0 can now be computed by creative telescoping:

(34)
$$L = t^{2}(1+4t)(8t-1)(2t-1)(1+t)\partial_{t}^{3} + t(200t^{3}+576t^{4}-33t-252t^{2}+5)\partial_{t}^{2} + 4(22t^{3}-117t^{2}-12t+288t^{4}+1)\partial_{t}+384t^{3}-12-144t-72t^{2}.$$

60

Note that this is precisely the differential operator *guessed* in [84].

Moreover, factorization algorithms for linear differential operators [217, 353, 112, 380] can be used to prove that $L = L_2L_1$, where $L_1 = \partial_t + 1/t$ and

(35)
$$L_2 = t^2(1+4t)(1-8t)(1-2t)(1+t)\partial_t^2 + 2t(256t^4+80t^3-111t^2-14t+2)\partial_t + 768t^4+8t^3-306t^2-30t+2.$$

It follows that the Laurent power series

$$f(t) = \frac{dF}{dt}(1,1) + \frac{F(1,1)}{t} = t^{-1} + 6 + 54t + 420t^2 + 3420t^3 + 27300t^4 + O(t^6)$$

is a solution of L_2 . Starting from the second order operator L_2 , algorithmic methods explained in [77, §2.6] (see also [274, 240, 241]) allow to express f(t) as

$$f(t) = \frac{1}{t(1+4t)^3} \cdot {}_2F_1\left(\frac{3}{2}\frac{3}{2} \middle| \frac{16t(1+t)}{(1+4t)^2}\right).$$

Finally, solving the equation d/dt F(1,1) + F(1,1)/t = f(t) yields formula (9).

Similarly, for *indeterminates* α and β we obtain the formal residue representations for $F(\alpha, 0)$ and $F(0, \beta)$, and creative-telescoping techniques still allow the effective computation of differential operators for $F(\alpha, 0)$, resp. for $F(0, \beta)$. Owing to the additional symbolic indeterminate, the computations are much harder than for F(1,1), but still feasible. Each of the resulting differential operators factors again, this time as a product of an order-two operator and of three order-one operators. Moreover, the second-order operators are again solvable in terms of $_2F_1$ functions. Finally, a closed formula for $F(\alpha, \beta)$ is obtained from the closed formulas for $F(\alpha, 0)$ and $F(0, \beta)$ via the kernel equation (30). This detour is computationally crucial, since performing creative telescoping directly on the five-variable rational function from (33) is not feasible even using today's best algorithms.

A similar reasoning applies to any of the 19 models in Fig. 7 with finite group and non-zero orbit sum, and this allows to prove Theorem 14 with the help of the fundamental equation

$$\mathsf{GF} = \mathsf{PositivePart}\left(\frac{\mathsf{orbit sum}}{\mathsf{kernel}}\right)$$

3.5. Back to the exercise in §1.1. To conclude, we come back to the problem stated at the very beginning of the memoir, for which we have guessed the answer in §2.6. Recall that \mathfrak{S} denotes the step set { \uparrow , \leftarrow , \searrow }. For convenience, we will continue to use the shortcut notation $\bar{x} = 1/x$, $\bar{y} = 1/y$.

3.5.1. A functional equation for \mathfrak{S} **-walks in** \mathbb{N}^2 . Let us consider the full generating function for \mathfrak{S} -walks in \mathbb{N}^2

$$Q(x,y) = \sum_{n=0}^{\infty} \sum_{i=0}^{n} \sum_{j=0}^{n} q_{n;i,j} t^{n} x^{i} y^{j} \in \mathbb{Q}[x,y][[t]].$$

It satisfies the kernel equation (7), which reads:

(36)
$$(1 - t(y + \bar{x} + x\bar{y}))xyQ(x,y) = xy - tx^2Q(x,0) - tyQ(0,y).$$

We are interested in the generating function of diagonal returns $B(t) = [x^0] Q(x, \bar{x})$.

3.5.2. A functional equation for \mathfrak{S} -walks in $\mathbb{Z} \times \mathbb{N}$. Similarly, let $H(t; x, y) \equiv H(x, y)$ denote the full generating function for \mathfrak{S} -walks in $\mathbb{Z} \times \mathbb{N}$,

$$H(x,y) = \sum_{n=0}^{\infty} \sum_{i=-n}^{n} \sum_{j=0}^{\infty} h(n;i,j) t^{n} x^{i} y^{j} \in \mathbb{Q}[x,\bar{x},y][[t]].$$

It satisfies a functional equation very similar to (36), namely

(37)
$$(1 - t(y + \bar{x} + x\bar{y}))xyH(x,y) = xy - tx^2H(x,0)$$

This time, we are interested in $A(t) = [x^0] H(x, 0)$, the generating function of excursions in the upper half-plane.

3.5.3. The kernel method for $\mathbb{Z} \times \mathbb{N}$ **.** We solve Eq. 37 by using the same technique as we did for Dyck walks (Equation (3) from Example 5).

Let

$$y_0 = \frac{x - t - \sqrt{(t - x)^2 - 4t^2 x^3}}{2tx} \qquad = xt + t^2 + (x^2 + \bar{x})t^3 + (3x + \bar{x}^2)t^4 + \cdots$$

be the (unique) root in $\mathbb{Q}[x, \bar{x}][[t]]$ of $K(x, y_0) = 0$, where $K(x, y) = 1 - t(y + \bar{x} + x\bar{y})$. Then plugging y_0 in (37) yields

$$0 = K(x, y_0)yH(x, y_0) = y_0 - txH(x, 0),$$

and thus

$$H(x,0) = \frac{y_0}{tx}$$
 and $A(t) = \begin{bmatrix} x^0 \end{bmatrix} \frac{y_0}{tx}$.

This allows to express A(t) as a period of an algebraic integral. A differential equation satisfied by A(t) can then be computed using creative telescoping:

> y0:= - sqrt((t-x)^2 - 4*t^2*x^3)/(2*t*x): > DEtools[Zeilberger](1/x * y0/(t*x), t, x, Dt)[1];

which proves the equation

$$(27t4 - t)A''(t) + (108t3 - 4)A'(t) + 54t2A(t) = 0,$$

or equivalently, the recurrence relation on its coefficients:

$$27(n+2)(n+1)a_n = (n+6)(n+3)a_{n+3}$$

3.5.4. The kernel method for \mathbb{N}^2 . The inventory $\chi(x, y) = \bar{x} + y + x\bar{y}$ of \mathfrak{S} is left unchanged by the involutions

$$\Phi: (x, y) \mapsto (\bar{x}y, y)$$
 and $\Psi: (x, y) \mapsto (x, x\bar{y})$.

which generate a finite dihedral group D_3 of order 6:

$$(x,y) \xrightarrow{\Phi} (\bar{x}y,y) \xrightarrow{\Psi} (\bar{x}y,\bar{x}) \xrightarrow{\Phi} (\bar{y},\bar{x})$$

$$(x,y) \xrightarrow{\Psi} (x,x\bar{y}) \xrightarrow{\Phi} (\bar{y},x\bar{y}) \xrightarrow{\Psi}$$

Letting the group act on the kernel equation (36) gives six equations, whose alternate sum gives birth to the orbit equation:

$$\begin{aligned} xyQ(x,y) &- \bar{x}y^2Q(\bar{x}y,y) + \bar{x}^2yQ(\bar{x}y,\bar{x}) \\ &- \bar{x}\bar{y}Q(\bar{y},\bar{x}) + x\bar{y}^2Q(\bar{y},x\bar{y}) - x^2\bar{y}Q(x,x\bar{y}) = \\ &\frac{xy - \bar{x}y^2 + \bar{x}^2y - \bar{x}\bar{y} + x\bar{y}^2 - x^2\bar{y}}{1 - t(y + \bar{x} + x\bar{y})} \end{aligned}$$

Extracting the part with positive powers of x and y like in (3.4) gives

$$xyQ(x,y) = [x^{>0}y^{>0}] \frac{xy - \bar{x}y^2 + \bar{x}^2y - \bar{x}\bar{y} + x\bar{y}^2 - x^2\bar{y}}{1 - t(y + \bar{x} + x\bar{y})}.$$

Then, applying the method in [76] allows to express B(t) as a residue:

$$B(t) = [x^0]Q(x,\bar{x}) = [u^{-1}v^{-1}z^{-1}] \frac{\bar{u}\bar{v} - u\bar{v}^2 + u^2\bar{v} - uv + \bar{u}v^2 - \bar{u}^2v}{z(1-zu)(1-v\bar{z})(1-t(\bar{v}+u+\bar{u}v))}.$$

Finally, multivariate Creative Telescoping proves a differential equation for B(t):

> OS := x*y - y²/x + y/x² - 1/x/y + x/y² - x²/y; > ker := 1-t*(y + 1/x + x/y); > S:=normal(subs({x=1/u,y=1/v}, OS/ker)/(1-z*u)/(1-v/z)/z); > Mgfun:-creative_telescoping(S,t::diff,[z::diff,u::diff,v::diff]):

namely $(27t^4 - t)B''(t) + (108t^3 - 4)B'(t) + 54t^2B(t) = 0.$

3.5.5. Conclusion. We have proved that A(t) and B(t) are both solutions of

$$(27t4 - t)y''(t) + (108t3 - 4)y'(t) + 54t2y(t) = 0.$$

Solving this equation in closed form proves that

$$A(t) = B(t) = {}_{2}F_{1}\left(\frac{1/3}{2}\frac{2}{3}\right) = \sum_{n=0}^{\infty} \frac{(3n)!}{n!^{3}} \frac{t^{3n}}{n+1}.$$

Thus the two sequences are equal to

$$a_{3n} = b_{3n} = \frac{(3n)!}{n!^2 \cdot (n+1)!}$$
, and $a_m = b_m = 0$ if 3 does not divide *m*.

REFERENCES

- N. H. ABEL, Œuvres complètes. Tome II, Éditions J. Gabay, Sceaux, 1992. Reprint of the second (1881) edition. Available at http://gallica.bnf.fr/ark:/12148/bpt6k2438n.
- [2] M. ABRAMOWITZ AND I. A. STEGUN (EDITORS), <u>Handbook of mathematical functions with formulas, graphs, and mathematical tables</u>, Dover, New York, 1992. 1046 pages. Reprint of the 1972 edition.

- [3] B. ADAMCZEWSKI AND J. P. BELL, Diagonalization and rationalization of algebraic Laurent series, Ann. Sci. Éc. Norm. Supér. (4), vol. 46, no. 6 (2013), pp. 963–1004, doi:10.24033/asens.2207.
- [4] L. ADDARIO-BERRY AND B. A. REED, Ballot theorems, old and new, in Horizons of combinatorics, vol. 17 of Bolyai Soc. Math. Stud., Springer, Berlin, 2008, pp. 9–35, doi:10.1007/ 978-3-540-77200-2_1.
- [5] J. AEBLY, <u>Démonstration du problème du scrutin par des considérations géométriques</u>, Enseignement Math., vol. 23 (1923), pp. 185–186, doi:10.5169/seals-19738.
- [6] A. AEPPLI, <u>A</u> propos de l'interprétation géométrique du problème du scrutin, Enseignement Math., vol. 23 (1923), pp. 328–329, doi:10.5169/seals-19739.
- [7] M. ALBERT AND M. BOUSQUET-MÉLOU, Permutations sortable by two stacks in parallel and quarter plane walks, European J. Combin., vol. 43 (2015), pp. 131–164, doi:10.1016/j.ejc.2014.08.024.
- [8] G. L. ALEXANDERSON, The random walks of George Pólya, MAA Spectrum, Mathematical Association of America, Washington, DC, 2000, www.cambridge.org/9780883855287. With [reprinted] essays by Ralph P. Boas, Jr., Kai Lai Chung, D. H. Lehmer, R. C. Read, Doris Schattschneider, M. M. Schiffer, and Alan Schoenfeld.
- [9] G. ALMKVIST AND D. ZEILBERGER, The method of differentiating under the integral sign, J. Symbolic Comput., vol. 10, no. 6 (1990), pp. 571–591, doi:10.1016/S0747-7171(08)80159-9.
- [10] D. ANDRÉ, Solution directe du problème résolu par M. Bertrand, C. R. Acad. Sci. Paris, vol. 105 (1887), pp. 436-437.
- [11] Y. ANDRÉ, <u>G-functions and geometry</u>, Aspects of Mathematics, E13, Friedr. Vieweg & Sohn, Braunschweig, 1989, doi:10.1007/978-3-663-14108-2.
- [12] Y. ANDRÉ, Séries Gevrey de type arithmétique. I. Théorèmes de pureté et de dualité, Ann. of Math. (2), vol. 151, no. 2 (2000), pp. 705–740, doi:10.2307/121045.
- [13] Y. ANDRÉ, Sur la conjecture des *p*-courbures de Grothendieck-Katz et un problème de Dwork, in Geometric aspects of Dwork theory. Vol. I, II, Walter de Gruyter, Berlin, 2004, pp. 55–112.
- [14] M. APAGODU AND D. ZEILBERGER, Multi-variable Zeilberger and Almkvist-Zeilberger algorithms and the sharpening of Wilf-Zeilberger theory, Adv. in Appl. Math., vol. 37, no. 2 (2006), pp. 139–152, doi:10.1016/j.aam.2005.09.003.
- [15] R. APÉRY, Sur certaines séries entières arithmétiques, in Study group on ultrametric analysis, 9th year: 1981/82, No. 1, Inst. Henri Poincaré, Paris, 1983, pp. Exp. No. 16, 2, http://www. numdam.org/item/GAU_1981-1982_9_1_A9_0.
- [16] R. APÉRY, Irrationalité de $\zeta(2)$ et de $\zeta(3)$, Astérisque, vol. 61 (1979), pp. 11–13.
- [17] D. ARQUÈS, Dénombrements de chemins dans R² soumis à contraintes, RAIRO Inform. Théor. Appl., vol. 20, no. 4 (1986), pp. 473–482.
- [18] R. ASKEY, How can mathematicians and mathematical historians help each other?, in History and philosophy of modern mathematics (Minneapolis, MN, 1985), Minnesota Stud. Philos. Sci., XI, Univ. Minnesota Press, Minneapolis, MN, 1988, pp. 201–217.
- [19] A. AYYER, Towards a human proof of Gessel's conjecture, J. Integer Seq., vol. 12, no. 4 (2009), pp. Article 09.4.2, 15, http://cs.uwaterloo.ca/journals/JIS/VOL12/Ayyer/ayyer7.pdf.
- [20] A. BACHER, M. KAUERS, AND R. YATCHAK, Continued classification of 3D lattice models in the positive octant, in DMTCS Proceedings of FPSAC'16, Vancouver, Canada, 2016, pp. 95–106, http://www.lix.polytechnique.fr/~pilaud/FPSAC16/listPapers.html.
- [21] D. H. BAILEY AND J. M. BORWEIN, Exploratory experimentation and computation, Notices Amer. Math. Soc., vol. 58, no. 10 (2011), pp. 1410–1419, http://www.ams.org/notices/201110/ rtx111001410p.pdf.
- [22] D. H. BAILEY, J. M. BORWEIN, N. J. CALKIN, R. GIRGENSOHN, D. R. LUKE, AND V. H. MOLL, Experimental mathematics in action, A K Peters, Ltd., Wellesley, MA, 2007.
- [23] D. H. BAILEY, J. M. BORWEIN, V. KAPOOR, AND E. W. WEISSTEIN, Ten problems in experimental mathematics, Amer. Math. Monthly, vol. 113, no. 6 (2006), pp. 481–509, doi:10.2307/27641975.
- [24] K. BALL AND T. RIVOAL, Irrationalité d'une infinité de valeurs de la fonction zêta aux entiers impairs, Invent. Math., vol. 146, no. 1 (2001), pp. 193–207, doi:10.1007/s002220100168.
 [25] C. BANDERIER, Limit laws for basic parameters of lattice paths with unbounded jumps, in Math-
- [25] C. BANDERIER, Limit laws for basic parameters of lattice paths with unbounded jumps, in Mathematics and computer science, II (Versailles, 2002), Trends Math., Birkhäuser, Basel, 2002, pp. 33–47, doi:10.1007/978-3-0348-8211-8_2.
- [26] C. BANDERIER AND P. FLAJOLET, Basic analytic combinatorics of directed lattice paths, Theoret. Comput. Sci., vol. 281, no. 1-2 (2002), pp. 37–80, doi:10.1016/S0304-3975(02)00007-5.
- [27] C. BANDERIER AND M. WALLNER, The kernel method for lattice paths below a line of rational slope, Tech. Report 1606.08412, arXiv, 2016. Available on-line at http://arxiv.org/abs/1606.08412.
- [28] D. E. BARTON AND C. L. MALLOWS, Some aspects of the random sequence, Ann. Math. Statist., vol. 36 (1965), pp. 236–260, doi:10.1214/aoms/1177700286.
- [29] B. BECKERMANN AND G. LABAHN, A uniform approach for the fast computation of matrix-type Padé approximants, SIAM J. Matrix Anal. Appl., vol. 15, no. 3 (1994), pp. 804–823, doi:10.

1137/S0895479892230031.

- [30] A. BENOIT, A. BOSTAN, AND J. VAN DER HOEVEN, Quasi-optimal multiplication of linear differential operators, in Proc. FOCS '12, IEEE, October 2012, pp. 524–530, doi:10.1109/FOCS.2012.57.
- [31] O. BERNARDI, Bijective counting of Kreweras walks and loopless triangulations, J. Combin. Theory Ser. A, vol. 114, no. 5 (2007), pp. 931–956, doi:10.1016/j.jcta.2006.09.009.
- [32] O. BERNARDI, M. BOUSQUET-MÉLOU, AND K. RASCHEL, <u>Counting quadrant walks via Tutte's</u> invariant method, in DMTCS Proceedings of FPSAC'16, Vancouver, Canada, 2016, pp. 203– 214, http://www.lix.polytechnique.fr/~pilaud/FPSAC16/listPapers.html.
- [33] O. BERNARDI, M. BOUSQUET-MÉLOU, AND K. RASCHEL, Counting quadrant walks via Tutte's invariant method, 2017. Preprint, http://arxiv.org/abs/1708.08215.
- [34] J. BERTHOMIEU AND J.-C. FAUGÈRE, Guessing linear recurrence relations of sequence tuples and P-recursive sequences with linear algebra, in Proceedings of the 2016 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2016, pp. 95–102, doi:10.1145/2930889.2930926.
- [35] D. BERTRAND AND F. BEUKERS, <u>Équations différentielles linéaires et majorations de multiplicités</u>, Ann. Sci. École Norm. Sup. (4), vol. 18, no. 1 (1985), pp. 181–192, http://www.numdam.org/ item/ASENS_1985_4_18_1_181_0.
- [36] J. BERTRAND, Solution d'un problème, C. R. Acad. Sci. Paris, vol. 105 (1887), p. 369.
- [37] F. BEUKERS, Padé-approximations in number theory, in Padé approximation and its applications, Amsterdam 1980 (Amsterdam, 1980), vol. 888 of Lecture Notes in Math., Springer, Berlin-New York, 1981, pp. 90–99, doi:10.1007/BFb0095578.
- [38] F. BEUKERS, Irrationality of π^2 , periods of an elliptic curve and $\Gamma_1(5)$, in Diophantine approximations and transcendental numbers (Luminy, 1982), vol. 31 of Progr. Math., Birkhäuser, Boston, Mass., 1983, pp. 47–66.
- [39] F. BEUKERS, Algebraic A-hypergeometric functions, Invent. Math., vol. 180, no. 3 (2010), pp. 589– 610, doi:10.1007/s00222-010-0238-y.
- [40] F. BEUKERS AND G. HECKMAN, Monodromy for the hypergeometric function *nF_{n-1}*, Invent. Math., vol. 95, no. 2 (1989), pp. 325–354, doi:10.1007/BF01393900.
- [41] F. BEUKERS AND C. A. M. PETERS, A family of K3 surfaces and ζ(3), J. Reine Angew. Math., vol. 351 (1984), pp. 42–54, doi:10.1515/crll.1984.351.42.
- [42] P. BIANE, Minuscule weights and random walks on lattices, in Quantum probability & related topics, QP-PQ, VII, World Sci. Publ., River Edge, NJ, 1992, pp. 51–65.
- [43] D. BINI AND V. Y. PAN, Polynomial and matrix computations. Vol. 1, Progress in Theoretical Computer Science, Birkhäuser Boston Inc., Boston, MA, 1994, doi:10.1007/978-1-4612-0265-3. Fundamental algorithms.
- [44] P. BLANCHARD AND D. VOLCHENKOV, Random walks and diffusions on graphs and databases, Springer Series in Synergetics, Springer, Heidelberg, 2011, doi:10.1007/978-3-642-19592-1.
- [45] W. ВÖHM, Lattice path counting and the theory of queues, J. Statist. Plann. Inference, vol. 140, no. 8 (2010), pp. 2168–2183, doi:10.1016/j.jspi.2010.01.013.
- [46] F. BORNEMANN, D. LAURIE, S. WAGON, AND J. WALDVOGEL, <u>The SIAM 100-digit challenge</u>, Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2004, doi:10.1137/1. 9780898717969. A study in high-accuracy numerical computing, With a foreword by David H. Bailey.
- [47] A. A. BOROVKOV AND K. A. BOROVKOV, Asymptotic analysis of random walks, vol. 118 of Encyclopedia of Mathematics and its Applications, Cambridge University Press, Cambridge, 2008, doi:10.1017/CBO9780511721397. Heavy-tailed distributions, Translated from the Russian by O. B. Borovkova.
- [48] J. BORWEIN AND D. BAILEY, Mathematics by experiment, A K Peters, Ltd., Wellesley, MA, second ed., 2008, doi:10.1201/b10704. Plausible reasoning in the 21st Century.
- [49] J. BORWEIN, D. BAILEY, AND R. GIRGENSOHN, Experimentation in mathematics, A K Peters, Ltd., Natick, MA, 2004. Computational paths to discovery.
- [50] J. BORWEIN, P. BORWEIN, R. GIRGENSOHN, AND S. PARNES, <u>Making sense of experimental</u> mathematics, Math. Intelligencer, vol. 18, no. 4 (1996), pp. 12–18, doi:10.1007/BF03026747.
- [51] J. BORWEIN AND K. DEVLIN, The computer as crucible, A K Peters, Ltd., Wellesley, MA, 2009, doi:10.1201/b10684. An introduction to experimental mathematics.
- [52] J. M. BORWEIN, Exploratory experimentation: digitally-assisted discovery and proof, Austral. Math. Soc. Gaz., vol. 36, no. 3 (2009), pp. 166–175, doi:10.1007/978-94-007-2129-6_4.
- [53] J. M. BORWEIN AND R. M. CORLESS, <u>Emerging tools for experimental mathematics</u>, Amer. Math. Monthly, vol. 106, no. 10 (1999), pp. 889–909, doi:10.2307/2589743.
- [54] J. M. BORWEIN AND I. J. ZUCKER, Fast evaluation of the gamma function for small rational fractions using complete elliptic integrals of the first kind, IMA J. Numer. Anal., vol. 12, no. 4 (1992), pp. 519–526, doi:10.1093/imanum/12.4.519.

- [55] A. BOSTAN, Algorithmique efficace pour des opérations de base en Calcul formel, PhD thesis, École polytechnique, 2003, http://algo.inria.fr/bostan/these/These.pdf. 246 pages.
- [56] A. BOSTAN, Algorithmes rapides pour les polynômes, séries formelles et matrices, Les cours du CIRM, vol. 1, no. 2 (2010), pp. 75–262, doi:10.5802/ccirm.9. Actes des JNCF'10.
- [57] A. BOSTAN, S. BOUKRAA, G. CHRISTOL, S. HASSANI, AND J.-M. MAILLARD, Ising n-fold integrals as diagonals of rational functions and integrality of series expansions, J. Phys. A, vol. 46, no. 18 (2013), pp. 185202, 44, doi:10.1088/1751-8113/46/18/185202.
- [58] A. BOSTAN, S. BOUKRAA, S. HASSANI, J.-M. MAILLARD, J.-A. WEIL, AND N. ZENINE, <u>Globally</u> nilpotent differential operators and the square Ising model, J. Phys. A: Math. Theor., vol. 42, no. 12 (2009), pp. 125206, 50, doi:10.1088/1751-8113/42/12/125206.
- [59] A. BOSTAN, S. BOUKRAA, J.-M. MAILLARD, AND J.-A. WEIL, Diagonals of rational functions and selected differential Galois groups, J. Phys. A, vol. 48, no. 50 (2015), pp. 504001, 29, doi:10. 1088/1751-8113/48/50/504001.
- [60] A. BOSTAN, M. BOUSQUET-MÉLOU, M. KAUERS, AND S. MELCZER, On 3-dimensional lattice walks confined to the positive octant, Ann. Comb., vol. 20, no. 4 (2016), pp. 661–704, doi:10.1007/ s00026-016-0328-7.
- [61] A. BOSTAN, M. BOUSQUET-MÉLOU, AND S. MELCZER, <u>Walks with large steps in an orthant</u>. In preparation.
- [62] A. BOSTAN, X. CARUSO, AND E. SCHOST, <u>A fast algorithm for computing the characteristic polynomial of the *p*-curvature, in Proc. ISSAC'14, ACM, 2014, pp. 59–66, doi:10.1145/2608628.2608650.</u>
- [63] A. BOSTAN, X. CARUSO, AND E. SCHOST, A fast algorithm for computing the *p*-curvature, in Proc. ISSAC'15, ACM, 2015, pp. 69–76, doi:10.1145/2755996.2756674.
- [64] A. BOSTAN, X. CARUSO, AND E. SCHOST, Computation of the similarity class of the *p*-curvature, in Proc. ISSAC'16, ACM, 2016, pp. 111–118, doi:10.1145/2930889.2930897.
- [65] A. BOSTAN, S. CHEN, F. CHYZAK, AND Z. LI, Complexity of creative telescoping for bivariate rational functions, in Proc. ISSAC'10, ACM, 2010, pp. 203–210, doi:10.1145/1837934.1837975.
- [66] A. BOSTAN, S. CHEN, F. CHYZAK, Z. LI, AND G. XIN, Hermite reduction and creative telescoping for hyperexponential functions, in Proc. ISSAC'13, ACM, 2013, pp. 77–84, doi:10.1145/2465506. 2465946.
- [67] A. BOSTAN, M. F. I. CHOWDHURY, R. LEBRETON, B. SALVY, AND E. SCHOST, Power series solutions of singular (q)-differential equations, in ISSAC 2012—Proceedings of the 37th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2012, pp. 107–114, doi:10.1145/2442829.2442848.
- [68] A. BOSTAN, G. CHRISTOL, AND P. DUMAS, Fast computation of the Nth term of an algebraic series over a finite prime field, in Proc. ISSAC'16, ACM, 2016, pp. 119–126, doi:10.1145/2930889. 2930904.
- [69] A. BOSTAN, F. CHYZAK, T. CLUZEAU, AND B. SALVY, Low complexity algorithms for linear recurrences, in Proc. ISSAC'06, ACM, 2006, pp. 31–38, doi:10.1145/1145768.1145781.
- [70] A. BOSTAN, F. CHYZAK, M. GIUSTI, R. LEBRETON, G. LECERF, B. SALVY, AND E. SCHOST, Algorithmes efficaces en calcul formel. Lecture notes (in French) for the course at Master Parisien de Recherche en Informatique. Prepublication as of 2017-01-11, soon available as a printed book, 2017, http://hal.archives-ouvertes.fr/AECF.
- [71] A. BOSTAN, F. CHYZAK, AND N. LE ROUX, Products of ordinary differential operators by evaluation and interpolation, in Proc. ISSAC'08, ACM, 2008, pp. 23–30, doi:10.1145/1390768.1390775.
- [72] A. BOSTAN, F. CHYZAK, G. LECERF, B. SALVY, AND E. SCHOST, Differential equations for algebraic functions, in Proc. ISSAC'07, ACM, 2007, pp. 25–32, doi:10.1145/1277548.1277553.
- [73] A. BOSTAN, F. CHYZAK, AND A. MAHBOUBI, <u>A bijection between classes of North-West-South-East</u> walks. In preparation.
- [74] A. BOSTAN, F. CHYZAK, F. OLLIVIER, B. SALVY, É. SCHOST, AND A. SEDOGLAVIC, Fast computation of power series solutions of systems of differential equations, in Proc. of SODA'07, SIAM, 2007, pp. 1012–1021, http://dl.acm.org/citation.cfm?id=1283383.1283492.
- [75] A. BOSTAN, F. CHYZAK, B. SALVY, AND Z. LI, Fast computation of common left multiples of linear ordinary differential operators, in Proc. ISSAC'12, ACM, 2012, pp. 99–106, doi:10. 1145/2442829.2442847.
- [76] A. BOSTAN, F. CHYZAK, M. VAN HOEIJ, M. KAUERS, AND L. PECH, <u>Hypergeometric expressions for</u> generating functions of walks with small steps in the quarter plane, European J. Combin., vol. 61 (2017), pp. 242–275, doi:10.1016/j.ejc.2016.10.010.
- [77] A. BOSTAN, F. CHYZAK, M. VAN HOEIJ, AND L. PECH, Explicit formula for the generating series of diagonal 3D rook paths, Sém. Lothar. Combin., vol. 66 (2011), pp. Art. B66a, 27 pp., http://www.mat.univie.ac.at/~slc/s/s66bochhope.html.
- [78] A. BOSTAN, L. DUMONT, AND B. SALVY, Algebraic diagonals and walks, in ISSAC'15-Proceedings

of the 2015 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2015, pp. 77–84, doi:10.1145/2755996.2756663.

- [79] A. BOSTAN, L. DUMONT, AND B. SALVY, Efficient algorithms for mixed creative telescoping, in Proceedings of the 2016 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2016, pp. 127–134, doi:10.1145/2930889.2930907.
- [80] A. BOSTAN, L. DUMONT, AND B. SALVY, <u>Algebraic diagonals and walks</u>: algorithms, bounds, <u>complexity</u>, J. Symbolic Comput., vol. 83 (2017), pp. 68–92, doi:10.1016/j.jsc.2016.11.006.
- [81] A. BOSTAN, P. FLAJOLET, B. SALVY, AND É. SCHOST, Fast computation of special resultants, Journal of Symbolic Computation, vol. 41, no. 1 (2006), pp. 1–29, doi:10.1016/j.jsc.2005.07.001.
- [82] A. BOSTAN, C.-P. JEANNEROD, C. MOUILLERON, AND E. SCHOST, On Matrices With Displacement Structure: Generalized Operators and Faster Algorithms, SIAM J. Matrix Anal. Appl., vol. 38, no. 3 (2017), pp. 733–775, doi:10.1137/16M1062855.
- [83] A. BOSTAN, C.-P. JEANNEROD, AND É. SCHOST, Solving structured linear systems with large displacement rank, Theoret. Comput. Sci., vol. 407, no. 1-3 (2008), pp. 155–181, doi:10.1016/ j.tcs.2008.05.014.
- [84] A. BOSTAN AND M. KAUERS, Automatic classification of restricted lattice walks, in DMTCS Proc. FPSAC'09, 2009, pp. 203–217, http://dmtcs.episciences.org/2724.
- [85] A. BOSTAN AND M. KAUERS, The complete generating function for Gessel walks is algebraic, Proc. the American Mathematical Society, vol. 138, no. 9 (2010), pp. 3063–3078, doi:10.1090/ S0002-9939-2010-10398-2. With an Appendix by M. van Hoeij.
- [86] A. BOSTAN, I. KURKOVA, AND K. RASCHEL, A human proof of Gessel's lattice path conjecture, Trans. Amer. Math. Soc., vol. 369, no. 2 (2017), pp. 1365–1393, doi:10.1090/tran/6804.
- [87] A. BOSTAN, P. LAIREZ, AND B. SALVY, Creative telescoping for rational functions using the Griffiths-Dwork method, in Proc. ISSAC'13, ACM, 2013, pp. 93–100, doi:10.1145/2465506. 2465935.
- [88] A. BOSTAN, P. LAIREZ, AND B. SALVY, Multiple binomial sums, J. Symbolic Comput., vol. 80, no. 2 (2017), pp. 351–386, doi:10.1016/j.jsc.2016.04.002.
- [89] A. BOSTAN AND K. RASCHEL, Compter les excursions sur un échiquier, Pour la Science, vol. 449 (2015), pp. 40-46, http://www.pourlascience.fr/ewb_pages/a/ article-compter-les-excursions-sur-un-echiquier-33883.php.
- [90] A. BOSTAN, K. RASCHEL, AND B. SALVY, Non-D-finite excursions in the quarter plane, J. Combin. Theory Ser. A, vol. 121 (2014), pp. 45–63, doi:10.1016/j.jcta.2013.09.005.
- [91] A. BOSTAN AND E. SCHOST, Polynomial evaluation and interpolation on special sets of points, J. Complexity, vol. 21, no. 4 (2005), pp. 420–446, doi:10.1016/j.jco.2004.09.009.
- [92] A. BOSTAN AND É. SCHOST, Fast algorithms for differential equations in positive characteristic, in Proc. ISSAC'09, ACM, 2009, pp. 47–54, doi:10.1145/1576702.1576712.
- [93] M. BOUSQUET-MÉLOU, Walks on the slit plane: other approaches, Adv. in Appl. Math., vol. 27, no. 2-3 (2001), pp. 243–288, doi:10.1006/aama.2001.0734. Special issue in honor of Dominique Foata's 65th birthday (Philadelphia, PA, 2000).
- [94] M. BOUSQUET-MÉLOU, <u>Counting walks in the quarter plane</u>, in Mathematics and computer science, II (Versailles, 2002), Trends Math., Birkhäuser, Basel, 2002, pp. 49–67.
- [95] M. BOUSQUET-MÉLOU, Walks in the quarter plane: a functional equation approach, in Proceedings of FPSAC'02, Melbourne, Australia, 2002. Available on-line at http://igm.univ-mlv.fr/ ~fpsac/FPSAC02/articles.html.
- [96] M. BOUSQUET-MÉLOU, Walks in the quarter plane: Kreweras' algebraic model, Ann. Appl. Probab., vol. 15, no. 2 (2005), pp. 1451–1491, http://www.jstor.org/stable/30038361.
- [97] M. BOUSQUET-MÉLOU, Rational and algebraic series in combinatorial enumeration, in International Congress of Mathematicians. Vol. III, Eur. Math. Soc., Zürich, 2006, pp. 789–826.
- [98] M. BOUSQUET-MÉLOU, An elementary solution of Gessel's walks in the quadrant, Adv. Math., vol. 303 (2016), pp. 1171–1189, doi:10.1016/j.aim.2016.08.038.
- [99] M. BOUSQUET-MÉLOU, Square lattice walks avoiding a quadrant, J. Combin. Theory Ser. A, vol. 144 (2016), pp. 37–79, doi:10.1016/j.jcta.2016.06.010.
- [100] M. BOUSQUET-MÉLOU AND A. JEHANNE, Polynomial equations with one catalytic variable, algebraic series and map enumeration, J. Combin. Theory Ser. B, vol. 96, no. 5 (2006), pp. 623– 672, doi:10.1016/j.jctb.2005.12.003.
- [101] M. BOUSQUET-MÉLOU AND M. MISHNA, Walks with small steps in the quarter plane, in Algorithmic probability and combinatorics, vol. 520 of Contemp. Math., Amer. Math. Soc., Providence, RI, 2010, pp. 1–39.
- [102] M. BOUSQUET-MÉLOU AND M. PETKOVŠEK, Linear recurrences with constant coefficients: the multivariate case, Discrete Math., vol. 225, no. 1-3 (2000), pp. 51–75, doi:10.1016/ S0012-365X(00)00147-3. Formal power series and algebraic combinatorics (Toronto, ON, 1998).

- [103] M. BOUSQUET-MÉLOU AND M. PETKOVŠEK, Walks confined in a quadrant are not always D-finite, Theoret. Comput. Sci., vol. 307, no. 2 (2003), pp. 257–276, doi:10.1016/S0304-3975(03)00219-6.
- [104] M. BOUSQUET-MÉLOU AND G. SCHAEFFER, Counting paths on the slit plane (extended abstract), in Mathematics and computer science (Versailles, 2000), Trends Math., Birkhäuser, Basel, 2000, pp. 101–112.
- [105] M. BOUSQUET-MÉLOU AND G. SCHAEFFER, Walks on the slit plane, Probab. Theory Related Fields, vol. 124, no. 3 (2002), pp. 305–344, doi:10.1007/s004400200205.
- [106] R. BRAK AND A. J. GUTTMANN, Algebraic approximants: a new method of series analysis, J. Phys. A, vol. 23, no. 24 (1990), pp. L1331–L1337, doi:10.1088/0305-4470/23/24/008.
- [107] R. P. BRENT AND H. T. KUNG, Fast algorithms for manipulating formal power series, J. Assoc. Comput. Mach., vol. 25, no. 4 (1978), pp. 581–595, doi:10.1145/322092.322099.
- [108] R. P. BRENT AND J. F. TRAUB, On the complexity of composition and generalized composition of power series, SIAM J. Comput., vol. 9, no. 1 (1980), pp. 54–66, doi:10.1137/0209004.
- [109] C. BREZINSKI, Accélération de la convergence en analyse numérique, Lecture Notes in Mathematics, Vol. 584, Springer-Verlag, Berlin-New York, 1977, doi:10.1007/BFb0089363.
- [110] C. BREZINSKI, Convergence acceleration during the 20th century, J. Comput. Appl. Math., vol. 122, no. 1-2 (2000), pp. 1–21, doi:10.1016/S0377-0427(00)00360-5. Numerical analysis 2000, Vol. II: Interpolation and extrapolation.
- [111] M. BRONSTEIN, Formulas for series computations, Appl. Algebra Engrg. Comm. Comput., vol. 2, no. 3 (1992), pp. 195–206, doi:10.1007/BF01294333.
- [112] M. BRONSTEIN AND M. PETKOVŠEK, <u>An introduction to pseudo-linear algebra</u>, Theoret. Comput. Sci., vol. 157, no. 1 (1996), pp. 3–33, doi:10.1016/0304-3975(95)00173-5. Algorithmic complexity of algebraic and geometric models (Creteil, 1994).
- [113] P. BÜRGISSER, M. CLAUSEN, AND M. A. SHOKROLLAHI, <u>Algebraic complexity theory</u>, Springer-Verlag, 1997, doi:10.1007/978-3-662-03338-8.
- [114] P. CARTIER, Démonstration "automatique" d'identités et fonctions hypergéométriques (d'après D. Zeilberger), Astérisque, no. 206 (1992), pp. Exp. No. 746, 3, 41–91, http://www.numdam. org/item/SB_1991-1992_34_41_0. Séminaire Bourbaki, Vol. 1991/92.
- [115] E. C. CATALAN, Note sur une équation aux différences finies, J. Math. Pures Appl., vol. 3 (1838), pp. 508–515, http://sites.mathdoc.fr/JMPA/PDF/JMPA_1838_1_3_A40_0.pdf.
 [116] E. C. CATALAN, Solution nouvelle de cette question : Un polygône étant donné, de combien de
- [116] E. C. CATALAN, Solution nouvelle de cette question : Un polygône étant donné, de combien de manières peut-on le partager en triangles au moyen de diagonales ?, J. Math. Pures Appl., vol. 4 (1839), pp. 91–94, http://sites.mathdoc.fr/JMPA/PDF/JMPA_1839_1_4_A7_0.pdf.
- [117] A. CHAMBERT-LOIR, Théorèmes d'algébricité en géométrie diophantienne (d'après J.-B. Bost, Y. André, D. & G. Chudnovsky), Astérisque, no. 282 (2002), pp. Exp. No. 886, viii, 175–209, http: //www.numdam.org/item/SB_2000-2001_43_175_0. Séminaire Bourbaki, Vol. 2000/2001.
- [118] S. CHEN, H. HUANG, M. KAUERS, AND Z. LI, A modified Abramov-Petkovšek reduction and creative telescoping for hypergeometric terms, in ISSAC'15—Proceedings of the 2015 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2015, pp. 117–124, doi:10.1145/2755996.2756648.
- [119] S. CHEN, M. JAROSCHEK, M. KAUERS, AND M. F. SINGER, Desingularization explains order-degree curves for Ore operators, in ISSAC 2013—Proceedings of the 38th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2013, pp. 157–164, doi:10.1145/ 2465506.2465510.
- [120] S. CHEN AND M. KAUERS, Order-degree curves for hypergeometric creative telescoping, in IS-SAC 2012—Proceedings of the 37th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2012, pp. 122–129, doi:10.1145/2442829.2442850.
- [121] S. CHEN AND M. KAUERS, Trading order for degree in creative telescoping, J. Symbolic Comput., vol. 47, no. 8 (2012), pp. 968–995, doi:10.1016/j.jsc.2012.02.002.
- [122] S. CHEN AND M. KAUERS, Some open problems related to creative telescoping, J. Syst. Sci. Complex., vol. 30, no. 1 (2017), pp. 154–172, doi:10.1007/s11424-017-6202-9.
- [123] S. CHEN, M. KAUERS, AND C. KOUTSCHAN, A generalized Apagodu-Zeilberger algorithm, in IS-SAC 2014—Proceedings of the 39th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2014, pp. 107–114, doi:10.1145/2608628.2608641.
- [124] S. CHEN, M. KAUERS, AND C. KOUTSCHAN, Reduction-based creative telescoping for algebraic functions, in Proceedings of the 2016 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2016, pp. 175–182, doi:10.1145/2930889.2930901.
- [125] S. CHEN, M. KAUERS, AND M. F. SINGER, <u>Telescopers for rational and algebraic functions via</u> residues, in ISSAC 2012—Proceedings of the 37th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2012, pp. 130–137, doi:10.1145/2442829.2442851.
- [126] S. CHEN, M. KAUERS, AND M. F. SINGER, Desingularization of Ore operators, J. Symbolic Comput., vol. 74 (2016), pp. 617–626, doi:10.1016/j.jsc.2015.11.001.

- [127] S. CHEN, M. VAN HOEIJ, M. KAUERS, AND C. KOUTSCHAN, Reduction-based creative telescoping for fuchsian D-finite functions, J. Symbolic Comput., (2017), doi:10.1016/j.jsc.2017.07.005. In press.
- [128] S. CHEN, M. VAN HOEIJ, M. KAUERS, AND C. KOUTSCHAN, <u>Reduction-based creative telescoping</u> for fuchsian D-finite functions, J. Symbolic Comput., vol. 85 (2018), pp. 108–127, doi:10.1016/ j.jsc.2017.07.005.
- [129] S. CHOI, Counting lattice paths in restricted planes, Discrete Math., vol. 212, no. 3 (2000), pp. 191– 198, doi:10.1016/S0012-365X(99)00287-3. Combinatorics and applications (Tianjin, 1996).
- [130] G. CHRISTOL, Diagonales de fractions rationnelles et équations différentielles, in Study group on ultrametric analysis, 10th year: 1982/83, No. 2, Exp. No. 18, Inst. Henri Poincaré, Paris, 1984, pp. 1–10, http://archive.numdam.org/item/GAU_1982-1983_10_2_A4_0.
- [131] G. CHRISTOL, Diagonales de fractions rationnelles et équations de Picard-Fuchs, in Study group on ultrametric analysis, 12th year, 1984/85, No. 1, Exp. No. 13, Secrétariat Math., Paris, 1985, pp. 1–12, http://archive.numdam.org/item/GAU_1984-1985_12_1_A8_0.
- [132] G. CHRISTOL, <u>Diagonales de fractions rationnelles</u>, in Séminaire de Théorie des Nombres, Paris 1986–87, vol. 75 of Progr. Math., Birkhäuser Boston, 1988, pp. 65–90.
- [133] G. CHRISTOL, Globally bounded solutions of differential equations, in Analytic number theory (Tokyo, 1988), vol. 1434 of Lecture Notes in Math., Springer, Berlin, 1990, pp. 45–64, doi:10. 1007/BFb0097124.
- [134] G. CHRISTOL, Diagonals of rational fractions, Eur. Math. Soc. Newsl., no. 97 (2015), pp. 37-43.
- [135] D. V. CHUDNOVSKY AND G. V. CHUDNOVSKY, Applications of Padé approximations to Diophantine inequalities in values of *G*-functions, in Number theory (New York, 1983–84), vol. 1135 of Lecture Notes in Math., Springer, Berlin, 1985, pp. 9–51, doi:10.1007/BFb0074600.
- [136] D. V. CHUDNOVSKY AND G. V. CHUDNOVSKY, On expansion of algebraic functions in power and Puiseux series. I, J. Complexity, vol. 2, no. 4 (1986), pp. 271–294, doi:10.1016/0885-064X(86) 90006-3.
- [137] D. V. CHUDNOVSKY AND G. V. CHUDNOVSKY, On expansion of algebraic functions in power and Puiseux series. II, J. Complexity, vol. 3, no. 1 (1987), pp. 1–25, doi:10.1016/0885-064X(87) 90002-1.
- [138] G. V. CHUDNOVSKY, Padé approximation and the Riemann monodromy problem, in Bifurcation phenomena in mathematical physics and related topics (Proc. NATO Advanced Study Inst., Cargèse, 1979), vol. 54 of NATO Adv. Study Inst. Ser., Ser. C: Math. Phys. Sci., Reidel, Dordrecht-Boston, Mass., 1980, pp. 449–510.
- [139] G. V. CHUDNOVSKY, Rational and Padé approximations to solutions of linear differential equations and the monodromy theory, in Complex analysis, microlocal calculus and relativistic quantum theory (Proc. Internat. Colloq., Centre Phys., Les Houches, 1979), vol. 126 of Lecture Notes in Phys., Springer, Berlin-New York, 1980, pp. 136–169, doi:10.1007/3-540-09996-4_36.
- [140] F. CHYZAK, Fonctions holonomes en calcul formel, phd thesis, École polytechnique. INRIA, TU 0531., 1998.
- [141] F. Снуzак, Gröbner bases, symbolic summation and symbolic integration, in Gröbner bases and applications (Linz, 1998), vol. 251 of London Math. Soc. Lecture Note Ser., Cambridge Univ. Press, Cambridge, 1998, pp. 32–60.
- [142] F. CHYZAK, An extension of Zeilberger's fast algorithm to general holonomic functions, Discrete Math., vol. 217, no. 1-3 (2000), pp. 115–134, doi:10.1016/S0012-365X(99)00259-9. Formal power series and algebraic combinatorics (Vienna, 1997).
- [143] F. CHYZAK, The ABC of Creative Telescoping Algorithms, Bounds, Complexity, accreditation to supervise research (HDR), École polytechnique, Apr. 2014. Available at http://hal.inria. fr/tel-01069831.
- [144] F. CHYZAK AND B. SALVY, Non-commutative elimination in Ore algebras proves multivariate identities, J. Symbolic Comput., vol. 26, no. 2 (1998), pp. 187–227, doi:10.1006/jsco.1998.0207.
- [145] J. COCKLE, On transcendental and algebraic solution, Philosophical Magazine, vol. XXI (1861), pp. 379–383.
- [146] J. W. COHEN, Analysis of random walks, vol. 2 of Studies in Probability, Optimization and Statistics, IOS Press, Amsterdam, 1992, http://www.iospress.nl/book/analysis-of-random-walks.
- [147] С. Сокев, Enumerating a class of lattice paths, Discrete Math., vol. 271, no. 1-3 (2003), pp. 13–28, doi:10.1016/S0012-365X(03)00037-2.
- [148] L. COMTET, Calcul pratique des coefficients de Taylor d'une fonction algébrique, Enseignement Math. (2), vol. 10 (1964), pp. 267–270, doi:10.5169/seals-39424.
- [149] O. CORMIER, M. F. SINGER, B. M. TRAGER, AND F. ULMER, Linear differential operators for polynomial equations, J. Symbolic Comput., vol. 34, no. 5 (2002), pp. 355–398, doi:10.1006/ jsco.2002.0564.
- [150] J. COURTIEL, S. MELCZER, M. MISHNA, AND K. RASCHEL, Weighted lattice walks and universality

<u>classes</u>, J. Combin. Theory Ser. A, vol. 152 (2017), pp. 255–302, doi:10.1016/j.jcta.2017.06.008.
 [151] D. Cox, J. LITTLE, AND D. O'SHEA, <u>Using algebraic geometry</u>, vol. 185 of Graduate Texts in Mathematics, Springer-Verlag, New York, 1998, doi:10.1007/b138611.

- [152] Е. CSÁKI, S. G. MOHANTY, AND J. SARAN, <u>On random walks in a plane</u>, Ars Combin., vol. 29 (1990), pp. 309–318.
- [153] P. D'ARCO, V. LACIVITA, AND S. MUSTAPHA, <u>Combinatorics meets potential theory</u>, Electron. J. Combin., vol. 23, no. 2 (2016), pp. Paper 2.28, 17, http://www.combinatorics.org/ojs/index. php/eljc/article/view/v23i2p28.
- [154] P. DELIGNE, Intégration sur un cycle évanescent, Invent. Math., vol. 76, no. 1 (1984), pp. 129–143, doi:10.1007/BF01388496.
- [155] J. DENEF AND L. LIPSHITZ, Algebraic power series and diagonals, J. Number Theory, vol. 26, no. 1 (1987), pp. 46–67, http://dx.doi.org/10.1016/0022-314X(87)90095-3.
- [156] D. DENISOV AND V. WACHTEL, Random walks in cones, Ann. Probab., vol. 43, no. 3 (2015), pp. 992– 1044, doi:10.1214/13-AOP867.
- [157] K. DEVLIN, <u>The unfinished game</u>, Basic Books, New York, 2008. Pascal, Fermat, and the seventeenth-century letter that made the world modern.
- [158] A. C. DIXON, On the sum of the cubes of the coefficients in a certain expansion by the binomial theorem, Messenger of mathematics, vol. 20 (1891), pp. 79–80.
- [159] G. DOETSCH, Integraleigenschaften der Hermiteschen Polynome, Math. Z., vol. 32, no. 1 (1930), pp. 587–599, doi:10.1007/BF01194654.
- [160] P. G. DOYLE AND J. L. SNELL, Random walks and electric networks, vol. 22 of Carus Mathematical Monographs, Mathematical Association of America, Washington, DC, 1984, doi:10.5948/ UPO9781614440222.
- [161] T. DREYFUS, C. HARDOUIN, J. ROQUES, AND M. F. SINGER, On the nature of the generating series of walks in the quarter plane, 2017. Preprint, http://arxiv.org/abs/1702.04696.
- [162] D. K. DU, Q.-H. HOU, AND R.-H. WANG, Infinite orders and non-D-finite property of 3-dimensional lattice walks, Electron. J. Combin., vol. 23, no. 3 (2016), pp. Paper 3.38, 15, http://www. combinatorics.org/ojs/index.php/eljc/article/view/v23i3p38.
- [163] P. DUCHON, On the enumeration and generation of generalized Dyck words, Discrete Math., vol. 225, no. 1-3 (2000), pp. 121–135, doi:10.1016/S0012-365X(00)00150-3. Formal power series and algebraic combinatorics (Toronto, ON, 1998).
- [164] J. DURAJ, Random walks in cones: the case of nonzero drift, Stochastic Process. Appl., vol. 124, no. 4 (2014), pp. 1503–1518, doi:10.1016/j.spa.2013.12.003.
- [165] B. DWORK, On the zeta function of a hypersurface, IHES Publ. Math., no. 12 (1962), pp. 5–68.
- [166] В. DWORK, On the zeta function of a hypersurface. II, Ann. of Math. (2), vol. 80 (1964), pp. 227–299.
 [167] В. DWORK, Differential operators with nilpotent *p*-curvature, Amer. J. Math., vol. 112, no. 5 (1990),
- pp. 749–786, doi:10.2307/2374806. [168] B. DWORK, G. GEROTTO, AND F. J. SULLIVAN, An introduction to G-functions, vol. 133 of An-
- nals of Mathematics Studies, Princeton University Press, Princeton, NJ, 1994, http://press. princeton.edu/titles/5601.html.
- [169] B. DWORK AND F. LOESER, <u>Hypergeometric series</u>, Japan. J. Math. (N.S.), vol. 19, no. 1 (1993), pp. 81–129.
- [170] A. W. F. EDWARDS, Pascal's arithmetical triangle, Charles Griffin & Co., Ltd., London; The Clarendon Press, Oxford University Press, New York, 1987.
- [171] G. P. EGORYCHEV, Integral representation and the computation of combinatorial sums, vol. 59 of Translations of Mathematical Monographs, American Mathematical Society, Providence, RI, 1984, http://bookstore.ams.org/mmono-59. Translated from the Russian by H. H. McFadden, Translation edited by Lev J. Leifman.
- [172] G. EISENSTEIN, Über eine allgemeine Eigenschaft der Reihen-Entwicklungen aller algebraischen Funktionen, Bericht Königl. Preuss. Akad. Wiss., (1852), pp. 441–443.
- [173] M. ERICKSON, S. FERNANDO, AND K. TRAN, <u>Enumerating rook and queen paths</u>, Bull. Inst. Combin. Appl., vol. 60 (2010), pp. 37–48.
- [174] A. ERRERA, Zahlentheoretische Lösung einer functionentheoretischen Frage, Rend. Circ. Mat. Palermo, vol. 35 (1913), pp. 107–144.
- [175] L. EULER, Specimen de constructione aequationum differentialium sine indeterminatarum separatione, J. Math. Anal. Appl., vol. 6 (1733), pp. 168–174, doi:http://eulerarchive.maa. org/pages/E028.html.
- [176] M. C. FASENMYER, Some generalized hypergeometric polynomials, Bull. Amer. Math. Soc., vol. 53 (1947), pp. 806–812, doi:10.1090/S0002-9904-1947-08893-5.
- [177] G. FAYOLLE AND R. IASNOGORODSKI, Solutions of functional equations arising in the analysis of two-server queueing models, in Performance of computer systems (Proc. Fourth Internat. Sympos. Modelling Performance Evaluation Comput. Systems, Vienna, 1979), North-

Holland, Amsterdam-New York, 1979, pp. 289–303.

- [178] G. FAYOLLE AND R. IASNOGORODSKI, <u>Two coupled processors: the reduction to a Riemann-Hilbert problem</u>, Z. Wahrsch. Verw. Gebiete, vol. 47, no. 3 (1979), pp. 325–351, doi:10.1007/ BF00535168.
- [179] G. FAYOLLE AND R. IASNOGORODSKI, <u>Random walks in the quarter-plane: advances in explicit</u> criterions for the finiteness of the associated group in the genus 1 case, Markov Process. Related Fields, vol. 21, no. 4 (2015), pp. 1005–1032, <u>http://math-mprf.org/journal/articles/id1402</u>.
- [180] G. FAYOLLE, R. IASNOGORODSKI, AND V. MALYSHEV, Random walks in the quarter plane, vol. 40 of Probability Theory and Stochastic Modelling, Springer, second ed., 2017, doi:10.1007/ 978-3-319-50930-3. Algebraic methods, boundary value problems, applications to queueing systems and analytic combinatorics.
- [181] G. FAYOLLE AND K. RASCHEL, On the holonomy or algebraicity of generating functions counting lattice walks in the quarter-plane, Markov Process. Related Fields, vol. 16, no. 3 (2010), pp. 485–496, http://math-mprf.org/journal/articles/id1219.
- [182] G. FAYOLLE AND K. RASCHEL, Random walks in the quarter plane with zero drift: an explicit criterion for the finiteness of the associated group, Markov Process. Related Fields, vol. 17, no. 4 (2011), pp. 619–636, http://math-mprf.org/journal/articles/id1254.
- [183] G. FAYOLLE AND K. RASCHEL, Some exact asymptotics in the counting of walks in the quarter plane, in 23rd Intern. Meeting on Probabilistic, Combinatorial, and Asymptotic Methods for the Analysis of Algorithms (AofA'12), Montreal, Canada, Discrete Math. Theor. Comput. Sci. Proc., AQ, Assoc. Discrete Math. Theor. Comput. Sci., Nancy, 2012, pp. 109–124, http: //dmtcs.episciences.org/2988.
- [184] G. FAYOLLE AND K. RASCHEL, About a possible analytic approach for walks in the quarter plane with arbitrary big jumps, C. R. Math. Acad. Sci. Paris, vol. 353, no. 2 (2015), pp. 89–94, doi:10.1016/j.crma.2014.11.015.
- [185] W. FELLER, An introduction to probability theory and its applications. Vol. I, Third edition, John Wiley & Sons Inc., New York, 1968.
- [186] H. R. P. FERGUSON, D. H. BAILEY, AND S. ARNO, <u>Analysis of PSLQ</u>, an integer relation finding algorithm, Math. Comp., vol. 68, no. 225 (1999), pp. 351–369, doi:10.1090/ S0025-5718-99-00995-3.
- [187] M. FILASETA, A new method for solving a class of ballot problems, J. Combin. Theory Ser. A, vol. 39, no. 1 (1985), pp. 102–111, doi:10.1016/0097-3165(85)90085-8.
- [188] P. FLAJOLET, <u>Analytic models and ambiguity of context-free languages</u>, Theoret. Comput. Sci., vol. 49, no. 2-3 (1987), pp. 283–309, doi:10.1016/0304-3975(87)90011-9. Twelfth international colloquium on automata, languages and programming (Nafplion, 1985).
- [189] P. FLAJOLET AND R. SEDGEWICK, Analytic combinatorics, Cambridge University Press, Cambridge, 2009, doi:10.1017/CBO9780511801655.
- [190] L. FLATTO AND S. HAHN, Two parallel queues created by arrivals with two demands I, SIAM J. Appl. Math., vol. 44, no. 5 (1984), pp. 1041–1053, doi:10.1137/0144074.
- [191] L. FLATTO AND H. P. MCKEAN, Two queues in parallel, Comm. Pure Appl. Math., vol. 30, no. 2 (1977), pp. 255–263, doi:10.1002/cpa.3160300206.
- [192] L. FUCHS, Zur Theorie der linearen Differentialgleichungen mit veränderlichen Coefficienten, J. Reine Angew. Math., vol. 66 (1866), pp. 121–160, doi:10.1515/crll.1866.66.121.
- [193] L. FUCHS, Die Periodicitätsmoduln der hyperelliptischen Integrale als Functionen eines Parameters aufgefasst, J. Reine Angew. Math., vol. 71 (1870), pp. 91–127, doi:10.1515/crll. 1870.71.91.
- [194] H. FURSTENBERG, Algebraic functions over finite fields, J. Algebra, vol. 7 (1967), pp. 271–277.
- [195] A. GALLIGO, Some algorithmic questions on ideals of differential operators, in EUROCAL '85, Vol. 2 (Linz, 1985), vol. 204 of Lecture Notes in Comput. Sci., Springer, Berlin, 1985, pp. 413– 421, doi:10.1007/3-540-15984-3_301.
- [196] R. GARBIT AND K. RASCHEL, On the exit time from a cone for random walks with drift, Rev. Mat. Iberoam., vol. 32, no. 2 (2016), pp. 511–532, doi:10.4171/RMI/893.
- [197] S. GAROUFALIDIS, G-functions and multisum versus holonomic sequences, Adv. Math., vol. 220, no. 6 (2009), pp. 1945–1955, doi:10.1016/j.aim.2008.11.012.
- [198] I. M. GEL' FAND, M. I. GRAEV, AND A. V. ZELEVINSKIJ, Holonomic systems of equations and series of hypergeometric type, Dokl. Akad. Nauk SSSR, vol. 295, no. 1 (1987), pp. 14–19.
- [199] I. M. GEL' FAND, M. M. KAPRANOV, AND A. V. ZELEVINSKY, Generalized Euler integrals and <u>A-hypergeometric functions</u>, Adv. Math., vol. 84, no. 2 (1990), pp. 255–271, doi:10.1016/ 0001-8708(90)90048-R.
- [200] I. M. GEL' FAND, A. V. ZELEVINSKIĬ, AND M. M. KAPRANOV, Equations of hypergeometric type and Newton polyhedra, Dokl. Akad. Nauk SSSR, vol. 300, no. 3 (1988), pp. 529–534.

- [201] I. M. GEL' FAND, A. V. ZELEVINSKIĬ, AND M. M. KAPRANOV, <u>Hypergeometric functions and toric varieties</u>, Funktsional. Anal. i Prilozhen., vol. 23, no. 2 (1989), pp. 12–26, doi:10.1007/ BF01078777.
- [202] I. GESSEL, Two theorems on rational power series, Utilitas Math., vol. 19 (1981), pp. 247–253.
- [203] I. M. GESSEL, A factorization for formal Laurent series and lattice path enumeration, J. Combin. Theory Ser. A, vol. 28, no. 3 (1980), pp. 321–337, doi:10.1016/0097-3165(80)90074-6.
- [204] I. M. GESSEL, A probabilistic method for lattice path enumeration, J. Statist. Plann. Inference, vol. 14, no. 1 (1986), pp. 49–58, doi:10.1016/0378-3758(86)90009-1.
- [205] I. M. GESSEL AND D. ZEILBERGER, Random walk in a Weyl chamber, Proc. Amer. Math. Soc., vol. 115, no. 1 (1992), pp. 27–31, doi:10.1090/S0002-9939-1992-1092920-8.
- [206] M. L. GLASSER AND E. MONTALDI, Some integrals involving Bessel functions, J. Math. Anal. Appl., vol. 183, no. 3 (1994), pp. 577–590, doi:10.1006/jmaa.1994.1164.
- [207] M. L. GLASSER AND I. J. ZUCKER, Extended Watson integrals for the cubic lattices, Proc. Nat. Acad. Sci. U.S.A., vol. 74, no. 5 (1977), pp. 1800–1801.
- [208] R. W. GOSPER, JR., Decision procedure for indefinite hypergeometric summation, Proc. Nat. Acad. Sci. U.S.A., vol. 75, no. 1 (1978), pp. 40–42, http://www.jstor.org/stable/67570.
- [209] H. W. GOULD, Combinatorial identities, Henry W. Gould, Morgantown, W.Va., 1972. A standardized set of tables listing 500 binomial coefficient summations.
- [210] D. GOUYOU-BEAUCHAMPS, <u>Chemins sous-diagonaux et tableaux de Young</u>, in Combinatoire énumérative (Montreal, Que., 1985/Quebec, Que., 1985), vol. 1234 of Lecture Notes in Math., Springer, Berlin, 1986, pp. 112–125, doi:10.1007/BFb0072513.
- [211] D. J. GRABINER AND P. MAGYAR, <u>Random walks in Weyl chambers and the decomposition</u> of tensor powers, J. Algebraic Combin., vol. 2, no. 3 (1993), pp. 239–260, doi:10.1023/A: 1022499531492.
- [212] J. GRAY, Linear differential equations and group theory from Riemann to Poincaré, Birkhäuser, 1986, doi:10.1007/978-0-8176-4773-5.
- [213] J. J. GRAY, Fuchs and the theory of differential equations, Bull. Amer. Math. Soc. (N.S.), vol. 10, no. 1 (1984), pp. 1–26, doi:10.1090/S0273-0979-1984-15186-3.
- [214] P. A. GRIFFITHS, On the periods of integrals on algebraic manifolds, Rice Univ. Studies, vol. 54, no. 4 (1968), pp. 21–38, doi:http://projecteuclid.org/euclid.bams/1183531477.
- [215] P. A. GRIFFITHS, On the periods of certain rational integrals. I, II, Ann. of Math. (2), vol. 90 (1969), pp. 460–495, 496–541.
- [216] P. A. GRIFFITHS, Periods of integrals on algebraic manifolds: Summary of main results and discussion of open problems, Bull. Amer. Math. Soc., vol. 76 (1970), pp. 228–296, doi:10. 1090/S0002-9904-1970-12444-2.
- [217] D. Y. GRIGOR' EV, Complexity of factoring and calculating the GCD of linear ordinary differential operators, J. Symbolic Comput., vol. 10, no. 1 (1990), pp. 7–37, doi:10.1016/S0747-7171(08) 80034-X.
- [218] H. D. GROSSMAN, Fun with lattice points, Scripta Math., vol. 16 (1950), pp. 207-212.
- [219] A. GROTHENDIECK, On the de Rham cohomology of algebraic varieties, Inst. Hautes Études Sci. Publ. Math., no. 29 (1966), pp. 95–103, http://www.numdam.org/item/PMIHES_1966_29_ _95_0.
- [220] F. GUILLEMIN AND J. S. H. VAN LEEUWAARDEN, Rare event asymptotics for a random walk in the quarter plane, Queueing Syst., vol. 67, no. 1 (2011), pp. 1–32, doi:10.1007/s11134-010-9197-7.
- [221] A. J. GUTTMANN, On the recurrence relation method of series analysis, J. Phys. A, vol. 8, no. 7 (1975), pp. 1081–1088, 10.1088/0305-4470/8/1081.
- [222] A. J. GUTTMANN, Lattice Green's functions in all dimensions, J. Phys. A, vol. 43, no. 30 (2010), pp. 305205, 26, doi:10.1088/1751-8113/43/30/305205.
- [223] A. J. GUTTMANN, Series extension: predicting approximate series coefficients from a finite number of exact coefficients, J. Phys. A, vol. 49, no. 41 (2016), pp. 415002, 27, doi:10.1088/1751-8113/ 49/41/415002.
- [224] A. J. GUTTMANN AND G. S. JOYCE, On a new method of series analysis in lattice statistics, J. Phys. A: Gen. Phys., vol. 5, no. 9 (1972), pp. 81–84, doi:10.1088/0305-4470/5/9/001.
- [225] R. K. GUY, C. KRATTENTHALER, AND B. E. SAGAN, Lattice paths, reflections, & dimension-changing bijections, Ars Combin., vol. 34 (1992), pp. 3–15.
- [226] A. HALD, A history of probability and statistics and their applications before 1750, Wiley Series in Probability and Mathematical Statistics: Probability and Mathematical Statistics, John Wiley & Sons, Inc., New York, 1990, doi:10.1002/0471725161. A Wiley-Interscience Publication.
- [227] R. HARLEY, On the theory of the transcendental solution of algebraic equations, Quart. Journal of Pure and Applied Math, vol. 5 (1862), pp. 337–361.
- [228] R. HARTSHORNE, On the de Rham cohomology of algebraic varieties, Inst. Hautes Études Sci. Publ. Math., no. 45 (1975), pp. 5–99, http://www.numdam.org/item/PMIHES_1975_45_
_5_0.

- [229] S. HASSANI, C. KOUTSCHAN, J.-M. MAILLARD, AND N. ZENINE, Lattice Green functions: the $\frac{d$ -dimensional face-centered cubic lattice, d = 8,9,10,11,12, J. Phys. A, vol. 49, no. 16 (2016), pp. 164003, 30, doi:10.1088/1751-8113/49/16/164003.
- [230] W. HEBISCH AND M. RUBEY, Extended rate, more GFUN, J. Symbolic Comput., vol. 46, no. 8 (2011), pp. 889–903, doi:10.1016/j.jsc.2011.01.004.
- [231] E. HEINE, Der Eisensteinsche Satz über Reihen-Entwickelung algebraischer Functionen, J. Reine Angew. Math., vol. 45 (1853), pp. 285–302, doi:10.1515/crll.1853.45.285.
- [232] C. HERMITE, Sur l'intégration des fractions rationnelles, Ann. Sci. École Norm. Sup. (2), vol. 1 (1872), pp. 215–218, http://www.numdam.org/item/ASENS_1872_2_1_215_0.
- [233] C. HERMITE, Sur la fonction exponentielle, C. R. Acad. Sci. Paris, vol. 77 (1873), pp. 18-24.
- [234] C. HERMITE, Sur la généralisation des fractions continues algébriques, Ann. Mat. Pura Appl., vol. 21, no. 2 (1893), pp. 289–308.
- [235] T. HONDA, <u>Algebraic differential equations</u>, in Symposia Mathematica, Vol. XXIV (Sympos., IN-DAM, Rome, 1979), Academic Press, London-New York, 1981, pp. 169–204.
- [236] H. HUANG, New bounds for hypergeometric creative telescoping, in Proceedings of the 2016 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2016, pp. 279–286, doi:10.1145/2930889.2930893.
- [237] K. HUMPHREYS, <u>A history and a survey of lattice path enumeration</u>, J. Statist. Plann. Inference, vol. 140, no. 8 (2010), pp. 2237–2254, doi:10.1016/j.jspi.2010.01.020.
- [238] K. HUMPHREYS AND H. NIEDERHAUSEN, Counting lattice paths with an infinite step set and special access, in The Ninth Quadrennial International Conference on Graph Theory, Combinatorics, Algorithms and Applications, vol. 11 of Electron. Notes Discrete Math., Elsevier Sci. B. V., Amsterdam, 2002, p. 11.
- [239] K. HUMPHREYS AND H. NIEDERHAUSEN, Counting lattice paths taking steps in infinitely many directions under special access restrictions, Theoret. Comput. Sci., vol. 319, no. 1-3 (2004), pp. 385–409, doi:10.1016/j.tcs.2004.02.008.
- [240] E. IMAMOGLU AND M. VAN HOEIJ, Computing hypergeometric solutions of second order linear differential equations using quotients of formal solutions, in ISSAC'15—Proceedings of the 2015 ACM International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2015, pp. 235–242, doi:10.1145/2755996.2756651.
- [241] E. IMAMOGLU AND M. VAN HOEIJ, Computing hypergeometric solutions of second order linear differential equations using quotients of formal solutions and integral bases, J. Symbolic Comput., vol. 83 (2017), pp. 254–271, doi:10.1016/j.jsc.2016.11.014.
- [242] S. JOHNSON, M. MISHNA, AND K. YEATS, Towards a combinatorial understanding of lattice path asymptotics, Tech. Report 1305.7418, arXiv, 2013. Available on-line at http://arxiv.org/abs/ math/1304.7418.
- [243] S. JOHNSON, M. MISHNA, AND K. YEATS, <u>A</u> combinatorial understanding of lattice path asymptotics, Adv. in Appl. Math., vol. 92 (2018), pp. 144–163, doi:10.1016/j.aam.2017.08.001.
- [244] J. KAMPÉ DE FÉRIET, La fonction hypergéométrique. Mém. Sci. Math. 85. Paris: Gauthier-Villars. 87 p., 1937, http://www.numdam.org/item/MSM_1937_85_1_0.
- [245] N. M. KATZ, On the differential equations satisfied by period matrices, Inst. Hautes Études Sci. Publ. Math., no. 35 (1968), pp. 223–258, http://www.numdam.org/item/PMIHES_1968_ _35_71_0.
- [246] N. M. KATZ, Nilpotent connections and the monodromy theorem: Applications of a result of Turrittin, Inst. Hautes Études Sci. Publ. Math., no. 39 (1970), pp. 175–232, http://www.numdam.org/item/PMIHES_1970_39_175_0.
- [247] N. M. KATZ, The regularity theorem in algebraic geometry, in Actes du Congrès International des Mathématiciens (Nice, 1970), Tome 1, Gauthier-Villars, Paris, 1971, pp. 437–443.
- [248] N. M. KATZ, Algebraic solutions of differential equations (*p*-curvature and the Hodge filtration), Invent. Math., vol. 18 (1972), pp. 1–118, doi:10.1007/BF01389714.
- [249] N. M. KATZ, <u>A conjecture in the arithmetic theory of differential equations</u>, Bull. Soc. Math. France, vol. 110, no. 2 (1982), pp. 203–239, http://www.numdam.org/item/BSMF_1982_ _110_203_0.
- [250] N. M. KATZ AND T. ODA, On the differentiation of de Rham cohomology classes with respect to parameters, J. Math. Kyoto Univ., vol. 8 (1968), pp. 199–213, doi:10.1215/kjm/1250524135.
 [251] M. KAUERS, Guessing handbook, Tech. Report 09-07, RISC-Linz, 2009. Available at http://www.
- [251] M. KAUERS, <u>Guessing handbook</u>, Tech. Report 09-07, RISC-Linz, 2009. Available at http://www. risc.jku.at/publications/download/risc_3814/demo.nb.pdf.
- [252] M. KAUERS, The holonomic toolkit, in Computer algebra in quantum field theory, Texts Monogr. Symbol. Comput., Springer, Vienna, 2013, pp. 119–144, doi:10.1007/978-3-7091-1616-6_5.
- [253] M. KAUERS, <u>Computer algebra</u>, in Handbook of enumerative combinatorics, Discrete Math. Appl. (Boca Raton), CRC Press, Boca Raton, FL, 2015, pp. 589–678, doi:10.1201/b18255-13.

ALIN BOSTAN

- [254] M. KAUERS, C. KOUTSCHAN, AND D. ZEILBERGER, Proof of Ira Gessel's lattice path conjecture, Proc. Natl. Acad. Sci. USA, vol. 106, no. 28 (2009), pp. 11502–11505, doi:10.1073/pnas.0901678106.
- [255] M. KAUERS AND R.-H. WANG, Lattice walks in the octant with infinite associated groups, Electronic Notes in Discrete Mathematics, vol. 61, no. 4 (2017), pp. 703–709, doi:10.1016/j.endm.2017. 07.026. Proc. EUROCOMB'17.
- [256] M. KAUERS AND R. YATCHAK, Walks in the quarter plane with multiple steps, in Proceedings of FPSAC 2015, Discrete Math. Theor. Comput. Sci. Proc., Assoc. Discrete Math. Theor. Comput. Sci., Nancy, 2015, pp. 25–36, http://dmtcs.episciences.org/2463.
- [257] M. KAUERS AND D. ZEILBERGER, The quasi-holonomic ansatz and restricted lattice walks, J. Difference Equ. Appl., vol. 14, no. 10-11 (2008), pp. 1119–1126, doi:10.1080/10236190802332084.
- [258] M. KAUERS AND D. ZEILBERGER, The computational challenge of enumerating high-dimensional rook walks, Adv. in Appl. Math., vol. 47, no. 4 (2011), pp. 813–819, doi:10.1016/j.aam.2011. 03.004.
- [259] M. A. H. KHAN, High-order differential approximants, J. Comput. Appl. Math., vol. 149, no. 2 (2002), pp. 457–468, doi:10.1016/S0377-0427(02)00561-7.
- [260] J. F. C. KINGMAN, Two similar queues in parallel, Ann. Math. Statist., vol. 32 (1961), pp. 1314–1323, doi:10.1214/aoms/1177704869.
- [261] D. E. KNUTH, The art of computer programming. Vol. 1: Fundamental algorithms, Second printing, Addison-Wesley, Reading, MA, 1968.
- [262] C. KOUTSCHAN, Eliminating human insight: an algorithmic proof of Stembridge's TSPP theorem, in Gems in experimental mathematics, vol. 517 of Contemp. Math., Amer. Math. Soc., Providence, RI, 2010, pp. 219–230, doi:10.1090/conm/517/10143.
- [263] C. KOUTSCHAN, A fast approach to creative telescoping, Math. Comput. Sci., vol. 4, no. 2-3 (2010), pp. 259–266, doi:10.1007/s11786-010-0055-0.
- [264] C. KOUTSCHAN, Creative telescoping for holonomic functions, in Computer algebra in quantum field theory, Texts Monogr. Symbol. Comput., Springer, Vienna, 2013, pp. 171–194, doi:10. 1007/978-3-7091-1616-6_7.
- [265] C. KOUTSCHAN, Lattice Green functions of the higher-dimensional face-centered cubic lattices, J. Phys. A, vol. 46, no. 12 (2013), pp. 125005, 14, doi:10.1088/1751-8113/46/12/125005.
- [266] C. KOUTSCHAN, M. KAUERS, AND D. ZEILBERGER, Proof of George Andrews's and David Robbins's <u>q</u>-TSPP conjecture, Proc. Natl. Acad. Sci. USA, vol. 108, no. 6 (2011), pp. 2196–2199, doi:10. 1073/pnas.1019186108.
- [267] C. KOUTSCHAN AND T. THANATIPANONDA, Advanced computer algebra for determinants, Ann. Comb., vol. 17, no. 3 (2013), pp. 509–523, doi:10.1007/s00026-013-0183-8.
- [268] J. J. KOVACIC, An algorithm for solving second order linear homogeneous differential equations, J. Symbolic Comput., vol. 2, no. 1 (1986), pp. 3–43, doi:10.1016/S0747-7171(86)80010-4.
- [269] C. KRATTENTHALER, Lattice path enumeration, in Handbook of enumerative combinatorics, Discrete Math. Appl. (Boca Raton), CRC Press, Boca Raton, FL, 2015, pp. 589–678, doi:10.1201/ b18255-13.
- [270] G. KREWERAS, Sur une classe de problèmes de dénombrement liés au treillis des partitions des entiers, Cahiers du B.U.R.O., vol. 6 (1965), pp. 9–107, http://www.numdam.org/item/ BURO_1965__6__9_0.
- [271] G. KREWERAS AND H. NIEDERHAUSEN, Solution of an enumerative problem connected with lattice paths, European J. Combin., vol. 2, no. 1 (1981), pp. 55–60, doi:10.1016/S0195-6698(81) 80020-0.
- [272] E. E. KUMMER, über die hypergeometrische Reihe . (Fortsetzung), J. Reine Angew. Math., vol. 15 (1836), pp. 127–172, doi:10.1515/crll.1836.15.127.
- [273] J. P. S. KUNG AND A. DE MIER, <u>Catalan lattice paths with rook, bishop and spider steps</u>, J. Combin. Theory Ser. A, vol. 120, no. 2 (2013), pp. 379–389, doi:10.1016/j.jcta.2012.08.010.
- [274] V. J. KUNWAR AND M. VAN HOEIJ, Second order differential equations with hypergeometric solutions of degree three, in ISSAC 2013—Proceedings of the 38th International Symposium on Symbolic and Algebraic Computation, ACM, New York, 2013, pp. 235–242, doi:10.1145/2465506.2465953.
- [275] I. KURKOVA AND K. RASCHEL, Explicit expression for the generating function counting Gessel's walks, Adv. in Appl. Math., vol. 47, no. 3 (2011), pp. 414–433, doi:10.1016/j.aam.2010.11.004.
- [276] I. KURKOVA AND K. RASCHEL, Random walks in (Z+)² with non-zero drift absorbed at the axes, Bull. Soc. Math. France, vol. 139, no. 3 (2011), pp. 341–387, http://eudml.org/doc/272541.
- [277] I. KURKOVA AND K. RASCHEL, On the functions counting walks with small steps in the quarter plane, Publ. Math. Inst. Hautes Études Sci., vol. 116, no. 1 (2012), pp. 69–114, doi:10.1007/ s10240-012-0045-7.
- [278] I. KURKOVA AND K. RASCHEL, New steps in walks with small steps in the quarter plane: series expressions for the generating functions, Ann. Comb., vol. 19, no. 3 (2015), pp. 461–511,

doi:10.1007/s00026-015-0279-4.

- [279] J. LABELLE, Langages de Dyck généralisés, Ann. Sci. Math. Québec, vol. 17, no. 1 (1993), pp. 53-64.
- [280] J. LABELLE AND Y. N. YEH, Generalized Dyck paths, Discrete Math., vol. 82, no. 1 (1990), pp. 1–6, doi:10.1016/0012-365X(90)90039-K.
- [281] P. LAIREZ, Computing periods of rational integrals, Math. Comp., vol. 85, no. 300 (2016), pp. 1719– 1752, doi:10.1090/mcom/3054.
- [282] E. LANDAU, Eine Anwendung des Eisensteinschen Satzes auf die Theorie der Gaussschen Differentialgleichung., Journal für die reine und angewandte Mathematik, vol. 127 (1904), pp. 92–102, http://eudml.org/doc/149180.
- [283] E. LANDAU, Über einen zahlentheoretischen Satz und seine Anwendung auf die hypergeometrische Reihe, S.-B. Heidelberger Akad. Wiss., vol. 18 (1911), pp. 3–38, http://eudml.org/doc/149180.
- [284] G. F. LAWLER AND V. LIMIC, <u>Random walk: a modern introduction</u>, vol. 123 of Cambridge Studies in Advanced Mathematics, Cambridge University Press, Cambridge, 2010, doi:10.1017/ CBO9780511750854.
- [285] A. K. LENSTRA, H. W. LENSTRA, JR., AND L. LOVÁSZ, Factoring polynomials with rational coefficients, Math. Ann., vol. 261, no. 4 (1982), pp. 515–534, doi:10.1007/BF01457454.
- [286] F. LINDEMANN, Ueber die Zahl π.*), Math. Ann., vol. 20, no. 2 (1882), pp. 213–225, doi:10.1007/ BF01446522.
- [287] L. LIPSHITZ, The diagonal of a D-finite power series is D-finite, J. Algebra, vol. 113, no. 2 (1988), pp. 373–378, doi:10.1016/0021-8693(88)90166-4.
- [288] L. LIPSHITZ, <u>D-finite power series</u>, J. Algebra, vol. 122, no. 2 (1989), pp. 353–373, doi:10.1016/ 0021-8693(89)90222-6.
- [289] L. LIPSHITZ AND A. J. VAN DER POORTEN, Rational functions, diagonals, automata and arithmetic, in Number theory (Banff, AB, 1988), de Gruyter, Berlin, 1990, pp. 339–358.
- [290] K. MAHLER, Applications of some formulae by Hermite to the approximation of exponentials and logarithms, Math. Ann., vol. 168 (1967), pp. 200–227, doi:10.1007/BF01361554.
- [291] K. MAHLER, Perfect systems, Compositio Math., vol. 19, no. 2 (1968), pp. 95–166, http://eudml. org/doc/88959.
- [292] V. A. MALYSHEV, The solution of the discrete Wiener-Hopf equations in a quarter-plane, Dokl. Akad. Nauk SSSR, vol. 187 (1969), pp. 1243–1246.
- [293] V. A. MALYSHEV, Wiener-Hopf equations in the quarter-plane, discrete groups and automorphic functions, Math. USSR Sbornik, vol. 13 (1971), pp. 491–516. Translated from Mat. Sbornik, Tom 84 (126), no. 4, pp. 499–525, 1971.
- [294] V. A. MALYSHEV, An analytical method in the theory of two-dimensional positive random walks, Sib. Math. J., vol. 13 (1972), pp. 917–929. Translated from Sibirskii Matematicheskii Zhurnal, Vol. 13, No. 6, pp. 1314–1329, November-December, 1972.
- [295] V. A. MALYSHEV, Wiener-Hopf equations and their applications in probability theory, in Probability theory. Mathematical statistics. Theoretical cybernetics, Vol. 13 (Russian), Itogi Nauki i Tehniki, Akad. Nauk SSSR Vsesojuz. Inst. Naučn. i Tehn. Informacii, Moscow, 1976, pp. 5–35.
- [296] J. I. MANIN, Algebraic curves over fields with differentiation, Izv. Akad. Nauk SSSR. Ser. Mat., vol. 22, no. 6 (1958), pp. 737–756. English transl., Amer. Math. Soc. Transl. Ser. 2, vol. 37, Amer. Math. Soc., Providence, RI, 1964, pp. 59–78.
- [297] J. I. MANIN, Rational points of algebraic curves over function fields, Izv. Akad. Nauk SSSR. Ser. Mat., vol. 27, no. 6 (1963), pp. 1395–1440. English transl., Amer. Math. Soc. Transl. Ser. 2, vol. 59, Amer. Math. Soc., Providence, RI, 1966, pp. 189–234.
- [298] R. MANSUY, Promenade sur le quart de plan, La Recherche, vol. 510 (2016), p. 33, http://www. larecherche.fr/chronique-mathématique/promenade-sur-le-quart-de-plan.
- [299] M. MATSUDA, Lectures on algebraic solutions of hypergeometric differential equations, vol. 15 of Lectures in Mathematics, Kinokuniya Company Ltd., Tokyo, 1985, http://www.math. kyoto-u.ac.jp/library/lmku/lmku15.pdf.
- [300] W. H. MCCREA AND F. J. W. WHIPPLE, <u>Random paths in two and three dimensions</u>, Proc. Roy. Soc. Edinburgh, vol. 60 (1940), pp. 281–298, doi:10.1017/S0370164600020265.
- [301] S. MELCZER AND M. MISHNA, Singularity analysis via the iterated kernel method, in DMTCS Proceedings of FPSAC'13, Paris, France, 2013, pp. 481–492, http://www.dmtcs.org/dmtcs-ojs/ index.php/proceedings/article/viewArticle/dmAS0141.html.
- [302] S. MELCZER AND M. MISHNA, Asymptotic lattice path enumeration using diagonals, in Proceedings of the 25th International Conference on Probabilistic, Combinatorial and Asymptotic Methods for the Analysis of Algorithms, Discrete Math. Theor. Comput. Sci. Proc., BA, Assoc. Discrete Math. Theor. Comput. Sci., Nancy, 2014, pp. 313–324, http://hal.inria.fr/ hal-01077251.
- [303] S. MELCZER AND M. MISHNA, Enumerating lattice paths with symmetries through multivariate

ALIN BOSTAN

diagonals, in AofA 2014, Paris, France, Discrete Math. Theor. Comput. Sci. Proc., AK, Assoc.
 Discrete Math. Theor. Comput. Sci., Nancy, 2014, pp. 1–12, http://hal.inria.fr/hal-01077251.
 [304] S. MELCZER AND M. MISHNA, Asymptotic lattice path enumeration using diagonals, Algorithmica,

- vol. 75, no. 4 (2016), pp. 782–811, doi:10.1007/s00453-015-0063-1.
- [305] S. MELCZER AND M. C. WILSON, Asymptotics of lattice walks via analytic combinatorics in several variables, in DMTCS Proceedings of FPSAC'16, Vancouver, Canada, 2016, pp. 863–874, http://www.lix.polytechnique.fr/~pilaud/FPSAC16/listPapers.html.
- [306] D. MIRIMANOFF, À propos de l'interprétation géométrique du problème du scrutin, Enseignement Math., vol. 23 (1923), pp. 187–189, doi:10.5169/seals-19739.
- [307] M. MISHNA, Classifying lattice walks restricted to the quarter plane, in Proceedings of FPSAC'07, Tianjin, China, 2007. Available on-line at http://igm.univ-mlv.fr/~fpsac/FPSAC07/SITE07/ conpap.htm.
- [308] M. MISHNA, Classifying lattice walks restricted to the quarter plane, J. Combin. Theory Ser. A, vol. 116, no. 2 (2009), pp. 460–477, doi:10.1016/j.jcta.2008.06.011.
- [309] M. MISHNA, Algorithmic approaches for lattice path combinatorics, in Proc. ISSAC'17, ACM, 2017, pp. 9–10, doi:10.1145/3087604.3087664.
- [310] M. MISHNA AND A. RECHNITZER, Two non-holonomic lattice walks in the quarter plane, Theoret. Comput. Sci., vol. 410, no. 38-40 (2009), pp. 3616–3630, doi:10.1016/j.tcs.2009.04.008.
- [311] M. MOHAMMED AND D. ZEILBERGER, Sharp upper bounds for the orders of the recurrences output by the Zeilberger and q-Zeilberger algorithms, J. Symbolic Comput., vol. 39, no. 2 (2005), pp. 201–207, doi:10.1016/j.jsc.2004.10.002.
- [312] S. G. MOHANTY, Lattice path counting and applications, Academic Press [Harcourt Brace Jovanovich Publishers], New York, 1979. Probability and Mathematical Statistics.
- [313] P. MONSKY, Finiteness of de Rham cohomology, Amer. J. Math., vol. 94 (1972), pp. 237–245, doi:10.2307/2373603.
- [314] E. W. MONTROLL, Random walks in multidimensional spaces, especially on periodic lattices, J. Soc. Indust. Appl. Math., vol. 4 (1956), pp. 241–260.
- [315] T. V. NARAYANA, Lattice path combinatorics with statistical applications, vol. 23 of Mathematical Expositions, University of Toronto Press, Toronto, Ont., 1979.
- [316] I. NEMES, M. PETKOV^{*} SEK, H. S. WILF, AND D. ZEILBERGER, <u>How to do Monthly problems with</u> your computer, Amer. Math. Monthly, vol. 104, no. 6 (1997), pp. 505–519, doi:10.2307/ 2975078.
- [317] H. NIEDERHAUSEN, The ballot problem with three candidates, European J. Combin., vol. 4, no. 2 (1983), pp. 161–167, doi:10.1016/S0195-6698(83)80046-8.
- [318] H. NIEDERHAUSEN, Random walks in octants, and related structures, J. Statist. Plann. Inference, vol. 135, no. 1 (2005), pp. 165–196, doi:10.1016/j.jspi.2005.02.013.
- [319] M. OHTSUKI, On the number of apparent singularities of a linear differential equation, Tokyo J. Math., vol. 5, no. 1 (1982), pp. 23–29, doi:10.3836/tjm/1270215031.
- [320] H. PADÉ, Sur la généralisation des fractions continues algébriques, J. Math. Pures Appl., vol. 10, no. 4 (1894), pp. 291–330, http://eudml.org/doc/235790.
- [321] V. Y. PAN, Structured matrices and polynomials, Birkhäuser Boston, Inc., Boston, MA; Springer-Verlag, New York, 2001, doi:10.1007/978-1-4612-0129-8. Unified superfast algorithms.
- [322] R. PEMANTLE AND M. C. WILSON, <u>Analytic combinatorics in several variables</u>, vol. 140 of Cambridge Studies in Advanced Mathematics, Cambridge University Press, Cambridge, 2013, doi:10.1017/CBO9781139381864.
- [323] M. PETKOVŠEK, The irrational chess knight, in Proc. FPSAC'98, 1998, pp. 513–522.
- [324] M. PETKOVŠEK, Partial recurrences with constant coefficients: the algebraic case, Programmirovanie, no. 6 (1998), pp. 8–9.
- [325] M. PETKOVŠEK, H. S. WILF, AND D. ZEILBERGER, A = B, A. K. Peters, Wellesley, MA, 1996, http: //www.math.upenn.edu/~wilf/AeqB.html. With a foreword by Donald E. Knuth.
- [326] É. PICARD, Sur les intégrales doubles de fonctions rationnelles dont tous les résidus sont nuls, Bull. Sci. Math. (2), vol. 26 (1902), pp. 143–152.
- [327] É. PICARD AND G. SIMART, Théorie des fonctions algébriques de deux variables indépendantes, vol. I (1897) and II (1906), Gauthier-Villars et fils, 1897.
- [328] G. PÓLYA, Quelques problèmes de probabilité se rapportant à la «promenade au hasard », Enseignement Math., vol. 20 (1919), pp. 444–445.
 [329] G. PÓLYA, Über eine Aufgabe der Wahrscheinlichkeitsrechnung betreffend die Irrfahrt im
- [329] G. PÓLYA, <u>Uber eine Aufgabe der Wahrscheinlichkeitsrechnung betreffend die Irrfahrt im</u> Straßennetz, Math. Ann., vol. 84, no. 1-2 (1921), pp. 149–160, doi:10.1007/BF01458701.
- [330] G. PÓLYA, Sur les séries entières, dont la somme est une fonction algébrique, Enseignement Math., vol. 22 (1921/1922), pp. 38–47, doi:10.5169/seals-515729.
- [331] G. Półya, Guessing and proving, The Two-Year College Mathematics Journal, vol. 9, no. 1 (1978), pp. 21–27, http://www.jstor.org/stable/3026553.

- [332] G. PÓLYA, Mathematics and plausible reasoning. Vol. I, Princeton University Press, Princeton, NJ, 1990, Mathematicsandplausiblereasoning. Induction and analogy in mathematics, Reprint of the 1954 original.
- [333] G. PÓLYA, How to solve it, Princeton Science Library, Princeton University Press, Princeton, NJ, 2004, http://press.princeton.edu/titles/669.html. A new aspect of mathematical method, Expanded version of the 1988 edition, with a new foreword by John H. Conway.
- [334] E. G. C. POOLE, Introduction to the theory of linear differential equations, Oxford Univ. Press, London, 1936.
- [335] H. PRODINGER, The kernel method: a collection of examples, Sém. Lothar. Combin., vol. 50 (2004), pp. Art. B50f, 19 pp., http://www.mat.univie.ac.at/~slc/s/s50proding.html.
- [336] K. RASCHEL, Counting walks in a quadrant: a unified approach via boundary value problems, J. Eur. Math. Soc. (JEMS), vol. 14, no. 3 (2012), pp. 749–777, doi:10.4171/JEMS/317.
- [337] K. RASCHEL, Promenades sur le quart de plan, La Recherche, vol. 462 (2012), pp. 18–19, http: //www.lmpt.univ-tours.fr/~raschel/LR_462_018maths.pdf.
- [338] K. RASCHEL, Random walks in the quarter plane, discrete harmonic functions and conformal mappings, Stochastic Process. Appl., vol. 124, no. 10 (2014), pp. 3147–3178, doi:10.1016/j.spa. 2014.04.013. With an appendix by Sandro Franceschi.
- [339] K. RASCHEL, Walks in the quarter plane: analytic approach and applications, in Journées MAS 2012, vol. 44 of ESAIM Proc., EDP Sci., Les Ulis, 2014, pp. 140–149, doi:10.1051/proc/ 201444008.
- [340] M. RENAULT, Lost (and found) in translation: André's *actual* method and its application to the generalized ballot problem, Amer. Math. Monthly, vol. 115, no. 4 (2008), pp. 358–363, http://www.jstor.org/stable/27642480.
- [341] J. RIORDAN, Combinatorial identities, Robert E. Krieger Publishing Co., Huntington, N.Y., 1979. Reprint of the 1968 original.
- [342] F. RODRIGUEZ-VILLEGAS, Integral ratios of factorials and algebraic hypergeometric functions, 2005. Oberwolfach reports, http://arxiv.org/abs/math/0701362.
- [343] M. ROTHSTEIN, Aspects of symbolic integration and simplification of exponential and primitive functions, PhD thesis, The Univ. of Wisconsin - Madison, 1976.
- [344] M. RUBEY, Transcendence of generating functions of walks on the slit plane, in Mathematics and computer science. III, Trends Math., Birkhäuser, Basel, 2004, pp. 49–58.
- [345] B. SALVY AND P. ZIMMERMANN, Gfun: a Maple package for the manipulation of generating and holonomic functions in one variable, ACM Transactions on Mathematical Software, vol. 20, no. 2 (1994), pp. 163–177, doi:10.1145/178365.178368.
- [346] A. L. SCHMIDT, Legendre transforms and Apéry's sequences, J. Austral. Math. Soc. Ser. A, vol. 58, no. 3 (1995), pp. 358–375.
- [347] H. SCHMITT, Operators with nilpotent *p*-curvature, Proc. Amer. Math. Soc., vol. 119, no. 3 (1993), pp. 701–710, doi:10.2307/2160503.
- [348] A. SCHÖNHAGE AND V. STRASSEN, <u>Schnelle Multiplikation großer Zahlen</u>, Computing, vol. 7 (1971), pp. 281–292.
- [349] H. A. SCHWARZ, Über diejenigen Fälle, in welchen die Gaußische hypergeometrische Reihe einer algebraische Funktion ihres vierten Elementes darstellt, J. Reine Angew. Math., vol. 75 (1873), pp. 292–335, http://www.digizeitschriften.de/de/dms/img/?PID=GDZPPN002155206.
- [350] A. V. ŠERGEYEV, A recursive algorithm for Padé–Hermite approximations, USSR Comput. Maths Math. Phys, vol. 26, no. 2 (1986), pp. 17–22, doi:10.1016/0041-5553(86)90003-0. Translated from Zh. Vychisl. Mat. i Mat. Fiz., Vol. 26, No. 3, pp. 348–356, 1986.
- [351] M. F. SINGER. Private communication, 2014.
- [352] M. F. SINGER, Algebraic solutions of *n*th order linear differential equations, in Proceedings of the Queen's Number Theory Conference, 1979 (Kingston, Ont., 1979), vol. 54 of Queen's Papers in Pure and Appl. Math., Queen's Univ., Kingston, Ont., 1980, pp. 379–420.
- [353] M. F. SINGER, Testing reducibility of linear differential operators: a group-theoretic perspective, Appl. Algebra Engrg. Comm. Comput., vol. 7, no. 2 (1996), pp. 77–104, doi:10.1007/ BF01191378.
- [354] N. J. A. SLOANE, The on-line encyclopedia of integer sequences, http://oeis.org.
- [355] F. SPITZER, Principles of random walk, The University Series in Higher Mathematics, D. Van Nostrand Co., Inc., Princeton, N.J.-Toronto-London, 1964.
- [356] R. P. STANLEY, Differentiably finite power series, European J. Combin., vol. 1, no. 2 (1980), pp. 175– 188, doi:10.1016/S0195-6698(80)80051-5.
- [357] R. P. STANLEY, Enumerative combinatorics. Vol. 2, vol. 62, Cambridge Univ. Press, 1999, doi:10. 1017/CBO9780511609589.
- [358] R. P. STANLEY, Catalan numbers, Cambridge University Press, New York, 2015, doi:10.1017/ CBO9781139871495.

ALIN BOSTAN

- [359] J. STIENSTRA, GKZ hypergeometric structures, in Arithmetic and geometry around hypergeometric functions, vol. 260 of Progr. Math., Birkhäuser, Basel, 2007, pp. 313–371, doi:10.1007/ 978-3-7643-8284-1_12.
- [360] A. STORJOHANN, High-order lifting and integrality certification, J. Symb. Comp., vol. 36, no. 3-4 (2003), pp. 613–648, doi:10.1016/S0747-7171(03)00097-X.
- [361] A. STRAUB, Multivariate Apéry numbers and supercongruences of rational functions, Number Theory, vol. 8, no. 8 (2014), pp. 1985–2007, doi:10.2140/ant.2014.8.1985.
- [362] V. STREHL, <u>Recurrences and Legendre transform</u>, Sém. Lothar. Combin., vol. 29 (1992), pp. Art. B29b, 22 pp., <u>http://www.mat.univie.ac.at/~slc/s/s29strehl.html</u>. [Formerly: Publ. I.R.M.A. Strasbourg, 1993, 1993/33, p. 81-100.].
- [363] V. STREHL, <u>Recurrences and Legendre transform</u>, in Séminaire Lotharingien de Combinatoire (Thurnau, 1992), vol. 1993/33 of Prépubl. Inst. Rech. Math. Av., Univ. Louis Pasteur, Strasbourg, 1993, pp. 81–100.
- [364] V. STREHL, <u>Binomial identities</u>—combinatorial and algorithmic aspects, Discrete Math., vol. 136, no. 1-3 (1994), pp. 309–346, doi:10.1016/0012-365X(94)00118-3. Trends in discrete mathematics.
- [365] E. STRIDSBERG, Sur le théorème d'Eisenstein et l'équation différentielle de Gauss, Ark. Mat. Astron. Fys., vol. 6, no. 35 (1911), pp. 1–17.
- [366] R. A. SULANKE, <u>Counting lattice paths by Narayana polynomials</u>, Electron. J. Combin., vol. 7 (2000), pp. Research Paper 40, 9, http://www.combinatorics.org/Volume_7/Abstracts/ v7i1r40.html.
- [367] P. SUN, Proof of two conjectures of Petkovšek and Wilf on Gessel walks, Discrete Math., vol. 312, no. 24 (2012), pp. 3649–3655, doi:10.1016/j.disc.2012.09.003.
- [368] L. TAKÁCS, On the ballot theorems, in Advances in combinatorial methods and applications to probability and statistics, Stat. Ind. Technol., Birkhäuser Boston, Boston, MA, 1997, pp. 97– 114.
- [369] N. TAKAYAMA, <u>GrÖbner basis, integration and transcendental functions</u>, in Proceedings of the International Symposium on Symbolic and Algebraic Computation, ISSAC'90, New York, NY, USA, 1990, ACM, pp. 152–156, doi:10.1145/96877.96916.
- [370] N. Такауама, <u>An approach to the zero recognition problem by Buchberger algorithm</u>, J. Symbolic Comput., vol. 14, no. 2-3 (1992), pp. 265–282, doi:10.1016/0747-7171(92)90039-7.
- [371] N. TAKAYAMA, An algorithm for finding recurrence relations of binomial sums and its complexity, J. Symbolic Comput., vol. 20, no. 5-6 (1995), pp. 637–651, doi:10.1006/jsco.1995.1069. Symbolic computation in combinatorics Δ_1 (Ithaca, NY, 1993).
- [372] J. TANNERY, Propriétés des intégrales des équations différentielles linéaires à coefficients variables, Annales scientifiques de l'ENS Sér. 2, vol. 4 (1875), pp. 113–182, http://www.numdam.org/ item/ASENS_1875_2_4__113_0.
- [373] TCHEBICHEF, <u>Mémoire sur les nombres premiers.</u>, Journal de Mathématiques Pures et Appliquées, (1852), pp. 366–390, <u>http://eudml.org/doc/234762</u>.
- [374] B. M. TRAGER, Algebraic factoring and rational function integration, in Proceedings of the Third ACM Symposium on Symbolic and Algebraic Computation, SYMSAC'76, New York, NY, USA, 1976, ACM, pp. 219–226, doi:10.1145/800205.806338.
- [375] N. TREFETHEN, <u>A hundred-dollar, hundred-digit challenge</u>, SIAM News, vol. 35, no. 1 (2002), p. 1, http://people.maths.ox.ac.uk/trefethen/hundred.html.
- [376] J. VAN DER HOEVEN, FFT-like multiplication of linear differential operators, J. Symbolic Comput., vol. 33, no. 1 (2002), pp. 123–127, doi:10.1006/jsco.2000.0496.
- [377] J. VAN DER HOEVEN, On the complexity of skew arithmetic, Appl. Algebra Engrg. Comm. Comput., vol. 27, no. 2 (2016), pp. 105–122, doi:10.1007/s00200-015-0269-0.
- [378] A. VAN DER POORTEN, A proof that Euler missed...Apéry's proof of the irrationality of $\zeta(3)$, Math. Intelligencer, vol. 1, no. 4 (1978/79), pp. 195–203, doi:10.1007/BF03028234. An informal report.
- [379] M. VAN DER PUT AND M. F. SINGER, Galois theory of linear differential equations, vol. 328 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], Springer-Verlag, Berlin, 2003, doi:0.1007/978-3-642-55750-7.
- [380] M. VAN HOEIJ, Factorization of differential operators with rational functions coefficients, J. Symbolic Comput., vol. 24, no. 5 (1997), pp. 537–561, doi:10.1006/jsco.1997.0151.
- [381] M. VAN HOEIJ AND J.-A. WEIL, Solving second order linear differential equations with Klein's theorem, in ISSAC'05, ACM, New York, 2005, pp. 340–347, doi:10.1145/1073884.1073931.
- [382] P. VERBAETEN, The automatic construction of pure recurrence relations, SIGSAM Bull., vol. 8, no. 3 (1974), pp. 96–98, doi:10.1145/1086837.1086854. Proceedings of EUROSAM'74.
- [383] J. VON ZUR GATHEN AND J. GERHARD, Modern computer algebra, Cambridge University Press, Cambridge, third ed., 2013, doi:10.1017/CBO9781139856065.

- [384] G. H. WEISS, Aspects and applications of the random walk, Random Materials and Processes, North-Holland Publishing Co., Amsterdam, 1994.
- [385] H. S. WILF AND D. ZEILBERGER, An algorithmic proof theory for hypergeometric (ordinary and "q") multisum/integral identities, Invent. Math., vol. 108, no. 3 (1992), pp. 575–633, doi:10.1007/BF02100618.
- [386] J. WIMP AND D. ZEILBERGER, How likely is Polya's drunkard to stay in $x \ge y \ge z$?, J. Statist. Phys., vol. 57, no. 5-6 (1989), pp. 1129–1135, doi:10.1007/BF01020052.
- [387] W.-J. WOAN, <u>Diagonal lattice paths</u>, in Proceedings of the Thirty-second Southeastern International Conference on Combinatorics, Graph Theory and Computing (Baton Rouge, LA, 2001), vol. 151, 2001, pp. 173–178.
- [388] W. WOESS, <u>Random walks on infinite graphs and groups</u>, vol. 138 of Cambridge Tracts in Mathematics, <u>Cambridge University Press</u>, <u>Cambridge</u>, 2000, <u>doi:10.1017/CBO9780511470967</u>.
- [389] G. XIN, Proof of a conjecture on the slit plane problem, Discrete Math., vol. 282, no. 1-3 (2004), pp. 281–287, doi:10.1016/j.disc.2004.01.004.
- [390] D. ZEILBERGER, Sister Celine's technique and its generalizations, J. Math. Anal. Appl., vol. 85, no. 1 (1982), pp. 114–145, doi:10.1016/0022-247X(82)90029-4.
- [391] D. ZEILBERGER, André's reflection proof generalized to the many-candidate ballot problem, Discrete Math., vol. 44, no. 3 (1983), pp. 325–326, doi:10.1016/0012-365X(83)90200-5.
- [392] D. ZEILBERGER, A fast algorithm for proving terminating hypergeometric identities, Math., vol. 80, no. 2 (1990), pp. 207–211, doi:10.1016/0012-365X(90)90120-7.
- [393] D. ZEILBERGER, A holonomic systems approach to special functions identities, J. Comput. Appl. Math., vol. 32, no. 3 (1990), pp. 321–368, doi:10.1016/0377-0427(90)90042-X.
- [394] D. ZEILBERGER, The method of creative telescoping, J. Symbolic Comput., vol. 11, no. 3 (1991), pp. 195–204, doi:10.1016/S0747-7171(08)80044-2.
- [395] D. ZEILBERGER, The umbral transfer-matrix method. I. Foundations, J. Combin. Theory Ser. A, vol. 91, no. 1-2 (2000), pp. 451–463, doi:10.1006/jcta.2000.3110. In memory of Gian-Carlo Rota.
- [396] W. ZHOU, Fast Order Basis and Kernel Basis Computation and Related Problems, PhD thesis, Univ. of Waterloo, 2013.
- [397] I. J. ZUCKER, 70+ years of the Watson integrals, J. Stat. Phys., vol. 145, no. 3 (2011), pp. 591–612, doi:10.1007/s10955-011-0273-0.